At: Gadeirydd ac Aelodau'r Pwyllgor

Safonau

Dyddiad: 25 Chwefror 2016

Rhif Union: 01824 706204

ebost: dcc_admindenbighshire.gov.com.uk

Annwyl Aelod o'r Pwyllgor,

Fe'ch gwahoddir i fynychu cyfarfod y PWYLLGOR SAFONAU, DYDD GWENER, 4 MAWRTH 2016 am 10.00 am yn YSTAFELL BWYLLGORA 1A, NEUADD Y SIR, RHUTHUN LL15 1YN.

Yn gywir iawn

G Williams

Pennaeth Gwasanaethau Cyfreithiol, AD a Democrataidd

AGENDA

RHAN 1: GWAHODDIR Y WASG A'R CYHOEDD I FYNYCHU'R RHAN HON O'R CYFARFOD

- 1 YMDDIHEURIADAU
- 2 DATGAN CYSYLLTIAD (Tudalennau 5 6)

Dylai'r Aelodau ddatgan unrhyw gysylltiad personol neu gysylltiad sy'n rhagfarnu mewn unrhyw fater a nodwyd i'w ystyried yn y cyfarfod hwn.

3 MATERION BRYS FEL Y'U CYTUNWYD GAN Y CADEIRYDD

Rhybudd o eitemau y dylid, ym marn y Cadeirydd, eu hystyried yn y cyfarfod fel materion brys yn unol ag Adran 100B (4) Deddf Llywodraeth Leol, 1972.

4 COFNODION Y CYFARFOD DIWETHAF (Tudalennau 7 - 18)

Derbyn cofnodion cyfarfod y Pwyllgor Safonau a gynhaliwyd ar 4 Rhagfyr, 2015 (copi ynghlwm).

5 RHAGLEN GWAITH I'R DYFODOL Y PWYLLGOR SAFONAU (Tudalennau 19 - 20)

Ystyried Rhaglen Gwaith i'r Dyfodol y Pwyllgor Safonau (copi ynghlwm).

6 ADRODDIAD BLYNYDDOL Y CADEIRYDD 2015/16 (Tudalennau 21 - 28)

Ystyried adroddiad gan y Dirprwy Swyddog Monitro (copi ynghlwm) yn cyflwyno Adroddiad Blynyddol y Cadeirydd i'w ystyried cyn ei gyflwyno i'r Cyngor llawn.

7 HYGYRCHEDD GWYBODAETH GAN GYNGHORAU DINAS, TREF A CHYMUNED.

Derbyn adroddiad llafar gan Julia Hughes (Aelod Annibynnol) (copi ynghlwm).

8 LLYFR ACHOSION COD YMDDYGIAD - OMBWDSMON GWASANAETHAU CYHOEDDUS CYMRU (Tudalennau 29 - 50)

Ystyried adroddiad gan y Swyddog Monitro (copi ynghlwm) yn hysbysu'r aelodau o gyhoeddiadau diweddaraf Llyfr Achosion Cod Ymddygiad yr Ombwdsmon.

9 DIGWYDDIAD HYFFORDDIANT CLERCOD – IONAWR 2016

Derbyn adroddiad blynyddol gan y Dirprwy Swyddog Monitro ar y digwyddiad hyfforddiant diweddaraf a gynhaliwyd ar 11 Ionawr 2016.

10 BIL LLYWODRAETH LEOL (CYMRU) DRAFFT (Tudalennau 51 - 266)

Ystyried adroddiad gan y Swyddog Monitro (copi ynghlwm) yn hysbysu'r aelodau o ddarpariaethau Bil Llywodraeth Leol (Cymru) Drafft sy'n gysylltiedig â'r pwyllgor.

11 PRESENOLDEB MEWN CYFARFODYDD

Nodi presenoldeb aelodau'r Pwyllgor Safonau yng nghyfarfodydd Cynghorau Sir, Tref a Chymuned a derbyn eu hadroddiadau.

12 DYDDIAD Y CYFARFOD NESAF

Cynhelir cyfarfod nesaf y Pwyllgor Safonau am 10.00am ddydd Gwener 24 Mehefin 2016 yn Ystafell Gynadledda 1a, Neuadd y Sir, Rhuthun.

RHAN 2: EITEMAU CYFRINACHOL

Argymhellir, yn unol ag Adran 100A (4) Deddf Llywodraeth Leol 1972, y dylid gwahardd y Wasg a'r Cyhoedd o'r cyfarfod yn ystod trafod y materion canlynol oherwydd y tebygolrwydd y caiff gwybodaeth eithriedig, fel y diffinnir ym mharagraffau 12 ac 13 Rhan 4 o Atodlen 12A y Ddeddf, ei datgelu.

13 COD YMDDYGIAD - RHAN 3 DEDDF LLYWODRAETH LEOL 2000 (Tudalennau 267 - 268)

Ystyried adroddiad cyfrinachol gan y Swyddog Monitro (copi ynghlwm) sy'n rhoi trosolwg o gwynion yn erbyn aelodau a gyflwynwyd gydag Ombwdsmon Gwasanaethau Cyhoeddus Cymru.

AELODAETH

Aelodau Annibynnol: Ian Trigger (Cadeirydd), Mrs Paula White, Julia Hughes ac Anne Mellor

Aelod Cyngor Tref/Cymuned: Y Cynghorydd David E Jones

Cynghorwyr Sir: Y Cynghorydd Bill Cowie Y Cynghorydd Barry Mellor

COPIAU I'R:

Holl Gynghorwyr er gwybodaeth Y Wasg a'r Llyfrgelloedd Cynghorau Tref a Chymuned



Eitem Agenda 2



DEDDF LLYWODRAETH LEOL 2000

Cod Ymddygiad Aelodau

DATGELU A CHOFRESTRU BUDDIANNAU

Rwyf i, (enw)		
*Aelod /Aelod cyfetholedig o (*dileuer un)	Cyngor Sir Ddinbych	
YN CADARNHAU fy mod wedi datgan buddiant *personol / personol a sy'n rhagfarnu nas datgelwyd eisoes yn ôl darpariaeth Rhan III cod ymddygiad y Cyngor Sir i Aelodau am y canlynol:-		
Dyddiad Datgelu:		
Pwyllgor (nodwch):		
Agenda eitem		
Pwnc:		
Natur y Buddiant:		
(Gweler y nodyn isod)*		
Llofnod		
Dyddiad		

Noder: Rhowch ddigon o fanylion os gwelwch yn dda, e.e. 'Fi yw perchennog y tir sy'n gyfagos i'r cais ar gyfer caniatâd cynllunio a wnaed gan Mr Jones', neu 'Mae fy ngŵr / ngwraig yn un o weithwyr y ar gyfer caniatâd cynllunio a wnaeu gan canad cymni sydd wedi gwneud cais am gymorth ariannol'.

Tudalen 5



PWYLLGOR SAFONAU

Cofnodion cyfarfod o'r Pwyllgor Safonau a gynhaliwyd yn Ystafell Bwyllgora 1a, Neuadd y Sir, Rhuthun LL15 1YN, Dydd Gwener, 4 Rhagfyr 2015 am 10.00 am.

YN BRESENNOL

Yr Aelodau Annibynnol Mr Ian Trigger (Cadeirydd), Paula White, Julia Hughes, Anne Mellor a'r Cynghorydd W.L. Cowie.

Roedd y Cyng. M.Ll. Davies yn bresennol fel arsylwr.

HEFYD YN BRESENNOL

Swyddog Monitro (GW) a Swyddog Gweinyddol (CIW).

Cefnogodd yr Aelodau gais gan y Cadeirydd y dylai rhaglenni ar gyfer cyfarfodydd y dyfodol o'r Pwyllgor Safonau fod ar gael i Aelodau'r Pwyllgor un wythnos cyn dyddiad y cyfarfod.

(GW i Weithredu)

1 ETHOL IS-GADEIRYDD

PENDERFYNWYD - penodi Mrs Paula White yn Is-gadeirydd y Pwyllgor Safonau.

2 YMDDIHEURIADAU

Derbyniwyd ymddiheuriadau am absenoldeb oddi wrth y Cynghorwr(wyr) David Jones a/ac Barry Mellor.

3 DATGAN CYSYLLTIAD

Ni wnaeth unrhyw Aelod ddatgan cysylltiad personol na chysylltiad sy'n rhagfarnu ag unrhyw eitem a oedd i'w hystyried yn y cyfarfod.

4 MATERION BRYS FEL Y'U CYTUNWYD GAN Y CADEIRYDD

Yn unol â gofynion Adran 100B(4) Deddf Llywodraeth Leol 1972, nododd y Cadeirydd ei fwriad i gynnwys y mater canlynol i'w drafod oherwydd bod angen rhoi sylw brys iddo:

1. Ymgynghoriad ar Ombwdsmon Gwasanaethau Cyhoeddus Cymru: Cynllun Strategol Tair Blynedd

Hysbyswyd y Pwyllgor gan y Swyddog Monitro ei fod wedi cael dogfen ymgynghori mewn perthynas ag Ombwdsmon Gwasanaethau Cyhoeddus Cymru: Cynllun Strategol Tair Blynedd, ar 3 Rhagfyr, 2015, a bod ymateb wedi'i geisio erbyn 2 Chwefror, 2016.

Cytunodd y Pwyllgor â'r awgrym bod y Swyddog Monitro'n cylchredeg y ddogfen ymgynghori ac yn gofyn am ymatebion gan Aelodau'r Pwyllgor, a fyddai'n cael eu casglu a'u hail-ddosbarthu i'r Aelodau i'w hystyried ymhellach. Eglurodd y Swyddog Monitro nad oedd y geiriad "Cod Ymddygiad" neu "Cod Ymddygiad Aelodau" yn ymddangos yn y ddogfen, a'i fod yn bwriadu cyfeirio at hyn yn yr ymateb.

Cytunodd yr Aelodau i gynnwys eitem ynglŷn ag Ymgynghoriad Ombwdsmon Gwasanaethau Cyhoeddus Cymru: Cynllun Strategol Tair Blynedd, yn rhaglen gwaith i'r dyfodol y Pwyllgor ar gyfer ei gyfarfod ar 4 Mawrth, 2016.

5 COFNODION Y CYFARFOD DIWETHAF

Cyflwynwyd cofnodion cyfarfod y Pwyllgor Safonau a gynhaliwyd ar 22 Mai, 2015.

Cywirdeb:-

Yn bresennol:- Enw'r Cynghorydd D.E. Jones i'w dynnu oddi ar y rhestr o Aelodau a oedd yn bresennol gan ei fod wedi cyflwyno ymddiheuriad am absenoldeb.

Ymddiheuriadau:- Cynnwys enwau'r Cynghorwyr B. Mellor a D.E. Jones fel rhai a oedd wedi cyflwyno ymddiheuriadau am absenoldeb.

Materion yn codi:-

7. Ymweliadau â Chynghorau Dinas, Tref a Chymuned - Mewn ateb i gais gan yr Aelod Annibynnol J. Hughes (JH), cytunwyd darparu rhestr wedi'i diweddaru o ymweliadau gan Aelodau'r Pwyllgor rhwng y cyfarfodydd blaenorol a'r cyfarfodydd presennol, er mwyn sicrhau nad oedd unrhyw ddyblygiad o ymweliadau â Chynghorau Dinas, Tref a Chymuned.

Mewn ymateb i gwestiwn gan yr Aelod Annibynnol A. Mellor (AM), cytunwyd y gallai fod yn fuddiol os byddai JH, fel Aelod o'r Pwyllgor sy'n siarad Cymraeg, yn gallu mynd i gyfarfodydd penodol yn yr ardal a ddynodir fel Clwstwr 3. Cytunodd y Swyddog Monitro y dylid cylchredeg rhestr wedi'i diweddaru o Gynghorau Dinas, Tref a Chymuned, gan gynnwys manylion cyswllt, a lle bo modd, amseroedd a dyddiadau cyfarfodydd, i Aelodau'r Pwyllgor Safonau.

Mewn ymateb i gwestiwn gan y Cynghorydd W.L. Cowie ynglŷn â rôl Aelodau'r Pwyllgor Safonau wrth adrodd yn ôl i'r Pwyllgor Safonau yn dilyn eu presenoldeb mewn cyfarfodydd, yn enwedig mewn achosion lle'r oedd swyddog cyfreithiol y Cyngor hefyd wedi bod yn bresennol. Gofynnodd y Cynghorydd Cowie am eglurhad ynghylch rolau priodol Aelodau'r Pwyllgor Safonau a'r swyddogion cyfreithiol mewn amgylchiadau o'r fath.

Amlinellodd y Swyddog Monitro rôl y Pwyllgor Safonau a'i Aelodau wrth osod ac arsylwi ar safonau i adlewyrchu canfyddiad y cyhoedd o'r modd y mae'r Cyngor a'i Aelodau yn cynnal ei hun. Eglurodd mai cylch gorchwyl y Pwyllgor oedd darparu cefnogaeth i'r Swyddog Monitro ac Aelodau Etholedig y Cynghorau Sir, Dinas, Tref a Chymuned. Cyfeiriodd y Swyddog Monitro at y manteision posibl o drafod

materion a allai godi yn y cyfarfodydd a fynychir gan swyddogion cyfreithiol, a rhoddodd sicrwydd i'r Pwyllgor na fyddai swyddogion cyfreithiol yn teimlo eu bod yn cael eu tanseilio gan bresenoldeb Aelodau'r Pwyllgor Safonau mewn cyfarfodydd roeddent yn bresennol ynddynt.

Yn dilyn trafodaeth bellach:-

PENDERFYNWYD – yn amodol ar yr uchod, derbyn a chymeradwyo'r Cofnodion fel cofnod cywir.

(GW i Weithredu)

6 RHAGLEN GWAITH I'R DYFODOL

Cyflwynodd y Swyddog Monitro adroddiad (a ddosbarthwyd yn flaenorol) yn cyflwyno'r Rhaglen Gwaith i'r Dyfodol i'w hystyried a'i mabwysiadu.

Ystyriodd yr Aelodau Raglen Gwaith i'r Dyfodol y Pwyllgor Safonau a chytunwyd ar y diwygiadau canlynol:-

4 Mawrth, 2016:-

- (a) Ystyried Adroddiad Blynyddol drafft y Pwyllgor Safonau.
- (b) Diweddariad Ymgynghoriad Ombwdsmon Gwasanaethau Cyhoeddus Cymru: Cynllun Strategol Tair Blynedd.
- (c) Diweddariad Llyfr Achos Cod Ymddygiad.

Hysbyswyd yr Aelodau y byddai dyddiadau ar gyfer cyflwyno adroddiadau i'r Pwyllgor, mewn perthynas â chyfarfodydd Fforwm Safonau Gogledd Cymru, yn cael eu cynnwys yn y rhaglen gwaith i'r dyfodol yn dilyn cytundeb ar ddyddiadau cyfarfodydd yn y dyfodol.

PENDERFYNWYD - yn amodol ar ychwanegu'r uchod, bod y Pwyllgor Safonau'n cytuno ar y Rhaglen Gwaith i'r Dyfodol.

(GW, CW i Weithredu)

7 BIL DRAFFT OMBWDSMON GWASANAETHAU CYHOEDDUS (CYMRU) – YMGYNGHORI

Roedd copi o adroddiad gan y Dirprwy Swyddog Monitro (DMO), a oedd yn darparu copi o Fil drafft Ombwdsmon Gwasanaethau Cyhoeddus Cymru, ac yn gofyn am sylwadau gan y Pwyllgor er mwyn llunio ymateb, wedi'i gylchredeg ynghynt.

Eglurwyd y byddai'r adroddiad yn galluogi'r Aelodau i ystyried y cynigion a chyfrannu at y broses ymgynghori ffurfiol.

Mewn cyfarfod diweddar o'r Pwyllgor Safonau, cyflwynwyd eitem fusnes ar Adroddiad Blynyddol Ombwdsmon Gwasanaethau Cyhoeddus Cymru 2014/15, pan gyfeiriwyd at y posibilrwydd o ymestyn pwerau i'r Ombwdsmon yng Nghymru. Roedd y nodyn atgoffa a gyflwynwyd gan Ombwdsmon Gwasanaethau Cyhoeddus Cymru newydd wedi cael ei ymgorffori yn yr adroddiad, gyda'r Bil wedi'i gynnwys fel Atodiad 1 a llythyr ymgynghori fel Atodiad 2. Roedd cyfraniad cynharach Cymdeithas Llywodraeth Leol Cymru yn y broses yn amgaeedig fel Atodiad 3.

21 Hydref 2015, cyhoeddodd Pwyllgor Cyllid Cynulliad Cenedlaethol Cymru ymgynghoriad yn ceisio barn unigolion a sefydliadau ar y cynigion ym Mil Ombwdsmon Gwasanaethau Cyhoeddus (Cymru) drafft, a fwriedir i gryfhau pwerau Ombwdsmon Gwasanaethau Cyhoeddus Cymru. Yn dilyn argymhellion a ddarparwyd mewn adroddiad ym mis Mai, 2015, roedd cynigion y Pwyllgor Cyllid yn y Bil drafft yn cynnwys rhoi pwerau i Ombwdsmon Gwasanaethau Cyhoeddus Cymru:

- Gychwyn ei ymchwiliadau ei hun a derbyn cwynion llafar.
- Delio â chwynion ar draws gwasanaethau cyhoeddus.
- Ymestyn awdurdodaeth yr Ombwdsmon i gynnwys darparwyr gofal iechyd preifat (mewn rhai amgylchiadau).

Cadarnhaodd y Swyddog Monitro y byddai'r ymgynghoriad yn cau 18 Ionawr, 2016.

Gallai effaith y cynigion ar Bwyllgorau Safonau arwain at hyd yn oed mwy o ddibyniaeth ar Swyddogion Monitro a Phwyllgorau Safonau i orfodi'r Cod Ymddygiad, a llai o ddibyniaeth ar Swyddfa Ombwdsmon Gwasanaethau Cyhoeddus Cymru o ystyried diffyg cyfeiriad penodol, neu gyfeiriad penodol annigonol at God Ymddygiad Aelodau a Rôl Ombwdsmon Gwasanaethau Cyhoeddus Cymru, yn y Bil. Gwahoddwyd yr aelodau i roi sylwadau ar Atodiad 3, a'r sylwadau ynddo ar God Ymddygiad Aelodau.

Gwnaeth Aelodau'r Pwyllgor arsylwadau ac fe wnaethant fynegi'r safbwyntiau a ganlyn mewn perthynas â Bil drafft Ombwdsmon Gwasanaethau Cyhoeddus Cymru:-

- Cyfeiriodd y Cadeirydd at Dudalen 28, Adran 12 (1) o'r Rhaglen "Efallai na fydd yr Ombwdsmon yn ymchwilio i fater o dan y Rhan hon os yw'r person a dramgwyddwyd, os o gwbl, gyda neu wedi cael:- (c) datrysiad ar ffurf achos mewn llys barn". Amlygodd yr angen am eglurder ac eglurodd, mewn achosion lle bu honiadau o athrod, gellid delio â materion o'r fath drwy weithredu wedyn am ddifenwi. Fodd bynnag, cyfeiriodd at faterion eraill y gellid eu codi na fyddai'n cael eu cwmpasu'n uniongyrchol gan gamau gweithredu dilynol.

Eglurodd y Swyddog Monitro mai Bil drafft oedd hwn ac amlinellodd y pwerau sydd gan Ombwdsmon Gwasanaethau Cyhoeddus Cymru o ran awdurdodaeth y Pwyllgor Safonau mewn perthynas â'r Cod Ymddygiad, fel y crybwyllir yn Neddf Llywodraeth Leol 2000. Hysbysodd yr Aelodau fod Deddf gyfredol Ombwdsmon Gwasanaethau Cyhoeddus Cymru 2005, a ddisodlir gan Fil Ombwdsmon Gwasanaethau Cyhoeddus Cymru, yn delio â'r awdurdodaeth mewn perthynas â chwynion am wasanaeth neu gamweinyddu, ac yn darparu gwahaniaeth rhwng dwy rôl Ombwdsmon Gwasanaethau Cyhoeddus Cymru. Hysbyswyd yr Aelodau bod yr

adroddiad yn amlinellu gwaith Ombwdsmon Gwasanaethau Cyhoeddus Cymru, ac yn amlygu canlyniadau dilynol posibl y Bil. Cyfeiriodd y Swyddog Monitro at ganfyddiadau'r ymchwiliad, a gynhaliwyd gan Bwyllgor Cyllid y Cynulliad Cenedlaethol, ar waith Ombwdsmon Gwasanaethau Cyhoeddus Cymru.

Amlinellwyd y pryderon a godwyd gan Gymdeithas Llywodraeth Leol Cymru yn Atodiad 3 i'r adroddiad gan y Swyddog Monitro, a chyfeiriwyd yn arbennig at y canlyniadau posibl sy'n deillio o ddiffyg cyllid ac adnoddau ychwanegol, yr effaith ar faterion Cod Ymddygiad ac o bosibl, encilio o'r agenda moesegol. Mynegodd y Swyddog Monitro bryder hefyd mewn perthynas ag unrhyw ganlyniadau anfwriadol yn deillio o'r ddeddfwriaeth.

Mynegodd y Cadeirydd y farn y gallai cynnydd mewn pwerau, gyda chyllideb sefydlog ac adnoddau staff, gael effaith andwyol gyda phenderfyniadau yn gorfod cael eu gwneud ar ba gwynion i ymchwilio iddynt a'r angen am flaenoriaethu. Teimlai'r Cadeirydd y gellid rhoi ystyriaeth i gynyddu pwerau Pwyllgorau Safonau, o dan gyfyngiadau o'r fath.

Cyfeiriodd y Cadeirydd at Dudalen 46, Adran 44, (1) Y bobl sydd â hawl i gwyno wrth yr Ombwdsmon yw:- (a) aelod o'r cyhoedd sy'n honni neu wedi honni iddynt gael anghyfiawnder neu galedi". Teimlai fod angen eglurhad gyda diffiniad o ran "anghyfiawnder a chaledi". Cyfeiriodd yn benodol at y diffyg manylder ac ansicrwydd yn ymwneud â'r bobl sydd â hawl i gwyno.

Tynnodd yr Aelod Annibynnol J. Hughes (JH) sylw at y materion canlynol:-

- Dylid cael ymrwymiad o fewn y fframwaith, i ddarparu ymatebion electronig neu gopi caled.
- Mynegwyd cefnogaeth i Ombwdsmon Gwasanaeth Cyhoeddus Cymru gael pwerau i adennill y costau wrth ymchwilio i wasanaethau iechyd preifat, a fyddai'n helpu i gynorthwyo wrth ddilyn ymchwiliadau drwy'r broses gyfan.
- Cyfeiriwyd at Dudalen 91, Tystiolaeth Cymdeithas Llywodraeth Leol Cymru. Tynnwyd sylw at bwysigrwydd y broses a'r angen i sicrhau bod ymgynghori ac ymgysylltu effeithiol. Pwysleisiwyd na ddylid rhuthro'r broses i gyd-fynd ag Etholiadau Llywodraeth Cynulliad Cymru, a allai arwain at ganlyniadau anfwriadol. Teimlwyd y dylid archwilio pob opsiwn arall yn llawn, gan gynnwys unrhyw feysydd problemus posibl.
- Codwyd pryderon mewn perthynas â'r paragraff olaf ar dudalen 96. Cyfeiriwyd at y newidiadau posibl yn rôl Ombwdsmon Gwasanaethau Cyhoeddus Cymru a'r effeithiau dilynol sy'n codi o uno Awdurdodau. Roedd Aelodau'r Pwyllgor yn cefnogi'r farn y gallai diwygio rôl Ombwdsmon Gwasanaethau Cyhoeddus Cymru, a gwanhau'r Cod yn dilyn hynny, fod yn gam yn ôl ar adeg pan gynigiwyd diwygio mor ddadleuol o fewn Llywodraeth Leol. Mynegwyd y farn y gallai pwerau cynyddol arfaethedig i Ombwdsmon Gwasanaethau Cyhoeddus Cymru gael effaith andwyol ar ganfyddiad y cyhoedd mewn perthynas â chwynion yn erbyn Cynghorwyr.

Teimlai'r Aelod Annibynnol A. Mellor y gallai cryfhau rôl Ombwdsmon Gwasanaethau Cyhoeddus Cymru roi hyder i achwynwyr a allai werthfawrogi cyfranogiad corff annibynnol, yn enwedig o ran cwynion sy'n ymwneud â'r GIG. Fodd bynnag, roedd hi hefyd yn cydnabod effaith niweidiol bosibl ar y Cod Ymddygiad.

Mewn ymateb i gais gan y Cadeirydd i ystyried pryderon a godwyd gan y Pwyllgor, esboniodd y Swyddogion Monitro y byddai'r materion a godwyd yn cael eu cyfleu i'r swyddog sy'n llunio ymateb y Cyngor, er mwyn galluogi cynnwys barn y Pwyllgor yn ymateb ehangach y Cyngor.

Mynegodd Paula White, Aelod Annibynnol, ei gwerthfawrogiad am y gwaith a wnaed gan y Dirprwy Swyddog Monitro.

Yn dilyn trafodaeth bellach:-

PENDERFYNWYD -bod y Pwyllgor Safonau yn cytuno cyfleu'r materion a godwyd gan yr Aelodau i'r swyddog sy'n llunio ymateb y Cyngor, er mwyn galluogi cynnwys barn y Pwyllgor yn ymateb ehangach y Cyngor.

(GW i Weithredu)

8 ADRODDIAD AR GYNHADLEDD SAFONAU CYMRU GYFAN

Esboniodd y Cadeirydd ei fod wedi mynychu Cynhadledd Safonau Cymru Gyfan, a rhoddodd grynodeb manwl o'r cyfarfod a oedd yn cynnwys y pwyntiau amlwg a ganlyn:-

- Roedd y Gynhadledd wedi cael nifer dda yn mynychu gyda chynrychiolwyr o bob Awdurdod Lleol yng Nghymru, gan gynnwys llawer o gynrychiolwyr o Gynghorau Dinas, Tref a Chymuned.
- Rhoddwyd areithiau allweddol gan:-
- Nick Bennett (Ombwdsmon Gwasanaethau Cyhoeddus Cymru)
- Lyn Cadwallader (Prif Weithredwr Un Llais Cymru)
- Peter Davies (Llywydd rhan-amser Panel Dyfarnu Cymru)
- Jan Williams (Comisiynydd Cwynion Annibynnol yr Heddlu Cymru)

Cyfeiriodd y Cadeirydd at yr araith gan Ombwdsmon Gwasanaethau Cyhoeddus Cymru o'r enw "A yw Egwyddorion Nolan yn Addas i'r Diben". Cyfeiriwyd at y posibilrwydd efallai nad ydynt yn addas i'r diben oherwydd nad oedd digon o sylw wedi cael ei roi i arweinyddiaeth, a oedd wedi awgrymu y dylid cael hyfforddiant mwy effeithiol i arweinwyr. Cyfeiriodd y Cadeirydd at y gwaith rhagorol sy'n cael ei wneud yn hyn o beth gan Swyddog Monitro a Dirprwy Swyddog Monitro Sir Ddinbych.

- O'r 231 o gwynion a ddaeth i lawr gan Ombwdsmon Gwasanaethau Cyhoeddus Cymru dros gyfnod o ddeuddeg mis, a oedd yn ymwneud â materion Cod Ymddygiad, dim ond 17 oedd wedi cael eu nodi fel toriad posibl, gyda dim ond 8 yn cael eu cyfeirio at y Pwyllgorau Safonau.
- Roedd mater a nifer y cwynion blinderus wedi cael eu hamlygu gan Ombwdsmon Gwasanaethau Cyhoeddus Cymru. Cyfeiriwyd at ganfyddiad y cyhoedd ynglŷn â dosbarthiad y cwynion yn flinderus, a'r cyhoedd yn mabwysiadu agwedd negyddol o bosibl ynghylch cyflwyno cwynion.

- Y farn a fynegwyd gan Ombwdsmon Gwasanaethau Cyhoeddus Cymru yw y gallai cwynion blinderus godi o ganlyniad i arweinyddiaeth wael.

Cyfeiriodd y Cadeirydd at y dewis o weithdai sydd ar gael ac eglurodd ei fod wedi mynychu gweithdai "Codi Pryderon" a "Llywodraethu a Safonau". Cyfeiriodd at nifer o Awdurdodau yng Nghymru, gan gynnwys Caerdydd a Wrecsam, sydd wedi enwi eu Pwyllgorau Safonau yn "Bwyllgor Safonau a Moeseg". Esboniodd y Cadeirydd fod Pwyllgor Safonau a Moeseg Caerdydd wedi cael cyfrifoldeb i oruchwylio a monitro gweithdrefn codi pryderon y Cyngor, ac yn ystyried unrhyw faterion moesegol sy'n codi o hynny. Teimlai'r Cadeirydd y gallai'r dull hwn gael ei ystyried gan y Cyngor Llawn i'w fabwysiadu yn Sir Ddinbych.

Mewn ymateb i gwestiwn gan y Cadeirydd, rhoddodd y Swyddog Monitro fanylion y gyfraith sy'n rheoli diogelu gweithwyr sydd wedi codi pryderon neu gyflwyno cwyn. Cyfeiriodd at arwyddocâd cwynion neu honiadau o'r fath sydd wedi'u codi'n ddidwyll, a chadarnhaodd nad oedd unrhyw gyfyngiadau amser ynghylch y cyfnod diogelu.

Cylchredwyd deunydd darllen a ddarparwyd gan y Cadeirydd, a oedd wedi'u dosbarthu yn y gweithdy ar Safonau Llywodraethu Cynghorau Tref a Chymuned, i Aelodau'r Pwyllgor.

PENDERFYNWYD - bod y Pwyllgor Safonau yn derbyn a nodi cynnwys adroddiad y Cadeirydd ynglŷn â Chyfarfod Cynhadledd Safonau Cymru Gyfan.

9 HYGYRCHEDD GWYBODAETH GAN GYNGHORAU DINAS, TREF A CHYMUNED.

Cyfeiriodd yr Aelod Annibynnol Julia Hughes (JH) at drafodaethau blaenorol ar gyhoeddi gwybodaeth yn electronig gan Gynghorau Dinas, Tref a Chymuned, a oedd bellach yn orfodol, a thynnodd sylw at yr angen am fynediad rhwydd at wybodaeth.

Fel y cytunwyd yn y cyfarfod blaenorol, rhoddodd grynodeb manwl o'r archwiliad gwirfoddol i asesu hygyrchedd gwybodaeth oddi ar wefannau'r pedwar ar bymtheg Cyngor Dinas, Tref a Chymuned canlynol:-

Dinbych, Derwen, Dyserth, Efenechtyd, Gwyddelwern, Henllan, Llanarmon-yn-lâl, Llanbedr Dyffryn Clwyd, Llandegla, Llandrillo, Llandyrnog, Llanelidan, Llanfair Dyffryn Clwyd, Llanferres, Llangollen, Llangynhafal, Llanrhaeadr, Llantysilio, Llanynys.

Roedd y meysydd a archwiliwyd a'r wybodaeth a geisiwyd o'r archwiliad yn cynnwys:-

- Darparu gwefan a'i hygyrchedd
- Argaeledd cofnodion y cyfarfod blaenorol
- Manylion am ddyddiad, amser a lleoliad y cyfarfod nesaf, a mynediad i raglenni
- Darparu diweddariadau rheolaidd

Esboniodd JH, o ran meysydd i'w harchwilio, ei bod wedi cynnwys colofnau yn ymwneud â darpariaeth ddwyieithog ac unrhyw nodiadau cyffredinol.

Amlygwyd y pwyntiau amlycaf canlynol o'r arfer a gynhaliwyd:-

- Roedd nifer o Gynghorau yn methu â chwrdd â'r gofynion gorfodol. Fodd bynnag, nodwyd bod yna hefyd lawer o arferion da.
- Dylid ei wneud yn ofynnol i bob Cyngor Dinas, Tref a Chymuned gael dolen i wefan Sir Ddinbych.
- Roedd rhai gwefannau wedi dyddio.
- Yr angen i sicrhau bod dolenni cyswllt gwefan yn gywir ac yn gyfredol.
- Nid oedd pob gwefan yn ddwyieithog, a thynnwyd sylw at bwysigrwydd darparu gwybodaeth ddwyieithog.
- Mewn rhai achosion, roedd gwefannau ar hyn o bryd yn cael eu datblygu.
- Roedd gan nifer o wefannau raglenni safonol nad oedd byth yn newid.
- Mewn rhai achosion, darparwyd ffurflenni electronig at ddibenion cyswllt.
- Roedd buddion drwy ddarparu sawl dull o gael gafael ar wybodaeth.
- Mae angen nodi fod llawer o'r gwaith yn cael ei wneud ar sail wirfoddol.
- Yr angen am eglurhad mewn perthynas â'r ddarpariaeth orfodol a dymunol.
- Ystyriaeth yn cael ei rhoi i ddarparu hyfforddiant i Glercod Cynghorau Dinas, Tref a Chymuned.

Hysbyswyd y Pwyllgor gan y Swyddog Monitro bod pryderon wedi cael eu codi ynghylch argaeledd adnoddau o ran darparu a chynnal a chadw gwefannau, yn enwedig ar gyfer Cynghorau gwledig llai. Eglurodd hefyd fod Clercod i Gynghorau Dinas, Tref a Chymuned ond yn cael eu talu am nifer penodol o oriau.

Gyda chymeradwyaeth Aelodau, cytunodd y Swyddog Monitro y dylai canlyniad y trafodaethau mewn perthynas â'r archwiliad a gynhaliwyd gan JH gael ei anfon ymlaen at y Rheolwr Cynhwysiant Cymunedol, a gysylltodd yn rheolaidd gyda Chynghorau Dinas, Tref a Chymuned ac a allai gyfleu'r farn a fynegwyd i'r Cynghorau perthnasol, ac fe allai'r materion gael sylw ar sail lled uchelgeisiol. Amlinellodd gylch gwaith y Pwyllgor Safonau hefyd ac esboniodd nad oedd y materion a drafodwyd o angenrheidrwydd yn faterion Cod Ymddygiad.

Cyfeiriodd y Cynghorydd W.L. Cowie at y dull a fabwysiadwyd gan Gyngor Dinas Llanelwy o ran y polisi dwyieithog a darparu gwybodaeth drwy gyfrwng y Gymraeg. Ar gais y Cynghorydd Cowie, cytunodd y Swyddog Monitro i gysylltu â'r Swyddogion Arweiniol ar Safonau'r Gymraeg yn Sir Ddinbych, gyda'r bwriad o gysylltu a darparu cyngor a chymorth i Glerc Cyngor Dinas Llanelwy mewn perthynas â Pholisi'r Gymraeg.

Tynnwyd sylw at bwysigrwydd pennu'r amserlenni, gan y byddai'r wybodaeth a ddarperir yn yr adroddiad yn gallu newid ar ôl ei gynhyrchu.

Diolchodd y Cadeirydd i JH am y gwaith caled a wnaed i gynhyrchu'r wybodaeth, a fyddai'n galluogi'r Pwyllgor i fonitro tueddiadau a gwneud awgrymiadau ar gyfer gwelliannau yn y dyfodol. Cadarnhaodd y Pwyllgor y byddai'n derbyn adroddiad

diweddaru ysgrifenedig yn ei gyfarfod ym mis Mawrth, 2016, fel y nodir yn rhaglen gwaith i'r dyfodol y Pwyllgor.

PENDERFYNWYD - bod y Pwyllgor Safonau:-

- (a) yn derbyn ac yn nodi cynnwys yr adroddiad a gyflwynwyd gan yr Aelod Annibynnol Julia Hughes,
- (b) yn anfon canlyniad y trafodaethau mewn perthynas â'r archwiliad ymlaen at y Rheolwr Cynhwysiant Cymunedol, a fyddai'n cael cais i ymgysylltu â'r Cynghorau Dinas. Tref a Chymuned ynghylch y materion a godwyd.
- (c) y Swyddog Monitro i gysylltu â'r Swyddogion Arweiniol ar Safonau'r Gymraeg yn Sir Ddinbych, gyda'r bwriad o gysylltu a darparu cyngor a chymorth i Glerc Cyngor Dinas Llanelwy mewn perthynas â Pholisi'r Gymraeg.
- (d) yn cyflwyno adroddiad diweddaru ysgrifenedig i gyfarfod Mawrth, 2016 o'r Pwyllgor.

(GW i Weithredu)

10 PRESENOLDEB MEWN CYFARFODYDD

Gwahoddwyd Aelodau'r Pwyllgor i roi adborth o gyfarfodydd Cynghorau Dinas, Tref a Chymuned y gwnaethant eu mynychu'n ddiweddar, ac fe gymerodd Aelodau'r cyfle i gynnig crynodeb o sut roedd y Cynghorau perthnasol wedi gweithredu.

Roedd y Cynghorydd W.L. Cowie wedi mynychu cyfarfod Cyngor Dinas Llanelwy. Cadarnhaodd nad oedd wedi dod ar draws unrhyw broblemau ac yn fodlon ar y modd y cynhaliwyd y cyfarfod.

Mynychodd yr Aelodau Annibynnol Anne Mellor (AM) a Julia Hughes (JH) gyfarfod Cyngor Cymuned Aberchwiler 7 Hydref, 2015 ac fe dynnwyd sylw at y materion canlynol:-

- Nid oedd unrhyw feysydd o bryder wedi cael eu nodi, a chadarnhaodd y ddau Aelod Annibynnol eu bod yn fodlon ar y modd y cynhaliwyd y cyfarfod.
- Roedd cynrychiolydd y Cyngor Sir lleol wedi mynychu'r cyfarfod ac wedi bod yn hynod ddefnyddiol ac yn llawn gwybodaeth.

Mynychodd yr Aelod Annibynnol Julia Hughes (JH) gyfarfod y Cyngor Sir Arbennig ar 7 Hydref, 2015, a oedd a nifer dda yn bresennol, ac ni chodwyd unrhyw faterion o bryder.

Mynychodd yr Aelod Annibynnol Julia Hughes (JH) y cyfarfodydd canlynol ac amlygwyd y pwyntiau a'r materion canlynol:-

Cyngor Tref Bodelwyddan:-

- Ni fu unrhyw wybodaeth ar gael ar y wefan, a oedd angen ei diweddaru. Fodd bynnag, roedd y Clerc wedi bod yn ymatebol iawn, yn gadarnhaol ac yn ddefnyddiol.
- Rhoddwyd cefnogaeth gan y Clerc i'r Cadeirydd a'r Aelodau yn ystod y cyfarfod.

Cafwyd trosolwg gan JH a oedd yn cynnwys y meysydd canlynol o bryder:-

- Nid oedd y cyfarfod wedi'i reoli yn y modd mwyaf effeithiol
- Nid oedd yr aelodau'n parchu Cadeirydd y cyd Gynghorwyr yn ystod achosion. Cafodd hyn effaith andwyol ar reolaeth y cyfarfod, ac nid oedd Aelodau wedi cymhwyso protocol a oedd yn briodol i gyfrifoldeb eu swyddi fel Cynghorwyr Tref a chynrychiolwyr preswylwyr lleol.
- Cydnabuwyd yr anhawster wrth recriwtio pobl i'r swyddi gwirfoddol, gyda dim ond un cais wedi dod i law ar gyfer dwy swydd wag yn y Cyngor Tref.

Yn dilyn trafodaeth fer, gofynnodd y Pwyllgor bod y Swyddog Monitro'n ysgrifennu at Glerc Cyngor Cymuned Bodelwyddan yn tynnu sylw at y materion a godwyd gan yr Aelod Annibynnol J. Hughes, a'r pryderon dilynol a fynegwyd gan y Pwyllgor Safonau ar ôl ystyried ei hadroddiad.

Cyngor Cymuned Nantglyn - 6 Hydref, 2015:-

- Rhoddwyd enghraifft dda o sut y dylai cyfarfod redeg.
- Roedd gan y Cyngor wefan dda, yn llawn gwybodaeth ac wedi'i dylunio'n dda.
- Anfonwyd e-bost at y Clerc er mwyn cael manylion y lleoliad, y rhaglen a chofnodion y cyfarfod nesaf.
- Roedd Clerc y Cyngor wedi bod yn ddefnyddiol iawn.
- Ni wiriodd y Clerc Aelod o'r Cyngor a gyrhaeddodd yn hwyr, mewn perthynas â Datganiadau o Gysylltiad.
- Cymerodd Aelodau'r Cyngor ran yn holl eitemau'r rhaglen mewn modd cefnogol ac adeiladol.
- Rheolodd y Cadeirydd y cyfarfod yn dda ac fe gafodd ei gefnogi gan y Clerc, a chynhaliwyd busnes mewn modd trefnus a chywir.
- Roedd Ceisiadau Cynllunio wedi cael sylw yn ofalus ac wedi'u trin yn briodol.
- Cyflwynodd y Cadeirydd JH i Aelodau'r Cyngor a dywedodd ei fod yn synnu nad oedd wedi cael gwybod am fwriad y Pwyllgor Safonau. Amlinellodd JH ddiben yr ymweliadau i ddarparu cefnogaeth i'r Cynghorau perthnasol, a chodi proffil y Pwyllgor Safonau i osgoi bod yn gorff o bell.
- Rhoddwyd adborth am drafodion y cyfarfod gan JH.

Hysbyswyd y Pwyllgor bod Aelod o'r Cyngor Cymuned wedi mynegi'r farn nad oedd wedi cael gwybod am fwriadau'r Pwyllgor Safonau, ac yn dilyn trafodaethau gyda'r Swyddog Monitro, ei ddealltwriaeth ef oedd bod rôl y Pwyllgor Safonau'n gysylltiedig â materion sy'n ymwneud â'r Cod Ymddygiad ac nid materion Llywodraethu. Mewn ymateb i bryderon a fynegwyd ynghylch cwestiynau yn ymwneud â'r wefan, cyfeiriodd JH at gyflwyniad y ddeddfwriaeth ddiweddar.

Hysbyswyd yr Aelodau, yn dilyn y cyfarfod, bod JH wedi anfon e-bost at Glerc y Cyngor Cymuned, a ddarllenwyd yn y cyfarfod, yn crynhoi rôl, cylch gorchwyl a phrif ddyletswyddau'r Pwyllgor Safonau.

Mewn ymateb i bryderon a godwyd gan Aelodau'r Pwyllgor Safonau ynghylch Cynghorau Dinas, Tref a Chymuned yn dod yn ymwybodol o rôl y Pwyllgor Safonau, cytunodd y Swyddog Monitro i ysgrifennu at Gynghorau Dinas, Tref a Chymuned yn amgáu manylion rôl a chylch gorchwyl y Pwyllgor Safonau. Rhoddodd y Swyddog Monitro gadarnhad nad oedd yn orfodol i Aelodau'r Pwyllgor Safonau ymweld neu fynychu cyfarfodydd Cynghorau Dinas, Tref a Chymuned.

Hysbysodd y Cadeirydd y Pwyllgor ei fod ef a'r Swyddog Monitro wedi mynychu cyfarfod Clercod a Chynghorwyr Cymuned 25 Tachwedd, 2015. Rhoddwyd manylion ynghylch yr eitemau a drafodwyd, a'r gweithrediadau, gan y Cadeirydd. Cafwyd cadarnhad bod y cyfarfod wedi ei gadeirio'n dda ac ni chodwyd unrhyw faterion o bryder.

Hysbysodd y Swyddog Monitro'r Pwyllgor fod cwestiynau am ddarparu hyfforddiant, a darparwyr hyfforddiant, wedi cael eu codi o ran y posibilrwydd o ddyblygu darpariaeth. Eglurodd fod y Prif Weithredwr wedi gofyn am gael archwilio'r rhaglen gyfannol o hyfforddiant, ar y cyd â darparwyr eraill. Cyfeiriodd y Swyddog Monitro at ei gyfrifoldeb o ran lefel, amlder a safon y ddarpariaeth o hyfforddiant mewn perthynas â'r Cod Ymddygiad. Mynegodd y Cadeirydd y farn bod y lefel o hyfforddiant a ddarperir gan Sir Ddinbych wedi bod yn ardderchog.

Yn dilyn trafodaeth bellach:-

PENDERFYNWYD - bod y Pwyllgor Safonau:-

- (a) yn derbyn ac yn nodi adborth a gyflwynwyd o gyfarfodydd diweddar a fynychwyd gan Aelodau'r Pwyllgor.
- (b) yn gofyn i'r Swyddog Monitro ysgrifennu at Gynghorau Dinas, Tref a Chymuned yn amgáu manylion rôl a chylch gorchwyl y Pwyllgor Safonau, a
- (c) chytuno bod y Swyddog Monitro yn ysgrifennu at Glerc Cyngor Tref Bodelwyddan yn tynnu sylw at y materion a godwyd gan yr Aelod Annibynnol J. Hughes, a'r pryderon dilynol a fynegwyd gan y Pwyllgor Safonau.

(GW i Weithredu)

11 DYDDIAD Y CYFARFOD NESAF

Nododd yr aelodau amser, dyddiad a lleoliad y cyfarfod nesaf o'r Pwyllgor Safonau, a:-

PHENDERFYNWYD - bod y cyfarfod nesaf o'r Pwyllgor Safonau yn cael ei gynnal ddydd Gwener 4 Mawrth, 2015 yn Ystafell Gynadledda 1a, Neuadd y Sir, Rhuthun.

GWAHARDD Y WASG A'R CYHOEDD

PENDERFYNWYD - o dan Adran 100A o Ddeddf Llywodraeth Leol 1972, bod y Wasg a'r Cyhoedd i'w gwahardd o'r cyfarfod ar gyfer yr eitem(au) busnes canlynol ar y sail eu bod yn debygol o olygu datgelu gwybodaeth eithriedig fel y'i diffinnir ym Mharagraffau 12 a 13 o Ran 4 Atodlen 12A, Deddf Llywodraeth Leol 1972.

RHAN II

12 COD YMDDYGIAD - RHAN 3 DEDDF LLYWODRAETH LEOL 2000

Roedd copi o adroddiad cyfrinachol gan y Swyddog Monitro, a oedd yn rhoi trosolwg o gwynion a gyflwynwyd i Ombwdsmon Gwasanaethau Cyhoeddus Cymru, wedi ei gylchredeg gyda'r papurau ar gyfer y cyfarfod.

Roedd y Pwyllgor Safonau eisoes wedi gofyn am gael gwybod yn rheolaidd am lefel y cwynion a gyflwynwyd i Ombwdsmon Gwasanaethau Cyhoeddus Cymru. Roedd y tablau yn Atodiad 1 yn rhoi trosolwg o gwynion a gyflwynwyd ers 1 Ebrill 2015.

Cadarnhaodd y Dirprwy Swyddog Monitro nad oedd yr un o'r naw achos y cyfeiriwyd atynt yn yr adroddiad wedi cael eu dilyn gan Ombwdsmon Gwasanaethau Cyhoeddus Cymru.

PENDERFYNWYD - bod y Pwyllgor Safonau yn derbyn ac yn nodi cynnwys yr adroddiad.

Daeth y cyfarfod i ben am 12.50 p.m.

RHAGLEN GWAITH I'R DYFODOL Y PWYLLGOR SAFONAU

SYLWER BOD Y RHAGLEN I'W DOSBARTHU 1 WYTHNOS CYN Y CYFARFOD

DYDDIAD Y CYFARFOD	EITEMAU ADRODDIAD/MEYSYDD	AWDUR YR ADRODDIAD
24 Mehefin 2016	Eitem Sefydlog: Presenoldeb mewn Cyfarfodydd	Amh.
	Eitem Sefydlog Rhan 2: Trosolwg o Gwynion yn Sir Ddinbych	Gary Williams (Swyddog Monitro)
16 Medi 2016	Eitem Sefydlog: Presenoldeb Mewn Cyfarfodydd	Amh.
	Eitem Sefydlog Rhan 2: Trosolwg o Gwynion yn Sir Ddinbych	Gary Williams (Swyddog Monitro)
2 Rhagfyr 2016	Eitem Sefydlog: Presenoldeb Mewn Cyfarfodydd	Amh.
	Eitem Sefydlog Rhan 2: Trosolwg o Gwynion yn Sir Ddinbych	Gary Williams (Swyddog Monitro)

Diweddarwyd 10/04/2015 - SLW

Mae tudalen hwn yn fwriadol wag

Eitem Agenda 6

Adroddiad i'r: Pwyllgor Safonau

Dyddiad y Cyfarfod: 4 Mawrth 2016

Aelod / Swyddog Arweiniol: Ian Trigger, Cadeirydd y Pwyllgor Safonau

Awdur yr Adroddiad: Lisa Jones - Dirprwy Swyddog Monitro

Teitl: Adroddiad Blynyddol y Cadeirydd 2015/16

1. Am beth mae'r adroddiad yn sôn?

Mae'r adroddiad hwn i'r Pwyllgor Safonau i alluogi'r Pwyllgor llawn i gael golwg a chyfrannu at Adroddiad Blynyddol y Cadeirydd, cyn i'r eitem gael ei gyflwyno i'r cyngor llawn. Mae'r adroddiad drafft ynghlwm fel Atodiad 1.

2. Beth yw'r rheswm dros lunio'r adroddiad hwn?

Cytunodd Aelodau'r Pwyllgor hwn y dylai Adroddiad Blynyddol gael ei gyflwyno ar waith y pwyllgor a'i ganfyddiadau a'i arsylwadau, at holl Aelodau'r Cyngor fel rhan o ymgyrch y Pwyllgor i wella safonau ymddygiad moesegol ac i gydymffurfio â Chod Ymddygiad Aelodau.

3. Beth yw'r Argymhellion?

Bod yr Aelodau yn:

- (a) Nodi cynnwys yr adroddiad.
- (b) Darparu sylwadau i'r Swyddog Monitro ar gynnwys yr adroddiad i sicrhau ei fod yn adlewyrchu barn y Pwyllgor yn gywir.
- (c) Argymell y dylai'r Cadeirydd gyflwyno'r adroddiad i'r Cyngor Llawn.

4. Manylion yr Adroddiad

Mae'r Adroddiad ynghlwm fel Atodiad 1 i'w ystyried.

5. Sut mae'r penderfyniad yn cyfrannu at y Blaenoriaethau Corfforaethol?

Mae Pwyllgor Safonau sy'n gwbl weithredol a chynrychioliadol sy'n cynnal y safonau uchel a ddisgwylir gan aelodau yn cynorthwyo i ategu at weithredu swyddogaethau democrataidd y Cyngor.

6. Faint fydd hyn yn ei gostio a sut y bydd yn effeithio ar wasanaethau eraill?

Nid oes unrhyw oblygiadau ar gyfer gwasanaethau eraill o ganlyniad i'r adroddiad hwn.

7. Pa ymgynghori a wnaed?

Nid oes unrhyw ymgynghoriadau wedi eu cynnal ac eithrio ar gyfer yr adroddiad hwn i'r Pwyllgor Safonau, fel ymgynghorai, i gael adborth ar ei briodoldeb, cynnwys ac argymhellion.

8. Datganiad y Prif Swyddog Cyllid

Dim angen.

9. Pa risgiau sydd ac a oes unrhyw beth y gallwn ei wneud i'w lleihau?

Nid oes unrhyw risgiau a nodwyd.

10. Pŵer i wneud y Penderfyniad

Deddf Llywodraeth Leol 2000; Rheoliadau Pwyllgorau Safonau (Cymru) 2001 a Rheoliadau Pwyllgor Safonau (Cymru) (Diwygio) 2006.

Appendix 1 DRAFT REPORT FOR STANDARDS COMMITTEE CONSIDERATION

Agenda Item No.

Report To: Full Council

Date of Meeting: to be confirmed (proposed as May 2016)

Lead Member / Officer: Ian Trigger, Chair of Standards Committee

Report Author: Lisa Jones – Deputy Monitoring Officer

Title: Standards Committee Annual Report

1. What is the report about?

This is the second Annual Report of Standards Committee to the Full Council and covers the calendar year January to December 2015 only. It was agreed that the Chair will present their report on this basis annually to the Full Council, in order to keep Members informed of trends; issues in respect of compliance with the Members Code of Conduct generally across the County and the work of the Committee in driving up standards of behaviour at the County level, but also at Town, City and Community levels.

2. What is the reason for making this report?

It was agreed by Standards Committee that an Annual Report should be presented on the work of the committee and its findings and observations, to all Council Members as part of the Committee's drive to increase standards of ethical behaviour and compliance with the Members Code of Conduct.

3. What are the Recommendations?

That Members note the contents of the report.

4. Report details

4.1 Standards Committee main role is to monitor adherence to the Members Code of Conduct. All members are aware that their Code is founded (and should be read in conjunction with) the 7 Nolan Principles of Public Life. In Wales there are ten principles (those which were not included in the Nolan recommendations indicated with an *) as follows: -

Selflessness Honesty Integrity and propriety
Duty to uphold law *
Stewardship *
Objectivity in decision making
Equality and respect *
Openness
Accountability
Leadership

4.2 Standards Committee consists of the following Member types – 2 County Councillors, 4 Independent (co-opted) Members, and 1 Community Council Member (who is not also a twin hatted Member). The majority of the Members are therefore not elected, but are recruited from members of the public as per the requirements of the Standards legislation in Wales. The Committee can also only be quorate when at least half of those Members present are the independents.

Independent members are recruited via a public advert, which is open to all save for strict criteria in respect of previously being an Officer or Member of the Council within certain time frames, being of good standing and having certain attributes as laid down in the legislation. Other than this the opportunity is open to persons from all walks of life, in order to represent the views of the public with regard to the standards they expect of their elected members; and who will also then if required to do so, sit in judgement in respect of any references to the Committee from the Public Services Ombudsman for Wales, where there has been a breach of the Code of Conduct, which falls above the relevant threshold. The decision to investigate a breach, is at the discretion of the Ombudsman; based on the facts, evidence and nature of the breach and whether there is corroborative evidence or not, and if there is a public interest in exhausting time and public financial resources in the investigation and hearing itself; similar to the Crown Prosecution Service (CPS) approach to charges and prosecutions.

The Committee which receives a reference from the Public Services Ombudsman for Wales following his investigation into the complaint, will then sit in a quasi-judicial capacity whist they hear the matter. They have powers to suspend a Member from Office for a maximum of 6 months, during which time they will not be able to act in their capacity as an elected member and will not receive any member salary. The Committee has discretion to impose lesser sanctions such as a partial suspension from duties or a 'public censure' and to impose other conditions such as attendance at training.

There is also the option where the PSOW seeks the views of the Monitoring Officer and Standards Committee on whether in those circumstances where the Ombudsman decides not to investigate, that the Monitoring Officer may wish to investigate locally. In these circumstances, the Monitoring Officer consults the Chair and/or the Committee and each case is considered on its merits.

4.3 During the past year the Committee met on 5 occasions and no meeting was cancelled. The table below sets out a summary of the items under discussion: -

Date of Meeting	Report Items/Area s
23/1/15	Standing Item: Attendance at meetings Forward Work Programme Draft Chairs Annual Report PSOW Public Interest Test North Wales Standards Committee Forum feedback. Standing Item: overview of complaints in Denbighshire against Members
6/3/15	Standing Item: Attendance at meetings Forward Work Programme Chairs Annual Report final for approval Rhyl Town Council Application for Dispensation Training Provision – Clerks and Councillors Standing Item: overview of complaints in Denbighshire against Members.
22/5/15	Standing Item: Attendance at meetings Forward Work Programme Feedback from the Chair on his Annual Report to Full Council PSOW –Revised Code of Conduct Guidance Declaration of Interests and Hospitality Feedback on the Training of Clerks event Training for Chairs and Vice Chairs of Town, City and Community Councils. Standing Item: overview of complaints in Denbighshire against Members.
18/9/15	Standing Item: Attendance at meetings Forward Work Programme PSOW Annual Report PSOW Code of Conduct Factsheets PSOW Code of Conduct Casebooks Consideration of a draft Self Regulatory Protocol for Town, City and Community Councils. Feedback from the All Wales Standards Conference in Cardiff. Standing Item: Overview of Complaints against Members.

4/12/15	Standing Item: Attendance at meetings
	Forward Work Programme
	Election of Vice Chair
	PSOW (Wales) Bill Consultation
	Accessibility of Information to the public of
	meetings of Town, City and Community Councils.
	Standing Item: Overview of Complaints against
	Members.

4.4 Standing Items

The 3 standing items the Committee receives is working well: -

(a) Reports from Standards Members in respect of their attendance and observations at Committee and Council meetings whether at County or Community level. Denbighshire County Council Standards Committee is fairly unique in it's proactive approach to raising standards and awareness of the Code of Conduct. Committee Members attend to observe conduct and general effectiveness of the meetings; will then feed back to the Committee, who may make recommendations in respect of any training needs or trends or patterns of conduct in particular communities or at County level. Previously the majority of these attendances were selected randomly and were routine observation exercises; occasionally a particular Council is selected at the request of the Monitoring Officer, whose jurisdiction in respect of the Code of Conduct extends to town, city and community councils.

This year the Committee has taken a slightly more strategic approach to its attendances based on feedback from clerks or members who wish to attempt to reach all councils across the County as equitably as possible, and if required offer any targeted support if requested.

On the whole, the general ethical framework functions are at a good level in the community and at county level; and this is reflected in the continued downturn in complaints to the PSOW.

As emphasised last year, the Committee is keen to make it clear that their approach in attending in person at meetings is from a support and educational angle; in order to target resources in order to improve standards and the public's confidence in the vital work being done at community level, on a voluntary basis; and not from any enforcement or critical angle. The Committee fully recognises the value such councils and their members add to local communities. The Committee believes that the Code is there to give public confidence, but to also protect members and any efforts to raise this awareness for all serving members is a key theme they continually endorse.

- (b) Overview of Complaints lodged against Members with the Public Services Ombudsman for Wales. The report is presented in a closed session and members are provided with a limited amount of detail in respect of the complaint but with sufficient detail to enable members to consider if training or other actions can be recommended to particular town city and community councils which are experiencing issues or an increase in complaints.
- (c) Forward Work Programme. This aligns the approach of other council committees and encourages a more strategic approach to the role of the Committee as a proactive one not just reactive to complaints.
- 4.5 The Committee received notices of retirement from two of its independent members namely Reverend Wayne Trigger (Vice Chair of Standards Committee) and also Margaret Medley and the Committee expressed their gratitude for their contributions.
- 4.6 Independent member Paula White has been elected as Vice Chair.
- 4.7 Two new independent members have been recruited, Julia Hughes and Anne Mellor.

5. How does the decision contribute to the Corporate Priorities?

A fully functioning and representative Standards Committee which upholds the high standards expected of members helps underpin the Council's exercise of its democratic functions.

6. What will it cost and how will it affect other services?

There are no implications for other services as a result of this report. .

7. What consultations have been carried out?

No consultations have been carried out save that Standards Committee has been consulted on the content.

8. Chief Finance Officer Statement

Not required.

9. What risks are there and is there anything we can do to reduce them?

There are no identified risks.

10. Power to make the Decision

The Local Government Act 2000; the Standards Committee (Wales) Regulations 2001 and the Standards Committee (Wales) (Amendment) Regulations 2006.



Eitem Agenda 8

Adroddiad i'r: Pwyllgor Safonau

Dyddiad y Cyfarfod: 4 Mawrth 2016

Aelod / Swyddog Arweiniol: Gary Williams, Swyddog Monitro

Awdur yr Adroddiad: Gary Williams, Swyddog Monitro

Teitl: Llyfr Achosion Cod Ymddygiad Ombwdsmon

Gwasanaethau Cyhoeddus Cymru

1. Am beth mae'r adroddiad yn sôn?

1.1 Mae'r adroddiad yn ymwneud â Llyfr Achosion Cod Ymddygiad gan Ombwdsmon Gwasanaethau Cyhoeddus Cymru (yr Ombwdsmon).

2. Beth yw'r rheswm dros lunio'r adroddiad hwn?

2.1 Hysbysu'r Pwyllgor o rifynnau diweddaraf Llyfr Achos Cod Ymddygiad yr Ombwdsmon.

3. Beth yw'r Argymhellion?

3.1 Bod aelodau'r Pwyllgor yn nodi'r wybodaeth sydd yn Llyfr Achos y Cod Ymddygiad.

4. Manylion yr adroddiad.

- 4.1 Ers 2013 mae'r Ombwdsmon wedi cynhyrchu Llyfr Achosion Cod Ymddygiad (y Llyfr Achosion). Am beth amser cyn hynny bu'r Ombwdsmon yn cynhyrchu llyfr achosion oedd yn ymwneud â'r cwynion y bu'n eu hymchwilio o safbwynt camweinyddu honedig gan gyrff cyhoeddus. Yn dilyn galwadau i ddefnyddio dull tebyg o safbwynt cwynion cod ymddygiad, dechreuodd yr Ombwdsmon gyhoeddi'r Llyfr Achosion yn 2013.
- 4.2 Bu'r Llyfr Achosion yn cael ei gyhoeddi ddwywaith y flwyddyn, fodd bynnag mae'r Ombwdsmon wedi penderfynu cynhyrchu'r llyfr achosion yn chwarterol o Ebrill 2015. Mae'r Llyfr Achosion yn cynnwys crynodeb o bob achos y mae'r Ombwdsmon wedi cwblhau ymchwiliad iddynt yn ystod y cyfnod perthnasol.
- 4.3 Lle mae'r achosion hynny wedi'u cyfeirio at Bwyllgor Safonau neu Banel Dyfarnu Cymru, bydd dolen yn cael ei darparu yn y fersiwn electronig o'r Llyfr Achosion i'r adroddiad llawn o ganlyniad yr achos a gynhyrchwyd gan y Pwyllgor neu'r Panel perthnasol.
- 4.4 Cynhyrchwyd y Llyfr Achos i gynorthwyo Aelodau ac eraill i ystyried a yw'r amgylchiadau y maent yn eu profi yn arwain at dorri'r Cod. Roedd hyn yn

- ychwanegiad i gyhoeddi esiamplau gwirioneddol yng Nghanllaw Ombwdsman Gwasanaethau Cyhoeddus Cymru ynglŷn â'r Cod.
- 4.5 Bydd y Llyfr Achosion yn caniatáu mynediad i Bwyllgorau Safonau awdurdod lleol i wybodaeth am y ffordd mae Pwyllgorau Safonau eraill yng Nghymru yn gosod sancsiynau a chael gwared ar achosion ac yn helpu i esbonio pam, mewn rhai achosion y gall yr Ombwdsmon wrthod ymchwilio toriadau honedig ar y sail nad yw honiadau blaenorol tebyg wedi arwain at sancsiwn.
- 4.6 Mae Atodiad 1 yr adroddiad hwn yn cynnwys y Llyfr Achosion a gyhoeddwyd yn Hydref 2015 ac yn delio â'r cyfnod rhwng Gorffennaf a Medi 2015. Dylai Aelodau sylwi bod y mwyafrif o gwynion a ymchwiliwyd yn ystod y cyfnod hwn yn ymwneud â datgelu a datgan diddordeb. O'r 6 cwyn a gofnodwyd yn ystod cyfnod cyhoeddi'r llyfr achos hwn, mae 4 yn ymwneud â datganiadau o ddiddordeb. O'r 6 ymchwiliad a gynhaliwyd, daethpwyd i'r canlyniad nad oedd digon o dystiolaeth o dorri cod mewn 5 achos, a chafodd un ei gyfeirio at Banel Dyfarnu Cymru. Canlyniad y mater a ystyriwyd gan Banel Dyfarnu Cymru oedd atal yr aelod hwnnw am dri mis.
- 4.7 Ni chyfeiriwyd achosion i'r Pwyllgor Safonau yn ystod y cyfnod hwn ac ni ddaeth unrhyw ymchwiliad i'r casgliad nad oedd angen gweithredu ymhellach er bod y cod wedi ei dorri.
- 4.8 Mae Atodiad 2 yr adroddiad hwn yn cynnwys y Llyfr Achosion a gyhoeddwyd yn Ionawr 2016 sy'n delio â'r cyfnod rhwng Hydref a Rhagfyr 2015. Dim ond at 3 ymchwiliad y cyfeirir yn y cyhoeddiad hwn. O'r rhain, daethpwyd i'r casgliad nad oedd tystiolaeth o dorri'r cod mewn un achos, a chanlyniad dau arall oedd nad oedd angen gweithredu ymhellach. Ni chyfeiriwyd unrhyw achos ychwaith at Bwyllgor Safonau na Phanel Dyfarnu Cymru.
- 4.9 Roedd yr achosion yn ymwneud â chwynion am gydraddoldeb a pharch, datganiad diddordeb, uniondeb cymeriad a phriodoldeb
- 5. Sut mae'r penderfyniad yn cyfrannu at y Blaenoriaethau Corfforaethol?
- 5.1 Nid yw'r penderfyniad yn cael effaith uniongyrchol ar y blaenoriaethau corfforaethol.
- 6. Faint fydd hyn yn ei gostio a sut y bydd yn effeithio ar wasanaethau eraill?
- 6.1 Nid oes costau uniongyrchol yn gysylltiedig â'r adroddiad hwn.
- 7. Beth yw prif gasgliadau'r Asesiad o Effaith ar Gydraddoldeb (AoEaG) a gynhaliwyd ar y penderfyniad? Dylai'r templed AoEaG wedi'i lenwi gael ei atodi fel atodiad i'r adroddiad.
- 7.1 Nid oes angen asesiad o'r effaith ar gydraddoldeb ar gyfer yr adroddiad hwn.
- 8. Pa ymgynghoriadau a gynhaliwyd gyda'r Pwyllgorau Archwilio ac eraill?
- 8.1 Nid yw'r mater hwn wedi ei adrodd nac wedi ei ymgynghori arno yn unrhyw le arall.

- 9. Datganiad y Prif Swyddog Cyllid
- 9.1 Does dim goblygiadau ariannol uniongyrchol yn yr adroddiad hwn.
- 10. Pa risgiau sydd ac a oes unrhyw beth y gallwn ei wneud i'w lleihau?
- 10.0 Nid oes unrhyw risgiau uniongyrchol sy'n gysylltiedig â'r adroddiad hwn.
- 11. Pŵer i wneud y Penderfyniad
- 11.1 Does dim angen gwneud penderfyniad.



The Code of Conduct Casebook

Issue 6 October 2015

Inside this issue

Introduction	1
--------------	---

Case Summaries

N 1		_
INIO	evidence of breach	4

(

Referred to standards	
committee	7

Referred to	
Adjudication Panel For Wales	8

More	Information	C

Introduction

The Public Services Ombudsman for Wales considers complaints that members of local authorities in Wales have broken the Code of Conduct. The Ombudsman investigates such complaints under the provisions of Part III of the Local Government Act 2000 and the relevant Orders made by the National Assembly for Wales under that Act.

Where the Ombudsman decides that a complaint should be investigated, there are four findings, set out under section 69 of the Local Government Act 2000, which the Ombudsman can arrive at:

- (a) that there is no evidence that there has been a breach of the authority's code of conduct;
- (b) that no action needs to be taken in respect of the matters that were subject to the investigation;
- (c) that the matter be referred to the authority's monitoring officer for consideration by the standards committee;
- (d) that the matter be referred to the President of the Adjudication Panel for Wales for adjudication by a tribunal (this generally happens in more serious cases).

(Continued overleaf)





In the circumstances of (c) and (d) above, the Ombudsman is required to submit the investigation report to the standards committee or a tribunal of the Adjudication Panel for Wales and it is for them to consider the evidence found by the Ombudsman, together with any defence put forward by the member concerned. It is also for them to determine whether a breach has occurred and, if so, what penalty (if any) should be imposed.

The Code of Conduct Casebook contains summaries of reports issued by this office for which the findings were one of the four set out above. However, in reference to (c) and (d) findings, The Code of Conduct Casebook only contains the summaries of those cases for which the hearings by the standards committee or Adjudication Panel for Wales have been concluded and the outcome of the hearing is known. This edition covers April to June 2015, but also includes the summaries of older cases for which the standards committee or Adjudication Panel hearings were concluded during this period.



Case Summaries

No evidence of breach

Pembrokeshire County Council – Accountability and openness Case reference 201404748 – Report issued in July 2015

It was alleged that a member of Pembrokeshire County Council ("the Councillor"), used his position improperly to influence and undermine the functions of a Cross Party Disciplinary Investigation Committee. In particular it was alleged that he used "pressure" and "manipulation" to dissuade another elected member from co-operating with this Committee.

The Ombudsman commenced an investigation to determine whether there was evidence to suggest that the Councillor had breached paragraphs 6(1)(a) and 7(a) of the Code of Conduct for elected members.

Evidence was obtained from the Pembrokeshire County Council and a number of elected members. During the course of the investigation evidence which was suggestive of a further breach of the Code by the Councillor was identified. The Ombudsman decided to extend the scope of the original investigation to determine whether there was evidence that the Councillor had also used his position improperly to influence another elected member.

The Councillor was interviewed in relation to both matters and denied that he had in any way attempted to influence either member.

The Ombudsman found that the direct evidence obtained during the course of the investigation was not sufficient to support the allegation made and concluded that that there was no evidence of breach in respect of all matters investigated.

Monmouthshire County Council – Disclosure and registration of interests Case reference 201405619 - Report issued in July 2015

The Ombudsman received a complaint that a member of both Monmouthshire County Council and Abergavenny Town Council ("the Town Council") had breached the Code of Conduct for members at a Planning meeting of the Town Council on 8 October 2014. It was alleged that the member had failed to declare an interest in the matter as a former high street shop trader when considering an item relating to an application for consent to operate as a street trader. It was further alleged that the member sought to use his position improperly by influencing the views of the members of the Planning meeting on the merits of the street trader consent. The complainant said that the member's actions were discriminatory.

The investigation found no evidence that the accused member had an interest in the matter under consideration. In the absence of any evidence to suggest that the accused member, or someone closely associated to him, had any specific dealings with the applicant and/or link to the application,



the Ombudsman was not satisfied that an objective person would reasonably regard the factor of the member's former occupation as so significant that the member would be considered as having an interest in the matter. Furthermore, there was no evidence that the member had spoken on the merits of the application at the Planning meeting. Having considered the available information, the Ombudsman concluded there was no evidence that the member had breached the Code.

Monmouthshire County Council

Case Number: 201405619/201405638 - Report issued in July 2015

The Ombudsman received a complaint that a member of both Monmouthshire County Council and Abergavenny Town Council ("the Town Council") had breached the Code of Conduct for members at a Planning meeting of the Town Council on 8 October 2014. It was alleged that the member had failed to declare an interest in the matter as a former high street shop trader when considering an item relating to an application for consent to operate as a street trader. It was further alleged that the member sought to use his position improperly by influencing the views of the members of the Planning meeting on the merits of the street trader consent. The complainant said that the member's actions were discriminatory.

The investigation found no evidence that the accused member had an interest in the matter under consideration. In the absence of any evidence to suggest that the accused member, or someone closely associated to him, had any specific dealings with the applicant and/or link to the application, the Ombudsman was not satisfied that an objective person would reasonably regard the factor of the member's former occupation as so significant that the member would be considered as having an interest in the matter. Furthermore, there was no evidence that the member had spoken on the merits of the application at the Planning meeting. Having considered the available information, the Ombudsman concluded there was no evidence that the member had breached the Code.

Carmarthenshire County Council Case reference 201500393/201500441 — Report issued in July 2015

A member of the public complained about a Councillor of Llanelli Rural Council ("the Council"). They said that the Councillor spoke to her in a rude and aggressive manner during a discussion outside the Trallwm Hall on 9 April. The Ombudsman investigated whether the Councillor's alleged behaviour was such that it brought the office or Council into disrepute and whether it could be considered disrespectful or bullying.

As part of the investigation, a Council officer who overheard some of the discussed was interviewed. Whilst the officer said that the nature of the discussion was not cordial, he did not hear the Councillor being disrespectful and their behaviour did not leave cause for concern. There were differing accounts of the discussion given by the Councillor and the member of the public. Independent advice was provided by the Council officer who saw nothing inappropriate in the Councillor's behaviour. The Ombudsman found there had been no breach of the code.



Rhondda Cynon Taf County Borough Council – Disclosure and registration of interests Case reference 201408430 – Report issued in September 2015

A complaint was received about a member of Rhondda Cynon Taf CBC ("the Councillor") failing to declare a personal and prejudicial interest in planning applications made by the director of a company. On 14 April 2011, at a Development Control Committee (DCC) meeting, the Councillor did not declare a personal and prejudicial interest in a planning application by the company. On 18 October 2012, an application was considered by the same applicant. The Councillor said the applicant was a family friend, she had a prejudicial interest and left the meeting. On 21 March 2013, another planning application was made by the same applicant. The Councillor said that she knew the applicant's mother, but did not leave the meeting.

The Councillor said that at the meeting on 14 April 2011, she had not known the applicant and she opposed the application. She said at the meeting on 18 October 2012, she made a personal and prejudicial declaration, as she had publicly opposed the application and left the meeting. The Councillor said that on 21 March 2013, her interest had been personal and not prejudicial. She had no voted as the application was deferred. The Councillor said she had over declared her personal interest as she had known the applicant's mother who had been her children's school secretary who are now adults. She had not sought to progress any application by the director.

The Ombudsman determined that there was no evidence that the Councillor failed to comply with the Code of Conduct in accordance with S69 (4) (a).



No action necessary

There are no summaries in relation to this finding



Referred to standards committee

There are no summaries in relation to this finding



Referred to Adjudication Panel for Wales

Llanfihangel ar Arth Community Council - Disclosure and registration of interests Case reference 201305114 Report issued in August 2015

The Ombudsman received a complaint that a member ("the Councillor") of Llanfihangel ar Arth Community Council ("the Council") had breached the Code of Conduct for Local Authority Members during meetings of the Statkraft Alltwalis Wind Farm Community Benefit Fund (upon which she had been nominated to sit in her capacity as a community councillor) and of the Council. It was alleged that the Councillor should have declared an interest and left the room when items relating to her husband's building firm and an application for a wind farm to be built next to land she owned were discussed during the meetings.

The Ombudsman determined that it was appropriate to investigate whether the Councillor had breached the Code of Conduct. Evidence was obtained from the Council, the Statkraft Alltwalis Wind Farm Community Benefit Fund and persons present at the meetings.

The Ombudsman was satisfied that on balance the evidence suggested that whilst the Councillor had a personal and prejudicial interest in Council business relating to her husband's building firm, she had appropriately declared an interest and left the room at the relevant times. The Ombudsman found that the Councillor had not been required to declare a personal interest at a number of Council meetings. However, the Ombudsman did consider that the Councillor had failed to declare a personal and prejudicial interest and to withdraw from the room whilst matters relating to the wind farm were discussed and voted on at the Council meeting of 16 January 2012.

The Ombudsman referred the matter to the Adjudication Panel for Wales for adjudication by a tribunal.

The Tribunal concluded that the Councillor had breached the Code of Conduct. Accordingly, the Tribunal decided that the Councillor should be suspended the Council for a period of three months.

The decision of the Adjudication Panel for Wales can be found here.



More Information

We value any comments or feedback you may have regarding The Code of Conduct Casebook. We would also be happy to answer any queries you may have regarding its contents. Any such correspondence can be emailed to Matthew.Aplin@ombudsman-wales.org.uk or Lucy.Geen@ombudsman-wales.org.uk, or sent to the following address:

Public Services Ombudsman for Wales 1 Ffordd yr Hen Gae Pencoed CF35 5LJ

Tel: 01656 644200 Fax: 01656 641199

e-mail: ask@ombudsman-wales.org.uk (general enquiries)

Follow us on Twitter: @OmbudsmanWales

Further information about the service offered by the Public Services Ombudsman for Wales can also be found at www.ombudsman-wales.org.uk



The Code of Conduct Casebook

Issue 7 January 2016

Inside this issue

1

Case Summaries

No	action	necessary	, 4
----	--------	-----------	-----

Referred to standards	
committee	

Referred to	
Adjudication Panel For Wales	6

Mara	Information	
IVICITE	iniormation	

Introduction

The Public Services Ombudsman for Wales considers complaints that members of local authorities in Wales have broken the Code of Conduct. The Ombudsman investigates such complaints under the provisions of Part III of the Local Government Act 2000 and the relevant Orders made by the National Assembly for Wales under that Act.

Where the Ombudsman decides that a complaint should be investigated, there are four findings, set out under section 69 of the Local Government Act 2000, which the Ombudsman can arrive at:

- (a) that there is no evidence that there has been a breach of the authority's code of conduct;
- (b) that no action needs to be taken in respect of the matters that were subject to the investigation;
- (c) that the matter be referred to the authority's monitoring officer for consideration by the standards committee;
- (d) that the matter be referred to the President of the Adjudication Panel for Wales for adjudication by a tribunal (this generally happens in more serious cases).

(Continued overleaf)





In the circumstances of (c) and (d) above, the Ombudsman is required to submit the investigation report to the standards committee or a tribunal of the Adjudication Panel for Wales and it is for them to consider the evidence found by the Ombudsman, together with any defence put forward by the member concerned. It is also for them to determine whether a breach has occurred and, if so, what penalty (if any) should be imposed.

The Code of Conduct Casebook contains summaries of reports issued by this office for which the findings were one of the four set out above. However, in reference to (c) and (d) findings, The Code of Conduct Casebook only contains the summaries of those cases for which the hearings by the standards committee or Adjudication Panel for Wales have been concluded and the outcome of the hearing is known. This edition covers October to December 2015, but also includes the summaries of older cases for which the standards committee or Adjudication Panel hearings were concluded during this period.



Case Summaries

No evidence of breach

Cardiff Council - Promotion of equality and respect Case reference 201408688 — Report issued in October 2015

A complaint was received that a member of Cardiff Council ("the Councillor") had breached the Code of Conduct for members by using her position improperly to obtain "confidential" information relating to another member of Cardiff Council and by sharing this information with her political party in order to create a disadvantage for him and for political purposes.

Evidence was obtained from Cardiff Council and a number of elected members.

The Councillor was interviewed and said that she received a request for information from an outside organisation and that her role as councillor required her to seek appropriate advice and act in accordance with that advice.

The Ombudsman concluded that it was not unreasonable for the Councillor to rely on the advice provided to her and on this basis was not persuaded that evidence gathered was suggestive that she had breached the Code of Conduct.



No action necessary

Pembrokeshire County Council – Integrity / Disclosure and registration on interests Case reference 201500279/201500292 – Report issued in December 2015

A complaint was made that a member of Pembrokeshire County Council had breached the Code of Conduct by seeking to influence a decision of the Council over a matter in which he had a prejudicial interest, that he had failed to have regard to relevant advice provided by the Council's Monitoring Officer and that he had conducted himself in a manner which could reasonably be regarded as bringing his authority into disrepute.

Evidence was obtained from the Council and a formal interview was carried out with the Councillor.

The Councillor declared a prejudicial interest, but it was apparent at interview that he did not believe that his interest in the matter was prejudicial. The Councillor acted against the direction of the Standards Committee and rejected the advice of the Monitoring Officer, both of which were based upon his declaration that he had a prejudicial interest. These actions were not appropriate and may have amounted to a breach of the Code of Conduct. However, had the Councillor fully considered his position at the outset, he may not have declared a prejudicial interest. Since these events, the Councillor has sought and complied with advice from the Monitoring Officer and the Standards Committee.

The Ombudsman determined that it would not be proportionate or in the public interest to take further action against the Councillor.

Llanelli Town Council - Objectivity and propriety Case Number: 201500965 - Report issued in December 2015

A complaint was made that, due to his conduct at an event, a member of Llanelli Town Council may have breached the Code of Conduct by failing to show respect and consideration for others, using bullying or harassing behaviour towards another, using the resources of his authority imprudently and bringing his authority into disrepute. It was alleged that the Councillor had been drinking excessively at tax payers' expense, had been verbally abusive towards members of staff and had acted in a violent manner towards another guest at the event.

A number of witness statements were obtained and the Councillor was formally interviewed. The Ombudsman determined that the evidence was not sufficiently robust that it would satisfy a Standards Committee or the Adjudication Panel for Wales that a breach of the Code of Conduct had occurred and that a sanction would be appropriate, therefore no further action was necessary.



Referred to standards committee

There are no summaries in relation to this finding



Referred to Adjudication Panel for Wales

There are no summaries in relation to this finding



More Information

We value any comments or feedback you may have regarding The Code of Conduct Casebook. We would also be happy to answer any queries you may have regarding its contents. Any such correspondence can be emailed to Matthew.Aplin@ombudsman-wales.org.uk or Lucy.Geen@ombudsman-wales.org.uk, or sent to the following address:

Public Services Ombudsman for Wales 1 Ffordd yr Hen Gae Pencoed CF35 5LJ

Tel: 01656 644200 Fax: 01656 641199

e-mail: ask@ombudsman-wales.org.uk (general enquiries)

Follow us on Twitter: @OmbudsmanWales

Further information about the service offered by the Public Services Ombudsman for Wales can also be found at www.ombudsman-wales.org.uk



Eitem Agenda 10

Adroddiad i'r: Pwyllgor Safonau

Dyddiad y Cyfarfod: 4 Mawrth 2016

Aelod / Swyddog Arweiniol: Gary Williams, Swyddog Monitro

Awdur yr Adroddiad: Gary Williams, Swyddog Monitro

Teitl Mesur Drafft Llywodraeth Leol (Cymru)

1. Beth yw cynnwys yr adroddiad?

1.1 Mae'r adroddiad hwn yn ymwneud â darpariaethau'r Mesur Drafft Llywodraeth Leol (Cymru) ("y Mesur Drafft").

2. Beth yw'r rheswm dros wneud yr adroddiad hwn?

2.1 Y rheswm dros wneud yr adroddiad yw hysbysu'r Pwyllgor o ddarpariaethau'r Mesur Drafft sy'n ymwneud â'r Pwyllgor.

3. Beth yw'r Argymhellion?

3.1 Bod y Pwyllgor yn nodi cynnwys y Mesur Drafft ac ymateb ymgynghorol y Cyngor a nodir yn Atodiad 3.

4. Manylion yr Adroddiad

- 4.1 Bydd yr Aelodau'n cofio yn 2015 i'r Cyngor ddarparu ymateb i'r ymgynghoriad ar y Papur Gwyn dan y teitl "Diwygio Llywodraeth Leol Pŵer i Bobl Leol. (" y Papur Gwyn ") Mae Llywodraeth Cymru ers hynny wedi deddfu Deddf Llywodraeth Leol (Cymru) 2015 sy'n gwneud darpariaethau ar gyfer uno gwirfoddol o awdurdodau lleol, pwyllgorau pontio, yr adolygiad o drefniadau etholiadol ar gyfer awdurdodau newydd ac ataliadau ar drafodion a recriwtio gan awdurdodau uno.
- 4.2 Mae'r Mesur Drafft yn gwneud cynigion ar gyfer uno awdurdodau lleol yn y dyfodol yng Nghymru ac yn gofyn am sylwadau ar y cynigion hynny. Mae'r Mesur Drafft, fodd bynnag, yn cynnwys llawer mwy na chynigion ar gyfer uno. Mae'r cynigion uno wedi'u nodi yn Rhan 1 o'r Mesur Drafft. Mae 8 rhan i'r Mesur Drafft ac mae Rhannau 2-8, os yn cael eu deddfu fel darn o ddeddfwriaeth annibynnol, yn wynebu'r diwygiadau llywodraeth leol mwyaf sylweddol yng Nghymru ers Deddf Llywodraeth Leol 2000.
- 4.3 Mae Aelodau wedi cynnwys yr atodiadau canlynol i'r adroddiad hwn:

Atodiad 1 Crynodeb o brif ddarpariaethau'r Mesur Drafft

Atodiad 2 Dogfen Ymgynghori Atodiad 3 Ymateb Ymgynghorol y Cyngor Atodiad 4 Mesur Drafft Llywodraeth Leol Cymru

4.4 Mae'r Mesur Drafft wedi ei strwythuro mewn 8 rhan fel a ganlyn:

Rhan 1 Ardaloedd Llywodraeth Leol a Chynghorau Sir

Rhan 2 Pŵer Cymhwysedd Cyffredinol

Rhan 3 Hyrwyddo Mynediad i Lywodraeth Leol

Rhan 4 Swyddogaethau Cynghorau Sir a'u Haelodau

Rhan 5 Cynghorau Sir: Gwella Llywodraethu

Rhan 6 Cynghorau Sir

Rhan 7 Materion Gweithlu

Rhan 8 Cyffredinol ac Atodiadau

- 4.5 Mae'n bwysig nodi wrth gynhyrchu'r Mesur Drafft, bod Llywodraeth Cymru wedi cymryd i ystyriaeth nifer o'r pryderon a fynegwyd gan gynghorau yn ystod yr ymgynghoriad Papur Gwyn. Tydi rhai o'r cynigion dadleuol megis terfynau tymor ar gyfer cynghorwyr, etholiadau fesul cam ac adolygu tâl aelodau, ddim yn ymddangos yn y Mesur Drafft.
- 4.6 Y rhan mwyaf perthnasol o'r Mesur Drafft i'r Pwyllgor yw Rhan 4 sy'n ceisio cyflwyno dyletswyddau statudol newydd ar aelodau etholedig.
- 4.7 Rhaid i Aelod fynychu'r holl gyfarfodydd perthnasol oni bai bod ganddynt reswm da dros beidio. Mae hyn yn cynnwys yr holl gyfarfodydd ffurfiol yn ogystal ag unrhyw gyfarfod arall y byddai disgwyl i Aelod o fewn rheswm i fod yn bresennol ynddo wrth arfer ei swyddogaethau fel Aelod. Fodd bynnag, mae'r Mesur Drafft yn datgan yn benodol nad oes unrhyw beth yn y ddyletswydd hon yn effeithio ar neu'n newid y gofyniad cyfreithiol i Aelod gael ei wneud yn anghymwys os yw ef / hi yn methu mynychu am 6 mis.
- 4.8 Rhaid i Aelod gynnal o leiaf pedwar cymorthfeydd ym mhob cyfnod o 12 mis ar ôl cymryd y swydd, oni bai bod ganddynt reswm da dros beidio â gwneud hynny. Mae cymhorthfa yn cael ei gynnal os bydd Aelod yn gwneud ei hun ar gael ar gyfer o leiaf un awr i gwrdd ag aelodau o'r cyhoedd i drafod materion yn breifat. Rhaid i ddyddiad, amser a lleoliad y gymhorthfa gael ei gyhoeddi ar wefan y Cyngor o leiaf saith diwrnod cyn y gymhorthfa.
- 4.9 Rhaid i Aelod ymateb i'r holl ohebiaeth a anfonir at ei gyfeiriad swyddogol o fewn 14 diwrnod o dderbyn oni bai fod ganddo reswm da dros beidio.
- 4.10 Rhaid i Aelod gwblhau'r holl gyrsiau hyfforddi gorfodol oni bai bod ganddynt reswm da dros beidio. Mae hyfforddiant yn orfodol os yw'r Cyngor wedi rhoi gwybod i Aelod y mae'n rhaid iddynt ei gwblhau.
- 4.11 Rhaid i Aelod lunio adroddiad blynyddol am ei weithgareddau fel Aelod a chanlyniadau'r gweithgareddau hynny bob blwyddyn. Mae'n rhaid i'r rhain gael eu cyflwyno i'r Pennaeth Gwasanaethau Democrataidd ac mae'n rhaid eu

- cyhoeddi. Nid oes unrhyw ddarpariaeth ar gyfer cael rheswm da dros beidio â gwneud hyn.
- 4.12 Mae Rhan 4 hefyd yn rhoi dyletswydd bersonol ar arweinwyr grwpiau gwleidyddol i gydweithio â'r Pwyllgor Safonau a chymryd camau rhesymol i hyrwyddo a chynnal safonau uchel o ymddygiad gan Aelodau o'u Grŵp. Rhaid i Bwyllgorau Safonau drefnu hyfforddiant i Arweinwyr Grŵp a monitro eu cydymffurfiad â'r ddyletswydd hon.
- 4.13 Mae'r Mesur Drafft hefyd yn awgrymu mecanwaith gorfodi am dorri'r dyletswyddau hyn. Gall unrhyw berson wneud cwyn i'r Swyddog Monitro yn ysgrifenedig am Aelod sydd o bosib yn torri unrhyw un o'r dyletswyddau a nodir uchod, ac eithrio'r ddyletswydd i wneud Adroddiad Blynyddol.
- 4.14 Mae'n rhaid i'r Swyddog Monitro gyfeirio unrhyw gŵyn a dderbynnir i Gadeirydd y Pwyllgor Safonau, a gyda'i gilydd rhaid iddynt benderfynu a ddylai'r mater gael ei ymchwilio. Mae'n rhaid i'r Swyddog Monitro ymchwilio i'r gwyn oni bai bod Cadeirydd y Pwyllgor Safonau a'r Swyddog Monitro yn ystyried na ddylid ei ymchwilio.
- 4.15 Os bydd Pennaeth Gwasanaethau Democrataidd yn credu bod Aelod wedi torri'r ddyletswydd i lunio adroddiad blynyddol gall ef / hi ei gyfeirio at y Swyddog Monitro ar gyfer ymchwiliad. Wrth ystyried a ddylid ymchwilio rhaid i'r Swyddog Monitro ymgynghori â Chadeirydd y Pwyllgor Safonau.
- 4.16 Os caiff ymchwiliad ei gynnal, mae'n rhaid i adroddiad gael ei ddarparu i'r Pwyllgor Safonau gydag unrhyw argymhellion y mae'r Swyddog Monitro yn ystyried yn briodol.
- 4.17 Os bydd y Pwyllgor Safonau yn penderfynu bod Aelod wedi torri un o'r dyletswyddau yna gall achosi cerydd, gwaharddiad neu waharddiad rhannol am hyd at chwe mis neu gymryd unrhyw gamau pellach.
- 4.18 Gellir gweld ymateb y Cyngor i Ran 4 o'r Mesur Drafft yn Atodiad 3.
- 5. Sut mae'r penderfyniad yn cyfrannu at y Blaenoriaethau Corfforaethol?
- 5.1 Nid yw'r penderfyniad yn effeithio'n uniongyrchol ar flaenoriaethau corfforaethol.
- 6. Beth fydd yn ei gostio a sut bydd yn effeithio ar wasanaethau eraill?
- 6.1 Ni fydd ymateb i'r ymgynghoriad yn golygu unrhyw gostau ychwanegol.
- 7. Beth yw'r prif gasgliadau o'r Asesiad o'r Effaith ar Gydraddoldeb (EqIA) a gynhaliwyd ar y penderfyniad? Dylai'r templed EqIA wedi'i lenwi gael ei atodi i'r adroddiad.
- 7.1 Nid oes gofyniad am asesiad.
- 8. Pa ymgynghoriadau sydd wedi'u cynnal gydag Archwilio ac eraill?

8.1 Ni fu unrhyw ymgynghori ag Archwilio mewn perthynas â'r mater hwn, fodd bynnag, cyflwynwyd ymateb ymgynghorol y Cyngor yn dilyn gweithdy cyngor gyda'r holl aelodau a chymeradwyaeth o'r fersiwn terfynol gan yr Arweinydd mewn ymgynghoriad ag Arweinwyr Grŵp.

9. Datganiad y Prif Swyddog Cyllid

- 9.1 Costau ymateb i'r ymgynghoriad yn fach iawn a byddant yn cael eu cynnwys o fewn yr adnoddau presennol. Mae goblygiadau cost ehangach y Mesur o bosib yn sylweddol a bydd angen eu hasesu yn fwy manwl wrth i gynigion ddatblygu
- 10. Pa risgiau sydd ac a oes unrhyw beth y gallwn ei wneud i'w lleihau?
- 10.1 Nid oes unrhyw risgiau yn gysylltiedig yn uniongyrchol â'r adroddiad hwn.
- 11. Pŵer i wneud y Penderfyniad
- 11.1 Nid oes angen penderfyniad.

Appendix 1

<u>Draft Local Government Wales Bill – Summary of Provisions</u>

Part 1 – Local Government Areas and County Councils

Chapter 1 – Local Government Areas

- 1.1.1 The only provision of this Chapter of the Draft Bill is concerned with the creation of new local government areas from 1st April 2020. There are 2 tables set out in Schedule 1 to the Draft Bill, Table 1 which shows 8 counties including 2 in North Wales, and Table 2, which shows 9 counties including 3 in North Wales. Consultees are asked to provide their views on the options for North Wales and the overall configuration of local government areas in Wales as proposed in the Draft Bill.
- 1.1.2 Members will be most interested in the issues raised in respect of proposals for North Wales. These are set out on Pages 9 and 10 of the Consultation Document. These are described as being finely balanced and include consideration of population sizes, language, regional variances in financial indicators, and the sub-regional structures of other bodies such as health, police and fire.

Chapter 2 – County Councils

1.2.1 This Chapter contains provisions relating to the election of a Chair, the power for a County Council to give the Chair the title of Mayor, the power to appoint a presiding officer and sets a new fixed cycle of 5 year terms for County Councillors beginning in 2023. It introduces Schedule 2 to the Draft Bill which contains provisions relating to the first elections to and the first meetings of the new councils.

Chapter 3 – Establishing the Councils for the New Counties

- 1.3.1 This Chapter provides for the process by which the new councils come into being. The proposed election timetable is that elections to the existing councils will take place in May 2017 for a period of three years. In May 2019, there will be elections to the new councils which will be shadow authorities until 1st April 2020 when they will take on the full functions of local authorities for the new areas. There will then be elections to these authorities in May 2023 from which point Councillors will be elected for fixed 5 year terms.
- 1.3.2 This Chapter also introduces Schedule 3 to the Act which makes provision in respect of the new councils relating to finance issues such as council tax and non-domestic rating and council funds giving the shadow authorities the ability to make preparations for the new authorities. Further detail will be provided in Regulations. It also introduces Schedule 4 which makes provision for the transfer of staff, property, rights and liabilities.

Chapter 4 – Miscellaneous Provision

1.4.1 This Chapter makes various technical provisions to include the power for Welsh Ministers to make regulations to make consequential amendments to previous legislation and transitional provisions. It also makes provision for the amendment of previous legislational provisions.

Other Matters

- 1.5 The Consultation Document, in respect of Part 1 confirms that the previous White Paper proposal for phased elections will not be taken forward. The previous cap of 75 on the number of Councillors for a County Council has been removed. Consideration is currently being given to the appropriate number for the new councils.
- 1.6 The Government proposes to abolish the "preserved counties". These were the counties created under the Local Government Act 1972 which after local government reorganisation in 1996 were preserved for various administrative and judicial purposes. Since then most of the purposes for these areas have fallen away. The proposal is that these "preserved counties" be abolished and their place taken for those purposes that remain, e.g. the appointment of Lord-Lieutenants and High Sheriffs by the new counties. Views are sought on these issues.

Part 2 – General Power of Competence

Chapter 1 – The General Power

- 2.1.1 Chapter 1 provides for the introduction of a general power of competence for County Councils and Community Councils with competence. This power enables a Council to do anything that an individual may do provided that it is not otherwise prohibited by law from doing it. The Draft Bill places restrictions on the extent of the power. A Council may not charge for anything that it has a statutory duty to do. Any charge is subject to a duty that from one financial year to the other, the income from charges is not greater than the cost of providing the service.
- 2.1.2 The general power includes the power to do things for a commercial purpose provided that the Council is not under a statutory duty to provide the service and that the service is provided through a company. Welsh Ministers may make regulations regarding conditions to be placed on the exercise of the power and to remove any obstructions to the exercise of the power.
- 2.1.3 It is intended to bring this into force at the earliest opportunity to allow existing authorities to use it.

Chapter 2 – Community Councils with Competence

2.2.1 This Chapter sets out how Community Councils may become Community Councils with competence. A Community Council will be able to pass a resolution that it meets the competency requirements and that it is a Community Council with competence. It will then remain "competent" until the first meeting after the next ordinary election whether or not it continues to satisfy those requirements, unless it declares itself to no longer be competent. Even if it ceases to be competent it may continue to exercise the power of competence in respect of any activity it undertook whilst it was a Council with competence.

- 2.2.2 There are three requirements. Firstly that at least two thirds of the Members have been elected, secondly that the Clerk has a relevant professional qualification and thirdly that the Council has, for two consecutive years, received an unqualified Auditor's opinion.
- 2.2.3 This is different to the White Paper proposal which suggested that Community Councils would in addition only be competent if they had a turnover of greater than £200k and placed an obligation on the County Council to determine whether a Community Council continued to be competent.

Part 3 – Promoting Access to Local Government

Chapter 1

3.1.1 This chapter provides an overview of this Part.

Chapter 2 – Public Participation in Local Government

- 3.2.1 This chapter requires councils to encourage local people to participate in decision making by the Council and other authorities connected with the Council in the area. These are set out in the Draft Bill as Community Councils, the Fire and Rescue Authority for the area and a National Park Authority for a national park which is in any part of the Council's area.
- 3.2.2 A Council will also be required to prepare and publish a public participation strategy that sets out how it will comply with these requirements. The Draft Bill contains a list of issues that must be addressed by the strategy. When preparing a strategy the Council must consult local people. The strategy must be reviewed after each ordinary election, and local people must be consulted as part of the review.
- 3.2.3 This Chapter also requires a Council to consult prior to formally setting a budget. The Draft Bill sets out a list of those who must be consulted, including, local people, businesses in the area, persons who provide services on the Council's behalf, persons who carry out activities in the area for which the Council provides financial assistance, voluntary bodies and trade unions.
- 3.2.4 In the White Paper it was suggested that Councils should have to set up Youth Councils. This has not been included in the Draft Bill although guidance to be issued on public participation strategies will emphasise the role of children and young people.

Chapter 3 – Community Area Committees

3.3.1 Further provisions in Part 3 relate to the creation of Community Area Committees which must be fully established statutory committees. These committees will comprise County Councillors for that area, representation from Community Councils, other public bodies and voluntary bodies. Each of these members and representatives will be able to vote. They will publish an annual statement of local priorities and objectives which must have regard to the local well-being assessment conducted under the Well Being of Future Generations (Wales) Act. The committees must consult local people and voluntary bodies on a draft of The statement before it is submitted to the

County Council. The committees may have functions delegated to them and may themselves delegate to a sub-committee or officer. Welsh Ministers may make regulations to require or restrict the delegation of certain functions to these committees.

3.3.2 Part 3 also provides that "interested bodies" may make improvement requests to a County Council with the aim of entering a dialogue about how services could be improved. In respect of each request the Council would have to accept or refuse within 45 days. A County Council would have to agree to the request unless there are reasonable grounds for refusing it. If it agrees to enter discussions about the improvement request the County Council must enter into those discussions within 45 days and publish online, a summary of the discussions and agreed actions. The County Council must publish an annual report of all improvement requests and maintain a complaints system in respect of them. Welsh Ministers can make regulations to make further provision regarding the procedure to be used by Councils in dealing with requests, the reporting of them and the support that Councils must make available to interested bodies to enable them to make requests and enter into discussions.

Chapter 5 – Access to Meetings of Local Authorities

- 3.5.1 At least once a year the Leader must attend a meeting at which local people are given an opportunity to put questions to the Leader about the discharge of Cabinet functions.
- 3.5.2 All public meetings of Council or any committee or sub-committee must be broadcast live and must be available electronically for a reasonable period after the meeting. Welsh Ministers may make regulations for allowing persons to film, photograph or make sound recordings of meetings and to provide written or oral commentary as the meeting takes place or later. These regulations can set out the facilities to be made available by the Council to persons wishing to do this, and the steps to be taken by those persons before doing it.
- 3.5.3 The chapter also provides for Community Councils to make standing orders enabling members of the public in attendance at a meeting to make representations about the business to be conducted.

Chapter 6 – Further Duties of County Councils

- 3.6.1 The Council will have to produce a guide to the constitution explaining the content of the Constitution in ordinary language.
- 3.6.2 The Council must publish an official electronic and postal address for every member of the Council on its website to which correspondence may be sent.

Part 4 – Functions of County Council and their Members

Chapter 1

4.1.1 This provides an overview of this Part.

Chapter 2 – Duties of Members of County Councils

This Chapter places duties on individual members.

- 4.2.1 A member must attend all relevant meetings unless they have a good reason not to. This includes all formal meetings as well as any other meeting that a Member would be reasonably expected to attend in the exercise of his or her functions as a Member. However, the Draft Bill expressly states that nothing in this duty affects or changes the legal requirement that a Member be disqualified if he/she fails to attend for 6 months.
- 4.2.2 A Member must hold at least four surgeries in every 12 month period following their taking up office, unless they have a good reason not to. A surgery is held if a Member makes him or herself available for at least one hour to meet members of the public to discuss matters in private. The date, time and location of the surgery must be published on the Council's website at least seven days in advance of the surgery.
- 4.2.3 A Member must respond to all correspondence sent to his or her official address within 14 days of receipt unless he or she has a good reason not to.
- 4.2.4 A Member must complete all compulsory training courses unless they have a good reason not to. Training is compulsory if the Council has notified a Member that they must complete it.
- 4.2.5 A Member must make an annual report about his or her activities as a Member and the results of those activities every year. These must be submitted to the Head of Democratic Services and they must be published. There is no provision for having a good reason not to do this.

Chapter 3 – Breaches of Duties under Chapter 2

- 4.3.1 Any person may make a complaint, in writing, to the Monitoring Officer about a possible breach by a Member of any of the duties set out in Chapter 2 other than the duty to make an Annual Report.
- 4.3.2 The Monitoring Officer must refer any complaint received to the Chair of the Standards Committee and together they must decide whether or not the matter should be investigated. The Monitoring Officer must investigate the complaint unless both the Chair of Standards Committee and the Monitoring Officer consider that it should not be investigated.
- 4.3.3 If the Head of Democratic Services believes that a Member has breached the duty to make an annual report he/she can refer it to the Monitoring Officer for investigation. When considering whether to investigate the Monitoring Officer must consult the Chair of Standards Committee.
- 4.3.4 If an investigation is conducted a report must be provided to the Standards Committee with any recommendations that the Monitoring Officer considers appropriate.
- 4.3.5 A procedure is set out in this Chapter for the hearing of these matters by the Standards Committee. If the Standards Committee decides that a Member

- has breached one of the duties then it can impose a censure, a suspension or partial suspension for up to six months or take no further action.
- 4.3.6 Welsh Ministers may make regulations about how investigations and hearings into breaches of these duties are dealt with, as well as for an appeals mechanism.
- 4.3.7 Standards Committees must also arrange training for Members on these duties.
- 4.3.8 This Chapter also places a personal duty on leaders of political groups to cooperate with the Standards Committee and take reasonable steps to promote and maintain high standards of conduct by Members of their group. Standards Committee must arrange training for Group Leaders and monitor their compliance with this duty.
- 4.3.9 Welsh Ministers may make regulations regarding the circumstances in which Members are treated as being a political group and a Member is treated as a leader of a group.

Chapter 5 – Executives of County Councils

- 4.5.1 This Chapter requires the Leader as soon as practicable after they are elected to set objectives to be met by the executive (Cabinet). The Leader must keep these under review and report annually on performance against these objectives. The objectives and the annual report must be published.
- 4.5.2 All candidates for the post of Leader must prepare a written manifesto and this must be made available to all Members before they elect a Leader.
- 4.5.3 The Council's executive arrangements may include provision for the appointment by the Leader of Councillors to assist the executive who are referred to in the Draft Bill as Assistants to the Executive. They are not members of the executive but would be entitled to attend and speak at Cabinet meetings but not vote and would not be eligible for a senior salary. This replaces the proposal in the White Paper that there be deputy cabinet members.

<u>Chapter 6 – Appointment etc. of Certain Chief Officers</u>

- 4.6.1 This Chapter requires the appointment of a Chief Executive and provides that the role includes the duties currently required by legislation to be those of the Head of Paid Service.
- 4.6.2 The Leader must set objectives to be met by the Chief Executive, review annually the extent to which the Chief Executive has met the objectives and if appropriate revise them. The objectives must be published.
- 4.6.3 The Leader must prepare and publish a report on the review of achievement against the objectives. The Council must consider the report within 3 months of its publication.

4.6.4	of a Chief Officer and the prohibition on the Monitoring Officer from fulfilling that role has been removed.

Chapter 7 - Overview and Scrutiny Committees and Standards Committees

- 4.7.1 The law does not generally permit co-opted members of scrutiny committees to have voting rights with some exceptions e.g. education co-optees. This chapter gives Councils power to grant voting rights to co-optees on an individual basis which can be limited by time and by the questions upon which a vote may be cast. Welsh Ministers may make regulations to govern this provision which may also set maximum or minimum numbers of co-opted members on a Scrutiny Committee that may be able to vote.
- 4.7.2 This Chapter also gives Welsh Ministers power to make regulations to require Councils to establish Joint Scrutiny Committees, and regulations to require information to be made available to Scrutiny Committees about decisions that Cabinet has made or intends to make akin to the "key decisions" framework in England.
- 4.7.3 The Chapter also requires Standards Committees to make annual reports to Council on their activities to include in particular the committee's assessment of the extent to which Group Leaders have complied with their duties regarding conduct and a summary of the breaches of Members' duties as set out above that have been referred to it.
- 4.7.4 The Consultation Document in respect of this Part refers to other matters which are not included in the Draft Bill but were raised in the White Paper

Delegation of functions

4.8 It seeks views on the Government's intention to repeal Part 2 Deregulation and Contracting Out Act 1994 to be replaced with regulations allowing County Councils and others to delegate functions to third parties. It is intended to do this as soon as practicable after the passing of the Bill to make this available to existing Councils.

Electoral qualification

- 4.9 The government will not be proceeding with the proposal that they lift the bar on employees of a council seeking election to that Council.
- 4.10 The Draft Bill does not include proposals that candidates be required to declare membership of a political party, nor that members of County Councils be prevented from being members of a Community Council, nor that persons be prevented from being a member of more than one Community Council. They will give these issues further consideration.

Term Limits

4.11 The White Paper contained proposals for term limits for Councillors, Leaders and Cabinet Members. These will not be taken forward.

Release of Councillors from Employment

4.12 The White Paper proposal to require public bodies to release employees to undertake duties as Coun in the Draft Bill.

Remuneration of Councillors

4.13 The White Paper considered members' remuneration. No proposals are included in the Draft Bill, however, the Consultation Document seeks views on the proposal that the final Bill include power for Welsh Ministers to direct the Independent Remuneration Panel for Wales to have regard to guidance when reviewing members' remuneration.

Recall of Councillors

4.14 The Draft Bill does not include recall provisions but the Consultation Document states that provisions for recall be included presumably in the final Bill are being considered whereby a by-election would be triggered if a Member was suspended and twenty electors signed a petition. No views are specifically sought on this point.

Remote Attendance at Meetings

4.15 Although no provision is made in the Draft Bill it is Government's intention to include provisions in the final Bill provisions to further facilitate remote attendance at meetings. Views are sought on this point.

Costs of Senior Management in Local Government

4.16 The proposals in the White Paper for central appointments for Chief Executives and Chief Officers, term limits for Chief Executives and central control mechanisms for salaries of Chief Officers are not included in the Draft Bill. The Consultation Document says that these will be considered further as part of a wider exercise in respect of the whole public service. Views are not specifically sought on this.

The Role and Responsibilities of Chief Executives

- 4.17 No provision is made in the Draft Bill for including the Returning Officer role as an intrinsic duty of a Chief Executive for which no additional fee would be payable but the Consultation Document states that this will be brought forward in other legislation at some point.
- 4.18 It is proposed that Shadow Authorities appoint Returning Officers for the 2020 General Election given that it will take place only one month after the new authorities come into being.
- 4.19 There was in the White Paper a proposal that Chief Executives should have a specific duty to promote engagement and diversity in democracy. This is not included in the Draft Bill although it is suggested that future guidance to Leaders about the setting of objectives for Chief Executives will deal with this issue.

Power to Dismiss Senior Officers by a Vote of the Full Council

4.20 Currently, the law requires that the Chief Executive, Chief Finance Officer, Monitoring Officer and Head of **Demander** (Services can only be dismissed if

an independent person has conducted an investigation which would justify this. The Government is considering a change which would allow these officers to be dismissed by a vote in Full Council without the necessity of this independent investigation. This is not included in the Draft Bill but the Consultation Document seeks views on whether this approach should be taken in Wales as it has been in England.

Council Functions and Responsibilities

4.21 Currently the legality of whether a decision is to be made or a function exercised by Cabinet or Council is one which is prescribed in detailed regulations. These are liable to become out of date quite quickly and require frequent revision and amendment. The Government is suggesting that the final Bill includes provision for local authorities to decide which functions are exercised by whom subject to having to abide by certain principles, having regard to Ministerial Guidance and a reserve power for Welsh Ministers to direct a local authority to allocate responsibility in a certain way. The Consultation Document seeks views on this approach.

Transfer of Local Authority Assets

4.22 The White Paper contained proposals that community bodies could initiate the transfer to them of local authority assets. There are no proposals relating to this in the Draft Bill but the Consultation Document states that Government intends to include provisions in the final Bill and seeks views on a modified approach which would require local authorities to give statutory notice to community bodies that they intend to dispose of assets above a certain as yet unidentified value. There may also be further provisions relating to the length of notice given, how a local authority responds when a community body indicates an interest in a particular asset and how urgent situations should be dealt with.

Assets of Community Value

4.23 The White Paper contained a proposal that local authorities be required to maintain a register of assets of community value e.g. village pub etc. There is no provision for this in the Draft Bill but further ministerial consideration will be given to what action is to be taken and whether there is a need for legislation.

Part 5 – County Councils: Improvement of Governance

Chapter 1 – Duty To Make Arrangements To Secure Good Governance

- 5.1.1 Councils are currently required to secure continuous improvement. This Chapter will replace that duty with a duty to make implement and comply with arrangements to ensure good governance, accountability, and economy, efficiency and effectiveness in the use of Council resources.
- 5.1.2 Welsh Ministers will issue regulations setting out principles, processes and practices to be applied in making governance arrangements and can prescribe a Code of Practice or guidance.

Chapter 2 – Corporate plans

- 5.2.1 The White Paper proposed that a Council should have a Corporate Plan produced by the Chief Executive. The Consultation Document accepts the arguments made in response to the White Paper that the Corporate Plan should be approved by the Council.
- 5.2.2 The Corporate Plan must include a statement of the Council's priorities for the short, medium and long term and identify issues in respect of which it expects to have to make significant decisions. It must contain certain documents such as a self-assessment report about governance, a statement of well-being objectives published under the Well-Being of Future Generations Act and the assessment of local well-being produced by the Public Services Board.
- 5.2.3 It must also include details about the Council's strategies for workforce planning, financial planning, complaints handling, risk management, scrutiny, management of assets and public participation.
- 5.2.4 Welsh Ministers can make regulations about what else must be included.
- 5.2.5 The Corporate Plan must be produced within three months of the date of the first ordinary election of councillors after the coming into force of this part of the Draft Bill and each subsequent election.
- 5.2.6 A Council can review the plan from time to time but must do so at least once a year.
- 5.2.7 A Council must consult the Local Health Board, the Public Services Board and any other person it thinks appropriate before preparing or revising the Corporate Plan.

Chapter 3 – Assessments of Governance Arrangements

- 5.3.1 A Council must conduct a self-assessment of its compliance with the duty in Chapter 1 to make, implement and comply with arrangements to achieve good governance, accountability and efficient and effective use of resources. It must do this at least once a year. A draft of the self-assessment report must be put to the Corporate Governance and Audit Committee and take into account any changes it recommends. The final report must be published.
- 5.3.2 Welsh Ministers may make regulations about the form and content of a self-assessment report and how and when it must be published.
- 5.3.3 In addition, a Council must arrange for a peer assessment by a panel appointed by the Council. This must be done before 31st March 2023 and at least once during each subsequent electoral cycle. A model approach to peer assessment is set out in Annex A to the Consultation Document.
- 5.3.4 The Council must consider and publish the peer assessment report. It must also respond to it and publish the response setting out whether it accepts the recommendations made and identifying any action it proposes to take as a result.

- 5.3.5 Welsh Ministers may make regulations about the form and content or peer assessment reports and responses.
- 5.3.6 The Council's regulators must produce a combined assessment of governance arrangements at a council at such intervals as they think fit. There are similar provisions relating to the reporting and response to reports as are set out above in relation to peer assessments. Welsh Ministers will have the power to make regulations as to when such assessments should be carried out and the form and content of reports and responses.

Chapter 5 – Governance Reviews and Intervention

- 5.5.1 If Welsh Ministers believe that a Council is failing in its governance duty they may arrange for a governance review of the Council. Before doing so they must consult the Council and other persons they think appropriate. The Council will be required to co-operate with the review. Other public bodies such as the Public Services Board, Local Health Board, Community Council, Estyn and Auditor General for Wales will be required to take reasonable steps to facilitate the review by providing documents records and other information.
- 5.5.2 There must be a review report which the Council must publish within 14 days of receipt.
- 5.5.3 The Council must prepare a response taking into account the views of the Corporate Governance and Audit Committee and identifying the action it proposes to take.
- 5.5.4 Welsh Ministers can on receipt of the response give certain directions, called intervention directions, with which the Council must comply. These can include a direction to enter into a contract with a provider for the provision of advice training or support, a direction to take specific steps and a direction that specified functions of the Council be exercised by Welsh Ministers or persons nominated by them.
- 5.5.5 There is a duty to co-operate with Welsh Ministers or persons authorised by them in connection with an intervention direction.

<u>Chapter 5 – Co-ordination Between Regulators</u>

5.5.6 Welsh Ministers may make regulations in connection with the co-ordination by regulators of relevant functions.

<u>Chapter 6 – Miscellaneous Provisions about Chapters1-5</u>

5.6.1 This Chapter requires those exercising functions under this Part to have regard to Guidance. It also allows the Council to determine whether the Council or Cabinet exercises functions under this Part.

Chapter 7 – Corporate Governance and Audit Committees

5.7.1 Each Council must have a "Corporate Governance and Audit Committee" which will have in addition to its current functions as an Audit Committee the following additional functions as an Audit Committee the

- 5.7.2 It must assess the Council's ability to comply with its governance duty, its ability to make progress in relation to the priorities set out in the Corporate Plan, its arrangements for self and peer assessment and its ability to handle complaints effectively.
- 5.7.3 In addition the Committee will have functions relating to making recommendations to Council in respect of the content of self-assessment reports, peer assessment reports and responses, combined assessment reports and responses to governance reviews.
- 5.7.4 Membership of the Committee will be amended. At present our Corporate Governance Committee which is the Council's Audit Committee for the purposes of the current legislation, has the statutory minimum of one lay member. The Draft Bill proposes that there be a minimum of at least one third of the members of the Corporate Governance and Audit Committee who are lay members and that the Chair must be a lay member. The Deputy Chair must not be a member of the Cabinet.
- 5.7.5 Views are sought in the Consultation Document on the role of the Committee in the various assessment and review processes.

Other Matters

The Consultation Document then refers under this Part to matters which are not included in the Draft Bill.

Strengthening External and Internal Scrutiny

5.7.6 The White Paper proposed stronger links between scrutiny and external regulators. There are no provisions regarding this in the Draft Bill but the Consultation Document says that there will be provision made in the final Bill. The proposal is that regulators would be required to provide copies of their reports to Scrutiny at the same time that final reports are issued to the Council. The Council would be required to involve Scrutiny in the development of its response and have regard to any recommendations made by the relevant committee. The regulators would be required to present their report to the relevant Scrutiny Committee when requested to do so.

Report on the "State of Local Government"

5.7.7 The White Paper contained a proposal for regulators to make a State of Local Government Report". It has been decided not to include this in the Draft Bill since the Well-Being of Future Generations Act requires a number of reports.

Single Information Portal

5.7.8 The White Paper proposed a single information portal for key documents and data. There are no provisions in the Draft Bill but the Government intends to include provision in the final Bill.

Complaints

5.7.9 The White Paper suggested a statutory complaints system be established. A Draft Public Services Ombudsman (Wales) Bill has been introduced and the Government proposes to reflect on how this develops before making further proposals.

Local Public Accounts Committees

5.7.10 The White Paper contained a proposal to create a Public Accounts Committee to examine the impact of public spending on outcomes in an area. This is no longer proposed. The Consultation Document does however ask for views on whether Public Services Boards created under the Well-Being of Future Generations (Wales) Act should be the appropriate forum for examining prospective policy choices facing local public services, and if so, whether they should be given additional powers to call officers to give evidence and commission external experts to inform their investigations.

Shared Services

- 5.7.11 The Consultation Document makes reference to the Williams Commission's recommendation that the NHS Wales Shared Service Partnership was a good model for public sector wide shared services. It also makes reference to the KPMG report into support functions which suggests cost savings and improved service through operating on a larger scale.
- 5.7.12 The Consultation Document asks for views on whether the current legislative framework is appropriate for enabling public sector wide shared services and asks what legislative measures could be taken to improve this.

Part 6 – Community Councils

Chapter 1 – Review of Community Council Arrangements

6.1.1 The White Paper proposed that County Councils conduct a review of Community Councils in their area with a view to having fewer, larger Councils. The Government has recognised that this would impose a significant burden on new Councils. This task is now proposed to be done by the Local Democracy and Boundary Commission who will submit reports to the new Councils on or after 1st April 2020. Views are sought on whether this is brought forward to May 2019 in order that the reviews may be undertaken in a more timely fashion. Views are also sought on whether the new Councils should implement the reviews, which is the current proposal in the Draft Bill, or whether the Commission should do so themselves.

Chapter 2 – Members of Community Councils to Complete Training

6.2.1 The Draft Bill contains provisions that a County Council must, having regard to guidance from Welsh Ministers, decide whether it should be compulsory for members of Community Councils in their area to complete training on certain matters. If they do so decide then the County Council must provide that

- training or secure its provision. Members of Community Councils must complete compulsory training unless they have a good reason not to.
- 6.2.2 Any person can complain that a Member has not completed compulsory training. The Draft Bill imposes a duty on the Clerk to the Community Council to determine whether or not the Member has breached their duty and if so must notify the Community Council of the breach at a meeting of the Community Council.

Chapter 3 – Community Council Election Dates

6.3.1 The Draft Bill contains provision that Community Council elections be held in 2017 and 2023 and thereafter at five yearly intervals on the same cycle as County Councils. Views are sought on this proposal.

Other Matters

There are in the Consultation Document in reference to this Part, other matters which are not included in the Draft Bill.

Capping the Community Council Precept

6.3.2 The White Paper contained proposals to cap the precept of non-competent Community Councils. This is not in the Draft Bill and will be considered on a longer term basis.

Annual Reports

6.3.3 It is the Government's intention to include in the final Bill a requirement for Community Councils to produce an Annual Report on their activities. No views are particularly sought on this.

Training

6.3.4 A new proposal in the Consultation Document which does not appear in the Draft Bill but will appear in the final Bill is that Community Councils should consider the training needs of their staff and members and, where appropriate make that training compulsory. Views are sought on this proposal.

Performance Management

6.3.5 The White Paper contained a proposal that the Chair of a Community Council should set objectives for the Clerk. This does not appear in the Draft Bill but views are sought on whether the duty should rest with the Council as a whole or just the Chair, and whether it should apply to all Community Councils or only some.

Community Polls

6.3.6 The White Paper said that the framework for Community Polls would be clarified.

- 6.3.7 These are costly and the turnout generally low. The outcomes are not binding.
- 6.3.8 The Government intends to introduce into the final Bill a provision repealing the legislation regarding Community Polls and replace it with an obligation on County Councils to have a petitions scheme and an e-petition facility. This would allow communities to express their views without the cost of Community Polls and there would be a requirement on County Councils to respond. Views are sought on this issue.

Part 7 - Workforce Matters

<u>Chapter 1 – Guidance to Public Bodies on Workforce Matters</u>

7.1.1 This Chapter provides power for Welsh Ministers to issue guidance to public bodies on workforce matters. A public body must have regard to any such guidance issued to it. They must consult a public body or bodies before issuing guidance.

<u>Chapter 2 – Public Services Staff Commission</u>

- 7.2.1 A non-statutory Public Services Staff Commission has already been established. The Consultation Document suggests that the passing by the UK Government of the Wales Bill may, depending on the content, mean that a statutory staff commission would be more constrained on the matters on which it could offer workforce guidance than a non-statutory commission. The Consultation Document asks for views on whether it would still be desirable to establish a statutory commission even if it were more constrained.
- 7.2.2 Even if it is established, such a commission would only exist until 1st April 2021 unless extended in being by Welsh Ministers to a date not beyond 1st April 2023.

Part 8 - General

This contains interpretation clauses and technical clauses relating to powers to make transitional and consequential provisions.

- 8.1 It is made clear that reference to County Councils in the Draft Bill apply to the existing County and County Borough Councils until 1st April 2020.
- 8.2 The Bill, if enacted, will be known as the Local Government (Wales) Act 2017.

Number: WG26897



www.gov.wales

Welsh Government

Consultation Document

Devolution, Democracy and Delivery

Draft Local Government (Wales) Bill and Explanatory Memorandum

Reforming Local Government

Date of issue: 24 November 2015

Action required: Responses by 15 February 2016

Overview

This Welsh Government consultation seeks views on the content of the Draft Local Government (Wales) Bill, Draft Explanatory Memorandum and Draft Regulatory Impact Assessment.

The objective of the Draft Bill is to complete the programme of Local Authority mergers and set out a new and reformed legislative framework for Local Authority democracy, accountability, performance and elements of finance. It will also establish a statutory Public Services Staff Commission.

How to respond

Closing date for responses is **15 February 2016**.

Responses should be sent by e-mail or by post to the address below. Responses are welcome in English or Welsh.

Further information and related documents

Large print, Braille and alternative language versions of this document are available on request.

The consultation documents can be accessed via the Welsh Government's website at www.gov.wales/consultations

Contact details

For further information, please contact:

Reforming Local Government Welsh Government Cathays Park Cardiff CF10 3NO

e-mail: RLGProgramme@wales.gsi.gov.uk

Data protection

How the views and information you give us will be used

Any response you send us will be seen in full by Welsh Government staff dealing with the issues which this consultation is about. It may also be seen by other Welsh Government staff to help them plan future consultations.

The Welsh Government intends to publish a summary of the responses to this document. We may also publish responses in full. Normally, the name and address (or part of the address) of the person or organisation who sent the response are published with the response. This helps to show that the consultation was carried out properly. If you do not want your name or address published, please tell us this in writing when you send your response. We will then blank them out.

Names or addresses we blank out might still get published later, though we do not think this would happen very often. The Freedom of Information Act 2000 and the Environmental Information Regulations 2004 allow the public to ask to see information held by many public bodies, including the Welsh Government. This includes information which has not been published. However, the law also allows us to withhold information in some circumstances. If anyone asks to see information we have withheld, we will have to decide whether to release it or not. If someone has asked for their name and address not to be published, that is an important fact we would take into account. However, there might sometimes be important reasons why we would have to reveal someone's name and address, even though they have asked for them not to be published. We would get in touch with the person and ask their views before we finally decided to reveal the information.

Contents

Foreword by the Minister for Public Services	1
Introduction – What this consultation is about	2
The Draft Bill	4
Consequential Amendments	
Part 1: Local Government Areas and County Councils	5
Chapter 1: Local Government Areas	5
Chapter 2: Ordinary elections of Councillors and term of office	12
Chapter 3: New Councils: Finance	
Other matters	
Part 2: General Power of Competence	
Chapter 2: Community Councils with Competence	
Part 3: Promoting Access to Local Government	
Chapter 2: Public Participation in Local Government	
Chapter 3: Community Area Committees	
Chapter 4: Improvement Requests	
Chapter 5: Access to Meetings etc.	
Other matters	
Part 4: Functions of County Councils and their Members	
Chapter 4: Further Provision about Duties on Members	27
Chapter 7: Overview and Scrutiny Committees and Standards	07
Committees	
Other Matters	
Part 5: County Councils: Improvement of Governance	
Chapter 1: Duty to make arrangements to secure good governance et	
Chapter 2: Corporate Plans	
Chapter 3: Assessments of Governance Arrangements	
Chapter 6: Miscellaneous Provision	
Chapter 7: Role of the Corporate Governance and Audit Committee	
Other Matters	
Part 6: Community Councils	
Chapter 1: Review of Community Council Arrangements	
Chapter 2: Members of Community Councils to Complete Training	
Chapter 3: Community Council Election Dates	
Other matters	
Part 7: Workforce Matters	
Chapter 2: Public Services Staff Commission	47
Part 8: General and Schedules	
Schedules	48
Annex A: Peer assessment	49
Governance Themes	49
Reviewers	49
Example Process	
Annex B: Consultation Questions	53

Foreword by the Minister for Public Services

The White Paper *Reforming Local Government: Power to Local People* set out my vision for Local Government in Wales. This is based on activist Councils, engaged in delivering modern, accessible, high quality public services with their local communities. Councils should be acting as community leaders and agents of change, with leadership focusing on excellence, and Councils committed to looking outwards in their place-shaping role.

It also set out the vision for the different spheres of government in Wales and the relationship between them. We wish to see Local Government which is strong, flexible and empowered to deliver the strategic direction set by the Welsh Government in the light of local circumstances. The Well-being of Future Generations (Wales) Act 2015 provides the framework for coordinating the different parts of government in Wales. This Draft Bill and the programme of reform will contribute to better outcomes by ensuring Local Government services are sustainable, integrated and involve the people of Wales.

In June I announced the Welsh Government's preference for the future configuration of Local Government in Wales. Maps, with two options in respect of North Wales, were published alongside this announcement. The maps set out our preference for the future structure in South, Mid and West Wales whilst facilitating further discussion around North Wales. The case in North Wales is finely balanced between two or three Local Authorities. I welcome views, through this formal consultation, on all our proposals for Local Authority mergers.

I understand this is an unsettling time, and I am committed to ensuring the terms and conditions of Local Authority staff are protected so no-one will be disadvantaged by transfer to a new Authority. The Bill will, therefore, also establish a statutory Public Services Staff Commission, a proposal which has attracted widespread support from Local Government. The Public Services Workforce Partnership Council will remain at the heart of the vision to support the development of a world class public service workforce. It will be the primary reference point for the Commission.

This consultation is very much part of an on-going dialogue about the reform of Local Government. I want to hear from you whether our proposals for legislation will help to achieve the objective of creating the 'activist' Councils Wales needs. I would also be interested to hear if you think there is more we might do to achieve this objective.

Leighton Andrews AM Minister for Public Services

Introduction - What this consultation is about

The Welsh Government is consulting on a Draft Local Government (Wales) Bill ("the Draft Bill") which will complete the programme of Local Authority mergers and set out a new and reformed legislative framework for Local Authority democracy, accountability, performance and elements of finance. It will also establish a statutory Public Services Staff Commission.

We have previously consulted on proposals to reform Local Government. The three White Papers *Reforming Local Government* (July 2014), *Public Services Staff Commission* (October 2014) and *Reforming Local Government: Power to Local People* (February 2015),¹ set out the reasons why we are proposing change and our proposed approach to the reform of Local Government. Full details of the consultation, responses received and links to the documentation are given in the Explanatory Memorandum to the Draft Bill.

As part of the package of reforms, the first Local Government (Wales) Bill was passed by the Assembly on 20 October. It is anticipated that Royal Assent will be received during the consultation period. This legislation makes provision for voluntary mergers and puts in place preparatory legislation for further mergers and reform. The legislation includes provisions relating to Transition Committees, electoral arrangements, remuneration arrangements, and restraints of transactions and recruitment etc.

Having taken into account the issues and views raised through consultation, we are now publishing the Draft Local Government (Wales) Bill, Draft Explanatory Memorandum, Draft Regulatory Impact Assessment and specific impact assessments for a further period of consultation. We will take account of comments received during this consultation as we prepare the Bill and accompanying documents for introduction into the next Assembly, after May 2016.

This Paper sets out additional policy explanation where we have modified or decided not to proceed with the proposals in the *Power to Local People* White Paper, additional proposals we wish to consult on, and matters which we have not included in the Draft Bill but intend including in the Bill for introduction. Where the provisions in the Draft Bill are substantially the same as the proposals in the White Paper, further explanation is given in the Explanatory Memorandum and is not repeated here. This Paper should therefore be read in conjunction with the Draft Bill and the accompanying documents. In structure, this Paper follows the Draft Bill rather than the White Paper.

http://gov.wales/consultations/localgovernment/power-to-local-people/?status=closed&lang=enhttp://gov.wales/consultations/improving/public-services-staff-commission-consultation/?lang=enhttp://gov.wales/consultations/improving/public-services-staff-commission-consultation/?lang=enhttp://gov.wales/consultations/improving/public-services-staff-commission-consultations/improving/public-services-staff-commission-consultations/improving/public-services-staff-commission-consultations/improving/public-services-staff-commission-consultations/improving/public-services-staff-commission-consultations/improving/public-services-staff-commission-consultations/improving/public-services-staff-commission-consultations/improving/public-services-staff-commission-consultation-consu

¹ Welsh Government | Devolution, Democracy and Delivery White Paper – Reforming Local Government

In this document, a reference to "the White Paper" is a reference to the *Reforming Local Government: Power to Local People* White Paper, unless stated otherwise.

The Draft Bill

The proposals contained within the Draft Bill will complete the programme of Local Authority mergers and set out a new and reformed legislative framework for Local Authority democracy, accountability, performance and elements of finance. It will also establish a statutory Public Services Staff Commission.

The purpose and intended effect of the provisions outlined in the Draft Bill can be found at Chapter 3 of the Explanatory Memorandum. Explanatory Notes for the provisions of the Draft Bill can also be found at Annex A to the Explanatory Memorandum. The following sections provide an update on proposals which were outlined in the White Paper, and new or changed proposals which we would like your views on.

Consequential Amendments

The Draft Bill does not include comprehensive consequential amendments or transitional arrangements. These will be drafted for inclusion in the Bill on introduction. For example, there are some statutory planning requirements that are linked to the ordinary elections of Councillors (by 'ordinary' we mean the regular election of the full Council). The intention is that transitional arrangements will be made to ensure that the ordinary elections to the Shadow Authorities of the new Authorities in 2019 do not trigger the review of strategies required under the Violence against Women, Domestic Abuse and Sexual Violence (Wales) Act 2015 or the preparation of Local Well-being Plans required under the Well-being of Future Generations (Wales) Act 2015.

Part 1: Local Government Areas and County Councils

Part 1 of the Draft Bill contains the provisions which will establish new Counties and their Councils. Chapter 1 specifies the Local Government areas in Wales. Chapter 2 provides for the constitution and election of the new Councils. Chapter 3 provides for establishment of the new Councils and Chapter 4 makes miscellaneous provision.

Question 1.1: Do you have any comments on any of the provisions in Part 1 of the Draft Bill?

We also welcome your views on the following matters.

Chapter 1: Local Government Areas

The Commission on Public Service Governance and Delivery ("the Commission") recommended reducing the number of Local Authorities in Wales through a series of mergers. The arguments for this recommendation and supporting evidence were set out extensively in the Commission's Report. The Welsh Government accepted these arguments in a White Paper published in July 2014, *Devolution, Democracy and Delivery: Reforming Local Government*. This document set out the Welsh Government's preference to merge Local Authorities on the basis of the Commission's Option 1, which would result in 12 new Authorities. It also signalled that Ministers remained open to considering alternative configurations.

The White Paper Reforming Local Government: Power to Local People highlighted the intention to proceed with mergers of Local Authorities through this Draft Bill. On 17 June 2015, the Welsh Government announced its preference for the future configuration of Local Government in Wales, based on eight or nine new Principal Local Authorities. These alternatives are set out in two tables in Schedule 1 to the Draft Bill, the difference between them being the configuration of the proposed new Counties in North Wales:

Table 1

_

Name	Area	
County 1 ²	Isle of Anglesey, Gwynedd and Conwy	
County 2	Denbighshire, Flintshire and Wrexham	
County 3	Ceredigion, Pembrokeshire and Carmarthenshire	
County 4	Swansea and Neath Port Talbot	
County 5	Bridgend, Rhondda Cynon Taf and Merthyr Tydfil	
County 6	Cardiff and the Vale of Glamorgan	
County 7	Blaenau Gwent, Caerphilly, Torfaen,	
	Monmouthshire and Newport	
Powys	Powys	

² The naming of the new Counties is dealt with below.

Table 2

Name	Area	
County 1	Isle of Anglesey and Gwynedd	
County 2	Conwy and Denbighshire	
County 3	Flintshire and Wrexham	
County 4	Ceredigion, Pembrokeshire and Carmarthenshire	
County 5	Swansea and Neath Port Talbot	
County 6	Bridgend, Rhondda Cynon Taf and Merthyr Tydfil	
County 7	Cardiff and the Vale of Glamorgan	
County 8	Blaenau Gwent, Caerphilly, Torfaen,	
	Monmouthshire and Newport	
Powys	Powys	

In setting out its preference, the Welsh Government emphasised there would be an opportunity for members of the public, interested organisations and stakeholders to put forward their views formally on the proposals. The following section sets out a range of issues associated with the proposed mergers and formally invites views on the future configuration of Local Authorities in Wales.

The Issues

The Commission considered possibilities that would have yielded seven or eight very large Local Authorities, but advised against pursuing them. They felt that the need to sustain local democracy and coherence might act as a limiting factor on the extent of such change. However, we believe that the proposals in the White Paper around greater public participation in democracy, more open and transparent decision making and greater diversity in Local Government, together with community area committees provide effective mechanisms for maintaining democratic engagement with communities in larger areas. It is also important to note there are many Authorities in other parts of the UK operating at the larger end of the scale of what is proposed in Wales.

On the matter of European Funding, the Commission identified an issue of alignment with the West Wales and the Valleys 'convergence' area, which has links to EU funding and state aid allowances. While a relevant consideration, we do not think that it should override a strategic, long-term case for mergers. There should be minimal impact on the delivery and related eligibility during the current 2014-2020 EU funding period. The majority of regions in the West Wales and the Valleys 'convergence' area remain in that area. The criteria for determining EU funding eligibility post-2020 is yet to be agreed and the Welsh Government will continue to press for the best possible settlement in negotiations for any new EU programmes.

The Commission took a wide range of factors into consideration including the minimum extent of mergers necessary to systematically address problems of scale, alignment of Local Authority areas with Local Health Board and Police Force areas, and avoiding the disruption entailed by splitting up existing Local Authorities. Apart from one exception in terms of alignment with Local Health Board boundaries (Bridgend), the maps published in June 2015 are consistent

with the principles set out in the Commission's Report. They are a refinement of the Commission's Option 1, informed by further consideration of a range of issues including the views of individual Local Authorities.

Carmarthenshire and Swansea

In terms of Carmarthenshire and Swansea, the Commission report itself noted that 'matters are less clear elsewhere – in the Bridgend / Neath Port Talbot / Swansea and Carmarthenshire / Ceredigion / Pembrokeshire areas – and our preferred options differ only in how they would affect those areas. ³ The report made it clear that all of its 'favoured options have different strengths and weaknesses' and that it had no preference between them. ⁴

In the case of Carmarthenshire, a merger with Ceredigion and Pembrokeshire would help integrate health and social care in the Hywel Dda Local Health Board area and bring together an area which has some significant common features in terms of rurality, local economy and, in much of the area, use of the Welsh language. This configuration fits within the option of "larger" configurations as expressed in the Prospectus⁵ issued inviting voluntary merger applications, and provides a single, stronger Authority with greater capacity and capability. It would also be compliant with the Commission's Option 2b and 3.

As for Swansea, there is a question as to whether the Authority is of sufficient scale to remain alone and sustainable in the longer term. It is already closely engaged with Neath Port Talbot on a project bringing together social services. A merger with Neath Port Talbot would avoid the anomaly of retaining Swansea as a single entity while merging the city of Cardiff with the Vale of Glamorgan, as well as combining two areas with similar socio-economic conditions. There is also a strong argument that Swansea is the focal point for economic growth in the wider Western Bay wider area (see "Regional Footprint and Collaborations" section below), and that future development is focussed towards its eastern boundary with Neath Port Talbot (for example the new university development). Indeed, both Local Authorities argued with some force back in November 2014 as part of deliberations on voluntary mergers, that they should form the core of a new city region authority for the area (albeit encompassing in addition the Llanelli / Ammanford area and south west Powys). Clearly, breaching existing boundaries in this way does not meet our current preferred options. On the other hand, leaving out the additional areas does not undermine the argument of bringing the two existing Authorities together. A merger between Swansea and Neath Port Talbot would also be partially in line with the Commission's Option 2a and 3.

Bridgend

The Commission's report did not highlight any specific issues regarding Bridgend. In Option 1 it was included with Swansea and Neath Port Talbot in line with the principle that there should be alignment of Local Authority areas with Local Health Board areas and this remains a strong argument. However,

³ Commission on Public Service Governance and Delivery, 2014, p96.

5 Invitation to Principal Local Authorities to Submit Proposals for Voluntary Mergers

Tudalen 80

⁴ Commission on Public Service Governance and Delivery, 2014, p98.

while some feel Bridgend is in a 'buffer zone' between south east and south west Wales, there is evidence that points towards its natural position lying with the South East Wales City Region rather than to the west.

The Local Authority as part of its voluntary merger expression of interest presented evidence around retailing patterns, transport links and travel to work destinations which all pointed towards the south east rather than west. There are more socio-economic similarities with the Vale of Glamorgan or southern Rhondda Cynon Taf than with Neath Port Talbot. In terms of collaboration arrangements and partnership agreements, again the evidence suggests it operates more naturally within the south east Wales region. The most obvious examples include its membership of the Cardiff Capital City Region and the Central South Education Consortium. It is also a member of a number of Local Government established forums including the South East Wales Strategic Planning Group, South East Wales Regional Housing Forum, and the South East Wales Regional Partnership Board.

The key issue to consider is that a configuration comprising Bridgend / Rhondda Cynon Taf / Merthyr would cross the current Local Health Board boundary and thus move away from the principles of alignment and coterminosity. Our view is while there would be challenges to work through, this cross health boundary arrangement could be made to work in ways which could resolve the additional complexity. We consider there are no challenges which could not be overcome and there appears to be willingness within the Health Boards to do what is necessary to make any new arrangements work. While there are no plans currently to propose any changes to any Local Health Board boundaries, consideration will be given to the implications for Local Health Board boundaries at the point in the process where the conversation is complete and there is a confirmed map in place. Irrespective of the final map, the Welsh Government wants Local Health Boards to work much more closely together to provide hospital services across traditional boundaries regardless of where the formal boundaries are drawn.

Of course, any merger comprising Bridgend / Rhondda Cynon Taf / Merthyr would result in a significantly larger new Authority with commensurate gains in capacity and capability, but these would need to be balanced against the possible disincentive of health boundary issues. Finally, another factor to consider is that in practical terms, Bridgend occupies a particular geographical location and its position in a new structure will determine other choices. This was made clear by the proposed merger of Bridgend and the Vale of Glamorgan which would have prevented delivery of the Option 1 map as the city of Cardiff would have stood alone. In turn, this may have driven different configurations across South East Wales which may have impacted on the proposals in Gwent and the Valleys. A merger between Bridgend and Rhondda Cynon Taf / Merthyr would not prevent the delivery of a modified Williams Option 1 map.

South East Wales

The Commission, in recommending mergers between Newport / Monmouthshire and Blaenau Gwent / Caerphilly / Torfaen, nevertheless recognised there may be alternatives. It stated the Gwent Police Force area

contains a diverse mixture of urban, valleys and rural Authorities, of areas of very high and very low deprivation, and of relatively very high and very low Council Tax. The Commission noted other combinations of Authorities were possible. However, it was concerned alternative combinations would breach the criterion on EU convergence funding. It also felt creating Local Authority areas like the whole of Gwent or the whole of North Wales could lead to difficulties in meeting multiple diverse local needs effectively or to maintaining fair democratic representation within such areas, thus potentially jeopardising some of the gains from merger.

It is clear a Gwent area configuration in South East Wales would create the biggest Local Authority in Wales with a combined population of over 580,000. This would make it on a par with other populous Unitary Authorities in the UK such as Glasgow and Cornwall, but still significantly less than Authorities such as Birmingham and Leeds. Despite this and the variety of geographical and socio-economic conditions within the area, we have seen stronger than average collaboration and a distinct sense of a Gwent identity among the public services. It is important to highlight that Authorities in the Gwent areas pride themselves on having established effective collaboration and partnership arrangements in recent years and those in place are working well. For example, the five Local Authorities are part of the Education Achievement Service Consortium, the Gwent Regional Collaborative Committee (Supporting People), and are piloting a regional funding approach for violence against women, domestic abuse and sexual violence services. On top of this, both the Police Force and the Local Health Board operate on a Gwent area footprint.

Another factor to note is the considerable opposition in the Gwent area to the potential configurations proposed by the Commission. Three of the five Local Authorities have expressed clear views around merging with others or remaining independent, with Newport against merging with Monmouthshire and vice versa, and Caerphilly against merging with anyone. Given the strength of feeling and clear opposition to the configurations suggested by the Commission, there is an argument merging the five Local Authorities and creating a larger Authority may be an advantage, not least in terms of maximising economies of scale and giving the Authority a strong voice. It would also be Option 1 compliant, albeit as part of a larger unit.

North Wales

In North Wales, all of the options presented by the Commission included three Local Authorities. While it is the case that collaboration tends to happen on a North Wales regional basis (GwE Consortium, North Wales Economic Ambition Board, Supporting People etc.), the Commission dismissed the creation of a single Local Authority area covering the whole of North Wales. Doing so could be likely to lead to difficulties in meeting multiple diverse local needs effectively, or to maintaining fair democratic representation.

In addition, the EU convergence funding criterion was not consistent with the division of North Wales into two Local Authorities rather than three by merging Anglesey, Gwynedd and Conwy; and Denbighshire, Flintshire and Wrexham. Denbighshire could jeopardise its qualification for convergence funding by

merger with Flintshire and Wrexham which are not eligible. Setting this issue to one side, the case for either two or three Local Authorities in the North Wales area is finely balanced. Issues include:

- population if the configuration in North Wales was three Authorities, two
 of those would have populations around the 200,000 mark i.e. Gwynedd /
 Anglesey and Conwy / Denbighshire. Notwithstanding Powys, this would
 be considerably lower than most of the other configurations in Wales
 which would be 380,000 plus with Wrexham / Flintshire just below the
 300,000 mark;
- Welsh Language there is an important question as to whether it would be supported better by creating three or two Authorities in North Wales. Some argue that placing Denbighshire with the border Counties of Flintshire and Wrexham might impact adversely on maintaining and strengthening the vibrancy of the language in the Denbighshire area. On the other hand, merging Conwy with the Isle of Anglesey and Gwynedd could create an area where possibly less than 50% of the population are Welsh speakers. This would mean that no Authority in Wales would have a majority of Welsh speakers and could impact on current internal administrative practice in Gwynedd;
- the creation of two rather than three Authorities would reduce the variance across North Wales on financial indicators, particularly in terms of the key indicators such as Council Tax raising ability (size of tax base) and spending power (gross revenue expenditure);
- current sub regional delivery structures for police, fire and health currently operate on a three area basis – Anglesey / Gwynedd, Conwy / Denbighshire and Wrexham / Flintshire. The Health Board has recently introduced an area director structure based on these areas; and
- the strong preference expressed by Conwy that in the event mergers were to proceed, a merger between themselves and Denbighshire should be examined further given their supporting evidence.

The case in North Wales is, therefore, finely balanced between two or three Local Authorities and there are pros and cons for both.

A full analysis of the options for eight or nine new Authorities is presented in the accompanying draft Regulatory Impact Assessment (RIA). We would welcome your views on the Welsh Government's preference for the future configuration of Local Government in Wales.

Question 1.2: What are your views on the options for 2 or 3 Counties in North Wales, as set out in Schedule 1 to the Draft Bill?

Question 1.3: What are your views on the proposed configuration of Local Government areas in Wales?

Powys

The Commission on Public Service Governance and Delivery recommended that because of the unique characteristics of the County of Powys and the

distinctive patterns of service delivery this creates, Powys County Council (PCC) and Powys Teaching Health Board (PTHB) should merge.

In considering the available options, Ministers recognise the difficulties in successfully merging two organisations with such different functions, governance arrangements and legislative basis. As a result, the Welsh Government believes the funding and governance issues mean a formal merger is not the best solution, but work should continue to encourage and explore other opportunities for greater front-line and strategic integration, including where these opportunities could result in the need to legislate to remove barriers or facilitate further progress.

PCC and PTHB are already co-terminous organisations, serving the same population, largely experiencing the same challenges and opportunities of a geographically large, sparsely populated, highly rural County. The organisations have a track record of working together to develop services for the people of Powys and have a history of working with communities, the voluntary sector and other stakeholders and partners to deliver improvements. This resulted in the formulation of the One Powys Plan 2014-17.

In November 2014, a joint Expression of Interest was submitted to the Welsh Government in which both organisations sought support to progress integration. In spring 2015, the Minister for Public Services agreed to provide Welsh Government support in response to proposals put forward by PCC and PTHB in *Working Together for a Thriving Powys*. This agreement includes a funding contribution towards the work.

Since November 2014, both Chief Executives together with their executive teams have been meeting regularly as a Joint Management Team. Research has been undertaken to identify what lessons could be learnt from other organisations already undertaking this level of integration. They have developed an integration plan which sets out the priorities for action. Their first priority is to create health and social care teams in the community. These could include staff such as district nurses, speech and language therapists, social workers, physiotherapists, and occupational therapists. These teams will work closely with local GPs to offer care in, or as close to people's homes as possible using technology where practicable.

Question 1.4: Do the Welsh Ministers need to seek any further powers to support the integration of Powys Teaching Health Board and Powys County Council?

Naming the New Counties

With the exception of Powys, which is a continuing Authority, the proposed new Counties are simply referred to as County 1 etc. in the Tables presented in Schedule 1 to the Draft Bill. The intention is that the Shadow Authorities, to be elected in 2019 will determine the names of their new County (in English and Welsh forms). The regulation-making power in paragraph 2(1) of Schedule 1 will enable the Welsh Ministers to give legal effect to the names determined by the Shadow Authorities.

Question 1.5: What are your views on the procedure for naming the new Counties?

Regional Footprint and Collaborations

In 2011 a common public service Regional Collaborative Footprint (RCF) was established based on six areas: North Wales, Mid & West Wales, Western Bay, Cwm Taf, Cardiff and Vale, and Gwent. The Commission on Public Service Governance and Delivery noted that the boundaries of most major service providers were reasonably well-aligned, but that despite the development of the RCF, regional collaborations were still too complex, fragmented and hard to administer. They argued that aligning public service collaborations would simplify working arrangements, avoid duplication and help create more integrated services for the public.

In July 2014, the Welsh Government published its vision for reforming devolved public services in the document 'Devolution, Democracy and Delivery: Improving Public Services for People in Wales'. One of the 20 high-level actions outlined in the document focuses on the importance of aligning existing collaborations with the boundaries of new, merged Local Authorities and other delivery partners.

Since then, work has been undertaken across the Welsh Government to identify the most strategic regional collaborations. The information received has highlighted that current collaborations and partnerships are broadly well-aligned with the proposed Local Government areas and are consistent with the Welsh Government's RCF, particularly in areas such as North Wales and Gwent.

The Welsh Government does not propose to take any action on aligning collaborations and partnerships ahead of a final decision on the Local Authority areas. In the meantime however, Ministers have agreed that there should be a moratorium on the establishment of any new collaborations and partnerships prior to finalising the map. Such a moratorium does not apply to any existing statutory requirements. For example, it will not impact Public Services Boards where provisions under the Well-being of Future Generations (Wales) Act 2015 are due to be brought into force in April 2016. The same is true for planned regulations under the Social Services and Well-being (Wales) Act 2014, such as for the establishment of the Social Services Regional Partnerships.

Chapter 2: Ordinary elections of Councillors and term of office

In accordance with the Fixed-term Parliaments Act 2011, from 2015 Parliamentary general elections are held every five years on a fixed term basis. The Wales Act 2014 likewise moved the National Assembly for Wales elections to a five-year fixed term, with effect from 2016. In accordance with those cycles and to avoid clashes, we are proposing to create five-year fixed term cycles for elections to Local Government (Principal and Community Councils), with effect from 2023.

Before moving to this five year cycle, we propose making various transitional arrangements as follows:

- The term of office of Councillors elected to existing Principal Authorities (Counties and County Boroughs) in the Local Government elections in May 2017 will be three years. The existing Authorities will be abolished from 1 April 2020. This is not the case for Powys which will be a continuing Authority, details of which are found below.
- Elections to the new County Councils will be held in May 2019 and Councillors will serve a four year term. For the first year of that term, the new County Councils will exist as Shadow Authorities, which will exercise only preparatory functions specified in regulations made by the Welsh Ministers. The new County Councils will take over the full range of Local Government functions from 1 April 2020. The Draft Bill provides that the next ordinary elections of the new County Councils will be in May 2023. This will give the new County Councils three years from 1 April 2020 in which to consolidate their new arrangements.
- Holding elections in 2017 and 2023 in Powys, as a continuing Authority, would mean Councillors elected in 2017 would serve a sixyear term. This would be an unprecedented length of term and to avoid straining the democratic mandate, ordinary elections will be held in Powys in 2020, meaning two three-year terms for Councillors. Community Councillors in Powys will, however, serve a six-year term between 2017 and 2023 (see Part 6 below).

4 May 2017	(i) Elections to continuing Authorities not subject to merger	(i) i.e. Powys. These Councillors will serve 3-year terms until May 2020
	(ii) Last elections to all Authorities which would be subject to merger	(ii) These Councillors will serve 3-year terms until 1 April 2020
May 2019	First elections to new County Councils	New County Councils exist in shadow form until Vesting Day on 1 April 2020. These Councillors will serve a 4 year term
May 2020	Elections to continuing Authorities	I.e. Powys. These Councillors will serve 3-year terms until 2023
May 2023	Elections to all Local Authorities in Wales	All Councillors will serve a 5 year term until May 2028

Question 1.6: What are your views on the proposed changes to the Local Government election timetable?

Chapter 3: New Councils: Finance

The Draft Bill includes legislative changes in relation to finance which are needed to support structural mergers and some of the wider reforms proposed elsewhere in the Draft Bill. The White Paper discussed the need to review the Local Government finance system over a longer timetable than that planned for mergers and the reforms set out in the Draft Bill. This proposal was accepted by most respondents, with the majority acknowledging the risks of fundamental change to the finance system too soon in the reform process.

The Welsh Government intends to consult on proposals including separate legislation dedicated to the mechanisms for distributing, raising, managing and accounting for the funding of Local Government in Wales. This will enable us to design a system which takes account of wider changes to the powers and fiscal responsibilities of the Assembly, and devolves greater financial independence and responsibility to Local Authorities.

Accounts and Audit Requirements

The Bill for introduction will contain updated and modernised provisions setting out how Local Government bodies in Wales should account for the money they spend and how their accounts will be audited. Local Government bodies in Wales abide by the requirements set out in regulations made under the Public Audit (Wales) Act 2004. We will take the opportunity provided by this Bill to modernise the accounts and audit requirements of local bodies in Wales in line with the latest good practice for financial management and internal control systems.

The Welsh Ministers will also make provision in regulations in relation to the mechanism by which Shadow Authorities will be funded during the shadow period. The regulations will include the processes a Shadow Authority will be required to undertake in relation to monitoring its internal resources and, if deemed necessary, the keeping of, any audit of accounts.

Council Tax

One of the core principles of local taxation is that tax-payers living within the same Local Authority area contribute equivalent amounts (relative to their circumstances) for the provision of local services in that area. Each Local Authority has democratic responsibility for deciding on the amount of Council Tax charged within its area. This means when existing Authorities are brought together to form a new larger Authority, each new Authority will need to set its own level of Council Tax.

In some cases, this may require a process of harmonisation to move to the same level of Council Tax across an Authority's area. This would not necessarily need to be achieved immediately upon the creation of a new Authority. For example, a period of transition could help to promote stability in

tax revenues and to ensure residents are protected from sudden changes in Council Tax liability. Conversely, early harmonisation could result in a fairer system for those taxpayers who would see a reduction in their bills. The existing finance legislation contains a broad range of powers which could be used to effect a number of policy approaches to the setting of Council Tax. There are also existing powers for Ministers to limit increases in Council Tax. We do not propose the Welsh Ministers take any new powers in relation to this and have not, therefore, included new provisions in the Draft Bill.

The Welsh Government is committed to consulting fully on any proposed future approach to Council Tax when the time is right. In light of the changing financial landscape, we consider it too early in the reform process to draw up firm plans for harmonisation, but we will examine and monitor the developing picture across Wales. We will expect Transition Committees and Shadow Authorities to carefully assess the particular characteristics of the new Authority, as well as consider the financial pressures on households, and use this information to inform immediate budgets and medium term plans for Council Tax. At the very least, we expect Transition Committees and Shadow Authorities to safeguard against any divergence in Council Tax levels in the period running up to mergers. The restrictions on certain transactions set out in the first Local Government (Wales) Bill will assist in the process of financial management and further guidance will be issued in due course.

Controls on Avoidance of Non-Domestic Rates

The White Paper recognised there are inconsistencies between local tax systems which we may wish to review as part of shorter term improvements to the finance system. We consider one of those inconsistencies to be the absence of a duty on Non-Domestic Rates-payers to notify their Billing Authority of a change in their circumstances which may affect their liability, or eligibility to claim a relief, discount or exemption. Although valuation officers can serve notices on rate-payers to provide information, there is no requirement for rate-payers to inform the valuation officers or the Billing Authority of a change in circumstances. This differs from some of the requirements placed on Council Tax payers and can result in large backdated bills when changes are made. The summary of responses to the recent discussion paper on rates avoidance⁶ produced by the Department for Communities and Local Government and HM Treasury indicates that one of the common avoidance measures is rate-payers notifying Billing Authorities retrospectively of periods of occupation, which are subsequently difficult to verify.

The extent of avoidance in the Non-Domestic Rates system in Wales, whether unintended or deliberate, is not currently known and will require further investigation. However, recent analysis undertaken by the Local Government Association estimates that in England around £230 million (approximately 1%) of rates revenue is lost to avoidance each year. ⁷ As this is a complex and

⁶ Business Rates Avoidance: Summary of Responses, Department for Communities and Local Government, HM Treasury, July 2015. https://www.gov.uk/government/consultations/business-rates-avoidance-discussion-paper

⁷ Local Government Association response to *Business Rates Avoidance* discussion paper, February 2015.

developing area of policy, we propose the Bill on introduction contains a provision to enable Welsh Ministers to make regulations in the area of Non-Domestic Rates avoidance.

We do not envisage providing for a detailed policy approach in the Bill as we will wish to develop the proposals in consultation with Local Government, rate-payers, the Valuation Office Agency, the Valuation Tribunal for Wales and the Lord Chief Justice Office. Full consideration will also be given to the appeals process for rate-payers aggrieved by a decision and to reasonable protocols for enforcement by Billing Authorities. In drawing up detailed proposals, we will take into account the available evidence, the findings of the Business Rates Panel, the outcome of a review on rates avoidance in England and powers proposed in the UK Government's Enterprise Bill to facilitate the sharing of more information between the Valuation Office Agency and Billing Authorities. Any regulations developed in due course would be subject to the affirmative legislative procedure, meaning the detailed proposals would undergo specific scrutiny before being debated by the National Assembly for Wales.

The policy development process will be consistent with the Welsh Government's approach to tax collection and management and the principles for tax policy published in 2014. Measures to tackle avoidance and fraudulent activity support the Welsh Government's principles of fairness to all taxpayers, by ensuring that the rates due are responsive to changes in circumstances and that liability is spread fairly across all taxpayers, of having clear rules which seek to minimise administration costs by improving the accuracy of information, and of providing stability and certainty to taxpayers by reducing the instance of backdated liabilities.

Question 1.7: Do you have any general comments on the provisions in section 16 and Schedule 3 of the Draft Bill relating to Local Government finance?

Question 1.8: How could the Welsh Government measure the current level of avoidance of Non-Domestic Rates?

Question 1.9: Do you have any comments or suggestions on how future legislation could help to reduce instances of avoidance of Non-Domestic Rates?

Question 1.10: In what other ways could the Welsh Government enable Local Government to reduce the level of avoidance and fraud within the Non-Domestic Rates system?

http://www.local.gov.uk/documents/10180/6869714/Business+rates+avoidance+-+discussion+paper+-+LGA+response.pdf/89897cc8-7bba-4257-a97b-243bf6d22ece

http://gov.wales/docs/det/publications/150202-devolution-of-business-rates-en.pdf

Tax powers, purpose, principles and priorities, Welsh Government, December 2014. http://gov.wales/docs/caecd/publications/150109-treasury-paper-en.pdf

⁸ The Devolution of Business Rates to Wales, February 2015.

Other matters

Phased Elections

The White Paper sought views on whether Local Government elections should be conducted in phases. In response to the concerns of stakeholders, the Welsh Government will not be proceeding with this proposal.

The Number of Councillors

In recognition of the democratic requirements of the new Authorities, the previous cap of 75 Councillors per Authority has been removed. Consultation on Draft Directions to the Local Democracy and Boundary Commission for Wales took place between August and November 2015. This consultation sought views on what an appropriate number of Councillors might be for the new Authorities to best deliver effective representation and democratic governance. We are currently considering the responses to this consultation.

The Preserved Counties

The preserved counties of Wales are the counties created by the Local Government Act 1972 ("the 1972 Act"). The counties created under the 1972 Act were abolished by the Local Government (Wales) Act 1994 ("the 1994 Act"). However, the 1994 Act created the concept of preserved counties based on the areas of the eight counties, to be used for a range of purposes largely relating to judicial and administrative purposes. The eight preserved counties are notional entities only; they have no elected Councils, no administrations and no functions of their own.

Since the 1994 Act, most of the purposes for which the counties were preserved have fallen away. For example, justice areas are now based on new geographical areas rather than the preserved counties. In some instances there remain obsolete references to "preserved counties" in legislation which will require amendment. However, there are references to preserved counties which remain relevant. They relate to lieutenancies, shrievalties and sea fisheries.

Lord Lieutenants and High Sheriffs

Lord-Lieutenants and High Sheriffs have traditionally been appointed by reference to local administrative areas. At the time of Local Government reform in 1972, the Local Government Act 1972 ("the 1972 Act") required the Queen to appoint a Lord-Lieutenant for each County in England and Wales. High Sheriffs are appointed annually by the Queen under the provisions of the Sheriffs Act 1887 and the 1972 Act put in place transitional arrangements so that High Sheriffs were appointed to the new counties established under the 1972 Act. Lord-Lieutenants and High Sheriffs continued to be appointed for the counties established under the 1972 Act between 1974 and 1996.

As stated above, the Local Government (Wales) Act 1994 reformed Local Government in Wales, but preserved the counties established under the 1972 Act for certain purposes. This included the geographical area for which Lord-Lieutenants and High Sheriffs would be appointed.

Today, the preserved counties continue to serve as the areas for defining the lieutenancies and shrievalties in Wales, in accordance with the Lieutenancies Act 1997 and the Sheriffs Act 1887. There is a Lord-Lieutenant and a High Sheriff for each preserved county in Wales, making eight Lord Lieutenants and eight High Sheriffs in total. Some Lord-Lieutenants and High Sheriffs cover two or more of the existing administrative Local Authorities created by the 1994 Act. The lieutenancy and the shrievalty of Gwent cover five Local Authorities (Blaenau Gwent, Caerphilly, Monmouthshire, Newport and Torfaen); in contrast, the lieutenancy and the shrievalty of Powys cover one Authority only, the administrative County of Powys.

Sea Fisheries (Shellfish) Act 1967

Section 10 of the Sea Fisheries (Shellfish) Act 1967 ("the 1967 Act") states that the portion of the sea shore to which an order under section 1 (power to make orders as to fisheries for shellfish) of that Act relates, shall for all purposes of jurisdiction be deemed to be within the body of the adjoining county, borough or burgh. The 1994 Act provided that the reference to "county" in section 10(1) of the 1967 Act in relation to Wales includes a reference to a preserved county.

The Welsh Government's proposals for Local Government mergers will result in the number of Principal Local Authorities being reduced to eight or nine, from 1 April 2020. Whichever of the two options is decided upon, there seems little purpose in retaining an equivalent number of preserved counties, to add unnecessary complexity to the Local Government map of Wales.

We propose to abolish the preserved counties in Wales with effect from 1 April 2020. In consequence, it will be necessary to amend the relevant statutes to ensure the appointments of Lord-Lieutenants and High Sheriffs relate to the new Counties in existence after 1 April 2020 and that the reference in the 1967 Act is updated. Lord-Lieutenants and High Sheriffs are appointed by the Queen, so any changes would be made in consultation with office-holders and the Queen's advisers. We will work with all affected to ensure that, where necessary, transitional arrangements are in place before 1 April 2020.

Question 1.11: Do you agree that the preserved counties be abolished and that consequential amendments are made so that the appointments of Lord-Lieutenants and High Sheriffs are made in respect of the Counties in existence after 1 April 2020?

Other Technical Matters

There are a number of issues which may be considered technical in nature on which we will be seeking views. These include:

- any property, rights and liabilities, including grants, which require specific provision for their effective transfer to the new Councils;
- any impact on existing contracts and property which would need express provision in the Bill;

- the length of time for suspension of by-elections in existing Councils prior to the new Councils assuming their full functions and responsibilities on 1 April 2020 e.g. where there are casual vacancies;
- mechanisms to preserve historic ceremonial rights, including city and borough status; and
- joint committees, joint boards and port health authorities.

It is our intention to engage directly with Local Authorities on these matters, however, we welcome views on these and any other similar technical matters as part of this consultation.

Question 1.12: Are there other matters of a technical nature which should also be considered?

Part 2: General Power of Competence

The provisions in Chapter 1 of Part 2 relate to County Councils' general power of competence and those in Chapter 2 set out the conditions which Community Councils must meet in order to be Community Councils with competence.

Question 2.1: Do you have any comments on any of the provisions in Part 2 of the Draft Bill?

We also welcome your views on the following matters.

Chapter 2: Community Councils with Competence

The Draft Bill includes conditions which a Community Council must meet if it is to resolve itself 'with competence'. Being 'with competence' would enable a Community Council to use the general power of competence and provide other organisations with a degree of confidence that the Council has the capacity and capability to 'do business'. There will also be a reputational benefit, with communities able to see that their Council is meeting certain standards.

The consultation on the White Paper established that there were clear views against the proposal that one of the competency tests should be a minimum annual turnover of £200,000. In particular, this was seen to disadvantage rural Councils. The Welsh Government accepts the argument and no such provision is included in the Draft Bill.

We also stated in the White Paper our intention that a Community Council resolving itself competent should be required to notify a committee of the Principal Authority for the area, and that the nominated committee should have powers to revoke the Community Council's competency qualification in some circumstances. More consultation responses were in favour than against these proposals. However, we have considered the implications this might have for the relationship between Principal Councils and Community Councils, notably the risk of blurring accountability and of creating tensions in the working relationship between the two tiers of Local Government. Therefore, we have taken a different approach in the Draft Bill.

The provisions in the Draft Bill enable a Community Council to pass a resolution at any meeting of the Council that it meets the competency requirements and that it is a Community Council 'with competence'. Such a Community Council will remain competent until the first annual general meeting following ordinary elections, when it must pass a resolution that it continues or ceases to be a Community Council with competence. A Community Council with competence may also pass a resolution at any meeting of the Council that it ceases to be a Community Council with competence. A Council that ceases to be a Community Council with

competence may continue any activities it has commenced whilst exercising the general power of competence.

This means that a Council would be able to resolve itself competent at its first meeting following an election and remain competent for the five years until its annual general meeting following the next elections, even though in each of the intervening years it might fail to meet any or all of the three competency requirements, including the requirement to have two years' unqualified accounts. During this period the Community Council would be able to exercise the general power of competence and do things it could not otherwise do if it had not resolved itself competent, as is the case with eligible Parish Councils in England. This approach would provide certainty for Community Councils undertaking activities in reliance on the general power and certainty for third parties in their dealings with Community Councils as to the extent of a particular Council's powers. The Welsh Government would welcome your views about this.

Question 2.2: Do you have any comments on our proposals relating to Community Councils with competence?

Part 3: Promoting Access to Local Government

This Part contains provisions relating to promoting access to Local Government. Chapter 1 gives an overview of the provisions. Chapter 2 requires Local Authorities to encourage public participation in Local Government. Chapter 3 establishes community area committees, for the purpose of ensuring that community interests and priorities are taken into account by the Council in exercising its functions. Chapter 4 deals with improvement requests, by which a Council enters into discussions with community bodies for the purpose of improving local outcomes. Chapter 5 extends public access to Local Authority meetings and Chapter 6 requires Local Authorities to publish a guide to their constitution and publish the official addresses of their members. The provisions relating to public participation, community area committees and improvement requests support the Welsh Government's *Principles for Working with Communities*. ¹⁰

Question 3.1: Do you have any comments on any of the provisions in Part 3 of the Draft Bill?

We also welcome your views on the following matters.

Chapter 2: Public Participation in Local Government

The White Paper makes a number of references to the need for Local Authorities to be more active in involving the public in the Council's work and in decisions which affect them. This part of the Draft Bill draws together the proposals for public participation into a single overarching duty on Local Authorities to encourage participation in the decision making of County Councils and their connected authorities (Community Councils, Fire and Rescue Authorities and National Park Authorities) and to set out how it means to achieve this in a strategy. The participation strategy must address promoting awareness of the Council's functions and how to become a member of the Council, facilitating access to information about decisions and ways for the public to make representations to the Council, public involvement in scrutiny and the use of social media.

Given the importance of the budget setting process, separate provision is made to ensure wide and meaningful consultation on the annual budget.

Question 3.2: Do you have any comments on the proposed public participation duty and the requirement to consult on the annual budget?

¹⁰ See Welsh Government Written Statement, 15 September 2015, http://gov.wales/about/cabinet/cabinetstatements/2015/workingcommunities/?lang=en

Chapter 3: Community Area Committees

In the White Paper we made a case for establishing a model of community governance so communities could maintain and improve their engagement with the larger, merged Local Authorities, with a leading role for Elected Members. We described an area-based approach, with 'area boards' made up of Elected Members, community bodies, the third sector, Community Councils and other public services.

Respondents to the consultation asked for more detail about what is intended, showed concern about the potential for increasing bureaucracy, and the importance of ensuring appropriate representation, especially from the third sector and Community Councils. The majority agreed that Local Authorities should have flexibility to design an approach suitable to their area. The public have also told us that, while they support bigger Local Authorities, they are concerned about them losing touch with communities.

The provisions included in the Draft Bill seek to articulate our intentions and provide a more informed basis from which to comment. They set out the intention to require Local Authorities to establish a committee in each of their community areas (as determined under the Well-being of Future Generations (Wales) Act 2015). Community area committees will provide a structured way for views on local priorities to be expressed and fed into the Local Authority budget planning process, and for communities and local people to engage with their Elected Members on practical matters of local importance. Local Authorities will also be able to seek the advice of these committees on any matter relating to the committee's community area. The provisions also enable the Local Authority to delegate functions to community area committees. Which functions may be delegated to community area committees will be determined by regulations to be made by the Welsh Ministers.

Question 3.3: How should community representatives to sit on community area committees be sought and selected?

Question 3.4: Do you agree County Councils should be able to delegate functions to a community area committee? If yes, are there any functions that should or should not be capable of being delegated?

Question 3.5: Do you have any views on whether transitional arrangements need to be put in place for existing area committees, or is a good lead-in time sufficient?

Chapter 4: Improvement Requests

In the White Paper we consulted on a proposal that there should be a 'right to participate'. This would enable community bodies to become involved in service improvement. The Draft Bill includes provisions which oblige Local

Authorities to enter into a dialogue with community bodies about how an outcome can be improved on receiving a request from a community body, unless they have reasonable grounds for not doing so. The definition of community bodies is widely drawn.

The procedure sets out that at the end of a period of dialogue, the Local Authority will publish on its website a summary of the discussions and the actions that have been agreed. We will expect both Local Authorities and community bodies to hold to the matters they have agreed publicly. However, we do not consider it would be conducive to good relations between Local Authorities and community groups or for fostering a culture of involvement if we were to impose more heavily prescribed duties on the Local Authority.

Question 3.6: Do you have any comments on the revised provisions for 'improvement requests' or on the interaction between these provisions and those relating to the public participation duty (Part 3, Chapter 2) and community area committees (Part 3, Chapter 3)?

Chapter 5: Access to Meetings etc.

In addition to the matters we consulted on in the White Paper and which are provided for in Part 3, Chapter 5 of the Draft Bill, we also wish to consult on various additional proposals relating to Council meetings.

- Electronic publication of notices of meetings. The Local Government (Democracy) (Wales) Act 2013 introduced a requirement on Community Councils to have websites and publish information concerning the Council. We propose that Principal Councils should be placed in a similar position to Community Councils.
- Electronic summons. We propose that Principal Councils should be able to choose to send out notices only by electronic means if they so wished.
- Prohibition of alcohol at Community Council meetings and removal of the restriction on having meetings in licenced premises. Society has changed considerably since the provisions of the Local Government Act 1972. Many more premises are licensed now and a great deal of public activity, such as weddings and conferences, takes place in places licensed to sell alcohol. The prohibitions, therefore, seem dated and we propose to repeal them. It would, however, be proposed that Community Councils may adopt standing orders which would prohibit Councillors or members of the public from consuming alcohol during a meeting.
- A regulation making power for the Welsh Ministers to require meetings
 of the Executive to keep and maintain minutes. At the moment,
 regulations made under the Local Government Act 2000 require that
 records of executive decisions are written, maintained and published.
 The proposed regulatory power would enable the Welsh Ministers to
 make meetings of Council Cabinets the subject of recorded minutes.
 We would welcome your views on this proposal in the light of our
 intention that Cabinet meetings would be subject to live broadcasting.

Question 3.7: Do you have any comments on any of our further proposals relating to access to meetings?

Other matters

Youth Councils

The White Paper proposed requiring the Chief Executive to establish a Youth Council, with the objective of encouraging better participation in democracy by children and young people. Following the consultation, further consideration has been given to this proposal, and it is now felt it is not sufficiently broad to reflect the best practice which already exists in many Local Authorities, in terms of engagement with children and young people, including through digital and social media.

Section 12 of the Children and Families (Wales) Measure 2010 ('the 2010 Measure') requires a Local Authority to make such arrangements as it considers suitable to promote and facilitate participation by children in decisions of the Authority which might affect them. Statutory guidance under the 2010 Measure sets out the Welsh Ministers' expectations that:

- Local Authorities should promote and facilitate children and young people's participation, within the broad context of the UN Convention on the Rights of the Child, as part of their policies, services and wider citizen engagement;
- children and young people's participation should be embedded into all aspects of planning, delivering and reviewing services, including local well-being plans; and
- Local Authorities should establish a county-wide youth forum or council, as a channel for young people's views to the Authority and other local and national decision-making bodies.

The guidance is currently being updated to reflect the commencement of the Well-being of Future Generations (Wales) Act 2015 from April 2016.

Whilst the 2010 Measure goes some way to fulfilling our policy aim of encouraging participation in local democracy, the public participation provisions in Part 3 of the Draft Bill would require a Local Authority to involve children and young people (amongst other 'local people') in the decision-making process of the Authority. This is a step further than section 12 of the 2010 Measure, which is limited to participation by children in decisions of the Authority 'which might affect them'. The role of children and young people would be emphasised in guidance to be issued under the Bill and we intend it to include specific measures which Councils could take to assist in the organisation of young people in their area and the channelling of their views into Council deliberations.

Question 3.8: Do you have any comments on our proposals to enhance participation by children and young people through the public participation duty?

Part 4: Functions of County Councils and their Members

The provisions in Part 4 of the Draft Bill deal with the functions of the new Councils and their members. Chapter 1 provides an overview of the provisions in this Part. Chapter 2 sets out the duties all Councillors must discharge. Chapter 3 sets out how breaches of the duties on Councillors are to be dealt with and Chapter 4 makes further provision in relation to these duties. Chapter 5 provides that the Elected Mayor or the Leader must set objectives for the Cabinet and that candidates for Elected Mayor or the Leader must prepare a written manifesto. It also enables Councillors to be appointed as assistants to the Executive. Chapter 6 makes provision about the appointment of the Chief Executive, setting his or her objectives, as well as making the post of Head of Democratic Services a chief officer. Chapter 7 makes various provisions relating to Overview and Scrutiny Committees and Standards Committees. Chapter 8 makes minor amendments to other legislation.

Question 4.1: Do you have any comments on any of the provisions in Part 4 of the Draft Bill?

We also welcome your views on the following matters.

Chapter 4: Further Provision about Duties on Members

In the White Paper we proposed that Group Leaders should be under a duty to ensure diversity is respected, and that Monitoring Officers and Standards Committees should have enforcement roles. The majority of consultees that responded on this issue were supportive. Chapter 4 puts a duty on leaders of political groups to promote and maintain high standards of conduct by the members of the group. Standards Committees are given new functions to monitor compliance by leaders of political groups with this duty and to advise and arrange training relating to the duty.

Chapter 7: Overview and Scrutiny Committees and Standards Committees

The Draft Bill gives Standards Committees new functions to handle complaints that Councillors have breached the duties imposed on them by Chapter 2 of this Part (sections 82 to 86), and to monitor compliance of leaders of political groups with the duty imposed on them by Chapter 4. Given the enhanced role of the Standards Committee, we believe there is merit in the Authority being provided with an overview of the work of the Standards Committee during the year, in all its functions, to gain a better understanding of trends in standards of conduct within the Authority. In Chapter 7, therefore, we put a duty on Standards Committees to publish an annual report and, if appropriate, make recommendations to the Authority.

Question 4.2: Do you have any comments on the proposed duty on leaders of political groups or the monitoring and reporting roles of the Standards Committee?

Other Matters

Delegation of Functions

In the White Paper we set out our intention to review the provisions in the Deregulation and Contracting Out Act 1994 ("the 1994 Act") with a view to allowing Local Authorities generally to make decisions on how they deliver services, other than in prescribed circumstances. This proposal was connected to the proposal to provide Principal Councils and eligible Community Councils with a general power of competence. Provision has been included within the Draft Bill for a general power of competence.

It is our intention, as is the case now under the 1994 Act, that Local Authorities should be permitted to delegate the exercise of certain of their functions to third parties. However, we consider the current provisions are inflexible. The contracting out power is not currently exercisable in relation to Wales by the Welsh Ministers, and the orders that have been made to date do little to encourage Local Authorities to be more innovative in the way they deliver their services.

We, therefore, propose that Part 2 of the 1994 Act, as it applies to the contracting out of functions of Local Authorities in Wales, should be repealed and replaced with a new regime that will allow for the delegation of Local Authority functions to third parties by regulations made by the Welsh Ministers. We propose this power of delegation should apply to Principal Councils, Community Councils, Fire and Rescue Authorities, National Park Authorities and Joint Planning Boards. As with the general power of competence, it is our intention to bring the provision into force at the earliest opportunity after the Bill is passed in order that the new regime is available to existing Authorities.

Question 4.3: Do you have any comments on our proposals in relation to the delegation of functions by Local Authorities?

Electoral Qualification

Respondents to the White Paper expressed concerns over lifting the prohibition on Local Authority officers standing for election to their own Authority. As previously announced, we will not be proceeding with this proposal.

In addition, the White Paper proposed:

- requiring candidates at Local Government elections to declare their membership of a political party, where appropriate;
- preventing Elected Members of Principal Councils from serving as a Community Councillor at the same time; and

 preventing a person from serving on more than one Community Council at the same time.

Further consideration is being given to these proposals and no provision is made in the Draft Bill.

Term Limits

There were clear views in the White Paper consultation regarding the proposal to limit the length of continual service of Elected Members, the Leader and Cabinet Members. As a result, and as previously announced, these proposals will not proceed.

Release of Councillors from Employment

The White Paper proposed placing a duty on devolved public services to release employees to undertake duties as Councillors. Given the provision in section 50 of the Employment Rights Act 1996 (right to time off for public duties) further provision has not been made for this in the Draft Bill. Raising awareness and promoting the use of this existing provision will be taken forward through the *Diversity in Democracy* programme.¹¹

Remuneration of Councillors

As stated in the White Paper, there is a justifiable expectation amongst the general public that elected politicians should receive no greater remuneration than can be justified in relation to the scale of their responsibilities and time commitment. Equally, the Welsh Government does not wish to financially disadvantage anybody for being a Councillor, as this would contradict our clear intention to increase diversity among Councillors.

The framework of remuneration for Councillors is decided by the Independent Remuneration Panel for Wales (IRPW). Although no provision is included in the Draft Bill, it is proposed the Welsh Ministers will have a power in the Bill for introduction to direct the IRPW to have regard to guidance when reviewing the remuneration framework for Councillors.

Question 4.4: Do you have any comments on our proposal to give the Welsh Ministers a power to direct the IRPW to have regard to guidance when reviewing the remuneration framework for Councillors?

Recall of Councillors

75% of online respondents agreed there should be a right of recall. Whilst five Local Authorities agreed (with conditions, especially safeguards against abuse), 13 were against. Although not included within the Draft Bill, provisions which introduce a right of recall over Members of Principal Councils are being considered, similar to arrangements that are in place in relation to Members of Parliament. Standards Committees would have their responsibilities strengthened by enabling them to consider allegations of a failure of a Councillor to perform his or her duties, as well as misconduct matters. In both cases, if a Councillor becomes subject to a suspension from office, this would enable local people, if they wished, to raise a petition calling for a by-election

¹¹ http://gov.wales/topics/localgovernment/diversity-in-democracy/?lang=en

in relation to that Councillor's seat. If a petition was signed by at least twenty percent of the electorate in that ward, a by-election would have to be called.

Remote Attendance at Meetings

The provisions enabling remote attendance at Council meetings, introduced in the Local Government (Wales) Measure 2011 ("the 2011 Measure"), have not been widely adopted. Feedback from Local Government has indicated either a lack of demand for the facility or a view that the technical challenges required to abide by the detailed provisions of the Measure has led to the reform not being implemented at Local Authority level.

The Welsh Government believes that enabling remote attendance has particular attractions for Members who may, for employment, travel or domestic reasons, find it difficult to attend some meetings. This may increase with the move to new Authorities, given their larger size and if the age, gender and employment profile of Councillors improves, as intended.

Although no provision is made in the Draft Bill, it is our intention to include in the Bill for introduction provisions which would amend the 2011 Measure in order to further facilitate the operation of remote attendance by Councillors at Council meetings.

Question 4.5: Do you agree the provisions relating to remote attendance in the 2011 Measure should be made more flexible?

The Cost of Senior Management in Local Government

The White Paper proposed a permanent control mechanism for salary awards for all Chief Officers through the IRPW, a 'central' appointment process for Chief Executives and Chief Officers, and term limits for Chief Executives. Following further consideration, these issues will be addressed as part of wider considerations in respect of senior managers across the public service to create a more transparent and open framework for senior appointments and pay in Local Government, and through the work of the Public Services Staff Commission.

The Role and Responsibilities of Chief Executives

We will seek a further appropriate legislative opportunity to provide that the Returning Officer role in each Principal Authority should form an intrinsic duty of the Chief Executive, for which no additional personal fee would be payable.

On the basis of existing electoral cycles, the new County Councils would assume the full functions of Local Government on 1 April 2020, just over a month before the scheduled General Election of May of that year. That being the case, it would not be practical for a new Returning Officer for that new Authority to be appointed at that time. Whilst there is no provision in the Draft Bill, we propose that the Shadow Authorities be given powers in the Bill for introduction to appoint Returning Officers to serve until such time as it was convenient.

Question 4.6: Do you have any comments on our proposal that Shadow Authorities should be required to appoint interim Returning Officers?

Chief Executives to Promote Engagement and Diversity in Democracy In the White Paper we consulted on a proposal that there should be a duty on the Chief Executive to promote engagement and diversity in democracy. On reflection, we consider the duty to promote engagement in democracy should rest on the Authority as a whole, as set out in Part 3 of the Draft Bill, Promoting access to Local Government.

Local Authority Chief Executives will inherit powers and duties currently placed upon the Local Authority's Head of Paid Services. This will include a requirement that they keep under review the proper management of the Council's staff and, if appropriate, make proposals to the Council. This means compliance with existing legal provision and legal duties (for example, in relation to bullying, harassment and discrimination) should be taken account of in proposing policies and procedures for the Authority to consider adopting. In addition, in Section 104 of the Draft Bill, there is a requirement for the Leader to set objectives for the Chief Executive, which could include objectives relating to equality and diversity. The Welsh Ministers could recommend in statutory guidance that Leaders should be mindful of the importance of setting and monitoring such objectives to ensure compliance with equalities legislation. There is, therefore, no specific additional provision for this in the Draft Bill, as there is extensive legislative provision already in place on these matters.

Power to Dismiss Senior Officers by a Vote of the Full Council

Certain senior officers (the Chief Executive, Chief Finance Officer, Monitoring Officer and the Head of Democratic Services) are subject to arrangements governing their employment which prevent an Authority dismissing them unless there has first been an investigation by an independent person which would justify this course of action. In England, the UK Government has replaced this with a vote in full Council being required to bring about a dismissal. There is a possibility this could give rise to claims against Local Authorities for unfair dismissal and careful consideration would need to be given if such an approach was adopted in Wales. Nevertheless, the Welsh Government would welcome views as to whether it should adopt a similar approach.

Question 4.7: Do you have any comments on the desirability of giving Councils the power to dismiss the Chief Executive, the Chief Finance Officer, the Monitoring Officer and the Head of Democratic Services through a vote?

Council Functions and Responsibilities

The arrangements for determining who within a Local Authority has responsibility for making decisions in relation to the various functions of Local

Government is provided for in regulations made under the Local Government Act 2000.¹²

In short, most functions are the responsibility of the Local Authorities' Executive, but the regulations describe those functions which cannot be the responsibility of the Executive, cannot solely be the responsibility of the Executive, or can either be the responsibility of the Executive or the full Council, dependent on local choice.

These regulations are detailed and prescriptive. They are also liable to become out of date and require frequent revision. We are proposing to simplify the system and give greater flexibility to new Authorities following mergers.

We propose the provisions of section 13 of the 2000 Act should be replaced by a more liberal provision in relation to the allocation of responsibility within the Local Authority. Authorities would be obliged to have regard to guidance from the Welsh Ministers when deciding responsibility for functions. They should, however, need to abide by certain principles which we propose to include in the Bill for introduction. We also propose there should be a power of direction vested in the Welsh Ministers requiring a Local Authority to allocate responsibility for a function in a particular way.

We propose that the principles which would guide the allocation of functions would be:

- when the Council is operating in a quasi-judicial role the functions involved must not be for the Executive. This would include decisions in relation to planning and licensing;
- approval of the Council's budget and financial planning, including the amount of Council Tax required, should fall to full Council;
- appointments of senior staff should be reserved to full Council;
- the appointment of the Electoral Registration Officer and electoral matters more generally should fall to full Council;
- remuneration of Members of the Authority should be reserved for full Council;
- functions related to the provision of services by the Council should be the responsibility of the Executive;
- the allocation of functions requires the agreement of both full Council and the Executive; and
- the Council's scheme of delegation should be published and be accessible through the Council's website.

Local Authorities and their Executives are able to delegate their functions, subject to some exceptions, if they wish and, under these proposals, this would include an ability to delegate certain functions to community area committees, as well as a new regime for the delegation of functions to third parties (see above).

_

¹² http://www.legislation.gov.uk/wsi/2007/399/contents/made

Question 4.8: Do you have any comments on our proposal to change the framework within which Councils and their Executive determine how their functions are to be allocated?

Transfer of Local Authority Assets

The White Paper included proposals to enable eligible community bodies to initiate the transfer of assets from a Local Authority. A majority of consultation responses were in favour (59% for, 22% against). The objective of the proposal was to facilitate asset transfer to community bodies.

From the perspective of community bodies, the issue which was raised most frequently at the consultation workshops and in the written responses as a barrier to asset transfer was the quality of engagement between Local Authorities and community bodies. The period of notice given to community bodies of a disposal and the quality of information provided (including about liabilities) were frequently mentioned. Community bodies were also looking to Local Authorities to adopt a partnership, rather than a transactional, approach and to recognise the need for ongoing support in the early stages of transfer. Local Authorities identified the lack of capacity among community bodies, such as legal and health and safety expertise, as a barrier to doing business.

The Welsh Government has considered the issues raised and proposes to modify its approach. The proposals as set out in the White Paper could lead to complex bureaucratic processes and create friction between Local Authorities and community bodies, rather than promoting better relations. The Draft Bill does not include any provision in relation to asset transfer, but the Welsh Government intends to include provision in the Bill for introduction.

We propose that community bodies are given statutory notice when Local Authority assets above a certain value are to be disposed of, giving them the opportunity to come forward with new ideas. This would not prevent Local Authorities from taking a planned approach to the disposal of assets whilst giving community bodies greater notice of local opportunities and ensuring more consistent practice across Wales.

For example, Local Authorities could be required to identify disposal of assets above a certain value as a 'key decision'. The Welsh Ministers have regulation making powers under section 22 of the Local Government Act 2000 to require Local Authority Executives to give members of the public or Councillors information about certain types of decisions. Section 107 of the Draft Bill amends section 22 of the 2000 Act to require the information is also provided to Scrutiny Committees and their sub-committees. The regulations can set out types of decision (which could include disposal of assets), the information that must be made available, and the length of advance notice required before a decision is taken.

These powers could be strengthened by including provision in the Bill for introduction addressing related matters, such as how a Local Authority should respond if a community body showed an interest in a particular asset, what assets (if any) might be exempt, how urgent situations should be dealt with, and so on. The financial implications and the duties on Authorities to deliver

value for money would need to be taken into account as part of these considerations. We would welcome your views.

Question 4.9: Do you have any comments on our proposals in relation to the disposal and transfer of Local Authority assets?

Assets of Community Value

The consultation responses showed strong support for a register of (private) assets of community value (86% for, 7% against). The matter has been subject to further consultation by the Minister for Communities and Tackling Poverty in *Protecting Community Assets*. There is no provision relating to assets of community value in the Draft Bill, but the Minister for Communities and Tackling Poverty will consider what action needs to be taken in due course, including the need for legislation.

Part 5: County Councils: Improvement of Governance

Part 5 of the Draft Bill sets out arrangements for a new improvement regime. Chapter 1 puts a general duty on Local Authorities to make and comply with governance arrangements. Chapter 2 requires Local Authorities to have a corporate plan, to consult on it, to keep it under review and report on progress made. Chapter 3 sets out how Local Authorities are to assess the quality of their governance through self assessment, peer assessment and combined assessment. Chapter 4 gives the Welsh Ministers a power to arrange a review of a Local Authority's governance arrangements and gives them a power to intervene when a Local Authority's governance arrangements are inadequate. Chapter 5 provides for better co-ordination between the regulators and Chapter 6 makes miscellaneous provision relating to the previous chapters. Chapter 7 sets out new functions and revised membership of Corporate Governance and Audit Committees (previously called Audit Committees).

It is our intention to commence the majority of these provisions when the Bill is enacted, in order that the transition to the new regime can start immediately and support the process of mergers.

Question 5.1: Do you have any comments on any of the provisions in Part 5 of the Draft Bill?

We also welcome your views on the following matters.

Chapter 1: Duty to make arrangements to secure good governance etc.

At present, County Councils are subject to a 'general duty in relation to improvement' and the associated improvement assessment and intervention regime provided for in Part 1 of the 2009 Local Government (Wales) Measure ("the 2009 regime"). The 2009 regime will be repealed in relation to County Councils by section 147 of the Draft Bill. The provisions in Part 5 replace the 2009 regime with a regime focused on the quality of governance in County Councils, with much greater local determination and accountability.

This chapter requires a County Council to make, implement and comply with arrangements in order to ensure good governance, accountability, and economy, efficiency and effectiveness in the use of the Council's resources. The Council's compliance with its own governance arrangements is assessed by way of self, peer and combined assessments, as set out below.

A County Council's governance arrangements must comply with the principles, processes and practices set out in regulations by the Welsh Ministers. The Welsh Ministers may prescribe in regulations a code of practice or guidance. An example of such a code of practice is the *Delivering Good Governance in Local Government: Framework* published jointly by the

Chartered Institute of Public Finance and Accountancy and the Society of Local Authority Chief Executives.

Question 5.2: Do you have any comments on our proposal to subject Local Authorities to a governance arrangements duty?

Chapter 2: Corporate Plans

Consultation responses were largely supportive of the proposals that Local Authorities should be required to produce a corporate plan. However, respondents generally did not agree that the corporate plan should be the preserve of the Chief Executive, but rather it should be approved by the Council. The Welsh Government accepts the arguments and the Draft Bill reflects changes we have made in response to these concerns.

Chapter 3: Assessments of Governance Arrangements

The Draft Bill imposes a new duty on Local Authorities to make and implement governance arrangements (see above). Compliance with this duty will be the subject of the proposed self assessments, peer assessments and combined assessments.

Self Assessment and Peer Assessment

The responses to the White Paper consultation with regards to self assessment and peer assessment were supportive. Local Authorities and the WLGA emphasised the benefits of peer assessment being sector-led. Concerns were raised about the burden of the proposed biennial peer assessment. The Draft Bill reflects changes we have made in response to these concerns, so that each Local Authority will only be required to undertake a peer assessment at least once per election cycle.

Annex A sets out how we would expect a peer assessment to be developed. This will form the basis of Welsh Government guidance. We would welcome your views on whether this model approach is reasonable and whether it could be strengthened in any way.

Question 5.3: Do you have any comments on the model approach to peer assessment set out in Annex A?

Combined Assessment

Online responses to the White Paper proposals regarding an external assessment of individual Local Authorities' governance arrangements were broadly supportive. There was limited discussion regarding these proposals in free-form responses. The WLGA and those Local Authorities that did respond had concerns about the burden of a biennial 'health check'. Following further discussions with the relevant regulators (the Auditor General for Wales, CSSIW and Estyn), the Welsh Government accepts this argument and the Draft Bill provides for a less prescriptive regime of combined assessments.

Chapter 6: Miscellaneous Provision

Amendment of Local Government (Wales) Measure 2009

The Draft Bill removes County and County Borough Councils from the definition of a 'Welsh improvement authority' in the 2009 Measure. This means the improvement regime in Part 1 of the Measure will no longer apply to Principal Councils. It is our intention that this provision should come into force at the same time as the duty in Part 5, Chapter 1 of the Draft Bill (*Duty to make arrangements to secure good governance etc.*) comes into force. For the purposes of consultation on the Draft Bill, Part 1 of the 2009 Measure will continue to apply to National Park Authorities and Fire and Rescue Authorities in Wales. The Welsh Government will consult fully on future arrangements for National Park Authorities and Fire and Rescue Authorities in Wales regarding performance management and governance mechanisms in due course.

Chapter 7: Role of the Corporate Governance and Audit Committee

It is proposed that the onus should be on the Corporate Governance and Audit Committee to hold the Local Authority to account for taking action in response to a self assessment, peer assessment, combined assessment and independent governance review. The Draft Bill requires the Corporate Governance and Audit Committee to review the Authority's response to reports and recommendations made under these assessments and, if appropriate, make recommendations to the Authority.

Question 5.4: Do you have any comments on the proposed role for the Corporate Governance and Audit Committee in relation to the Local Authority's response to the self assessment, peer assessment, combined assessment and governance review?

Other Matters

Strengthening Internal and External Scrutiny

Consultation responses to the proposal in the White Paper to strengthen the links between internal scrutiny and external scrutiny ("the relevant regulators") were broadly positive. Provisions for this are not included in the Draft Bill, but will be included in the Bill for introduction.

It is our intention to require the regulators to submit their reports to the relevant Overview and Scrutiny Committee at the same time final reports are issued to the Local Authority. The Local Authority would be required to involve the relevant Overview and Scrutiny Committee in the development of their response to regulators' reports. For example:

- a draft report/action plan in response to an audit or inspection would be prepared by the Local Authority;
- the relevant Overview and Scrutiny Committee would consider the draft report and provide proposals for change;

 the Local Authority, in finalising their response to a report, must have regard to any such proposals from the Committee.

The regulators would be required to present their findings to the relevant overview and scrutiny committee, when requested to do so, to aid them in their consideration of the Local Authority's response to reports.

Report on the 'State of Local Government'

Whilst the proposal for the relevant regulators to make a 'State of Local Government' report was generally supported by online respondents, there was limited feedback on the specifics of the proposal. The Local Authorities, in their free form responses, did not identify any specific issues, concerns or benefits for this proposal other than a general point about adding unnecessary burdens to Local Authorities.

Since consulting, the Well-being of Future Generations (Wales) Act 2015 has received Royal Assent and is coming into force. This Act will give rise to a number of reports. We will consider the proposal for a 'State of Local Government' report by the relevant regulators further, as implementation of the 2015 Act progresses.

Single Information Portal

Respondents to the consultation were supportive of more streamlined, transparent and accessible performance data. Online respondents supported a consistent approach to allow comparisons of performance between Local Authorities.

Whilst provisions relating to a single information portal do not feature in the Draft Bill, it is our intention that the Bill for introduction will include provisions requiring Local Authorities to publish key data and documents through an online portal. The provisions will enable the Welsh Government to specify the information that must be published.

Complaints

The White Paper set out our proposals that Local Authorities should establish a statutory complaints handling process, primarily online, and be required to record information, analyse and report on complaints received and how they have been handled. This would be based on the Model Concerns and Complaints Policy and sit alongside the existing requirement for Local Authorities to have a process in place to enable them to deal with complaints and representations about their social services functions. We recognise the National Assembly for Wales' Finance Committee has recently published the Draft Public Services Ombudsman (Wales) Bill for consultation and will reflect further on our proposals as this legislation develops.

Local Public Accounts Committees

The White Paper sought views on whether there was merit in establishing a system of local Public Accounts Committees (PACs) in Wales. Few responded directly on this point and many considered there was not enough information in the White Paper to form a view.

There have been a number of attempts in the past by Local Authorities, Local Health Boards and others to understand the combined impact of public spending on outcomes in a local area. The aim has generally been to identify and avoid duplication of spending, prevent cost-shunting and find smarter ways of doing things.

There are inherent difficulties in designing a local system where the PAC would need to report to a number of separate organisations rather than to a single body. The Centre for Public Scrutiny has put forward a system, suited to the kind of devolution arrangements emerging in England, such as combined authorities, which would establish local PACs as separate bodies with their own resources. This model would be a significant addition to the cost and complexity of public services in Wales, and is unlikely to add sufficient value to be justifiable.

In addition, PACs look retrospectively at what has been done and how money has been spent, rather than critically examining future policy choices which might lead to more cost effective outcomes. An alternative approach might build on the joint local public service leadership model which Public Services Boards (PSBs) establish. Within the framework of the Well-being of Future Generations (Wales) Act 2015, PSBs will have a strong interest in examining how the policy choices and resource decisions facing public services could be used to improve outcomes in an area more cost effectively. PSB members could jointly agree a programme of work and contribute to its resourcing, providing them with powerful evidence to take back into their organisations to inform corporate decisions.

In terms of the scrutiny of the work of PSBs, we believe the existing and proposed legislative provision is sufficient. There is a requirement in the Wellbeing of Future Generations (Wales) Act 2015 that the decisions, actions and governance arrangements of PSBs are scrutinised by a Local Authority Overview and Scrutiny Committee. Provisions in Part 4, Chapter 7 of this Draft Bill will enable the Local Authority to grant voting rights to co-opted members of overview and scrutiny committees. This would apply equally to committees scrutinising the work of PSBs. In addition, section 108 of the Draft Bill enables the Welsh Ministers to prescribe in regulations the circumstances in which two or more Local Authorities must establish a joint scrutiny committee.

We welcome your views on whether we are right to reject the idea of local PACs in Wales, whether PSBs are the right bodies to examine the prospective policy choices facing local public services, and if so, whether they would benefit from additional legal powers, for example, to call officials to give evidence and provide information, and to commission external expertise to inform their investigations.

Question 5.5: Do you have any comments on our proposal to reject local public accounts committees?

Question 5.6: Are Public Services Boards the right bodies to examine the policy choices facing local public services?

Question 5.7: If so, would they benefit from additional legal powers?

The Role of Co-operatives and Mutuals in Public Services

We advocate co-operative and mutual models of delivery and other alternative delivery models only as an alternative to ceasing or privatising services, as a 'least worst' option.

There are three important pre-conditions for ongoing work to develop a stronger framework for change and better support which are:

- accountability to Local Government;
- protection of employee terms and conditions;
- continuation of trades union recognition.

We are consulting separately on the development of a national framework to support decisions being made locally on the appropriateness of alternative delivery models for public services. This consultation will explore the practical support which should be made available to public service organisations, their workforce, citizens and communities in making decisions about how services should be designed and delivered.

The purpose of developing the framework and making support available is to ensure that:

- the wider environment within which proposals for new models are considered, and then established, is conducive for ongoing sustainability and success;
- new models are properly tested before being adopted; and
- the interests of citizens, communities and the workforce are properly considered and protected throughout.

The White Paper recognised that mutualism, co-operation and shared ownership with communities should be at the heart of the transformation of public services. This reflected the opportunities identified by the Welsh Co-operative and Mutuals Commission¹⁴ and in the subsequent report for the Minister for Public Services and the Minister for Economy Science and Transport, *Is the Feeling Mutual? New Ways of Designing and Delivering Public Services in Wales.*¹⁵

In response to these challenges, the Welsh Government proposes granting Local Authorities a general power of competence in the Draft Bill. Local Authorities will be able to utilise this power as a basis for developing alternative delivery models. The Welsh Government has also reviewed the powers of Local Authorities to delegate functions under the Deregulation and

⁵ http://gov.wales/topics/improvingservices/publications/is-the-feeling-mutual-report/?lang=en

1

http://gov.wales/consultations/improving/alternative-delivery-models-public-service-delivery/?status=open&lang=en

¹⁴ <u>http://gov.wales/topics/businessandeconomy/business/welsh-coop-mutuals-commission/?lang=en</u>

Contracting Out Act 1994, and is proposing that they should be repealed and replaced with a new regime that will allow for the delegation of Local Authority functions to third parties by regulations made by the Welsh Ministers. The Draft Bill also includes provisions to empower community bodies to take the initiative in relation to improvement requests and it is our intention to use existing legislation and, if necessary, include additional provision in the Bill for introduction relating to the disposal of assets by Local Authorities (see Transfer of Local Authority assets and Question 4.9 above).

Shared Services

The Report of the Commission on Public Service Governance and Delivery included consideration of the potential improvements in cost effectiveness through the adoption of shared services. It commended the NHS Wales Shared Services Partnership (NWSSP) and suggested it as a model for public sector-wide shared services.

For NWSSP to have the opportunity to take this wider role, the legislative framework would need to change, as hosting within an NHS Trust is constrained by the current legislation 'to provide goods and services for the purposes of the health service'.

Local Authority mergers will create larger service units within the new County Councils but there is also evidence, from the KPMG report into administrative and support functions, that cost savings and the opportunity for better service could be obtained through shared services at a larger scale. There is evidence elsewhere that the ability of Local Government to participate in joint ventures and similar arrangements with the public sector and other partners may open up opportunities for innovation and best practice. We wish to consider whether current legislation and regulations for Local Authorities provides the appropriate framework to support these opportunities. We would welcome your views.

Question 5.8: What legislative measures could be considered to enable Local Government to take a public sector-wide shared services role?

Part 6: Community Councils

The provisions in Part 6 of the Draft Bill relate to Community and Town Councils. Chapter 1 requires the Local Democracy and Boundary Commission for Wales to undertake a review of Community Council arrangements. Chapter 2 requires Community Councillors to complete training on matters specified by the Principal Council. Chapter 3 extends the terms of Community Councillors elected in 2017 to six years and provides that Community Council terms will be fixed at five years from 2023.

Question 6.1: Do you have any comments on any of the provisions in Part 6 of the Draft Bill?

We also welcome your views on the following matters.

Note: the term Local Authority generally means both Principal Councils (County and County Borough Councils) and Community Councils. In order to distinguish between them, in this section we use the terms Principal Council / County Council and Community Council, rather than Local Authority.

Chapter 1: Review of Community Council Arrangements

The White Paper set out our intention to strengthen the Community Council sector so that it has greater capacity and capability to take on responsibility for local facilities and services. The proposal was that Principal Councils would undertake a review of Community Council arrangements in their area, with the intention of grouping smaller communities under a Common Council.

Consultation responses from Principal Councils and the Welsh Local Government Association (WLGA) argued that while this approach was reasonable in principle, it would be difficult for Principal Councils to complete such a review by 2023, given the wider context of mergers and reform. The Welsh Government agrees and has previously announced that the review will now be conducted by the Local Democracy and Boundary Commission for Wales (LDBCW).

The provisions in the Draft Bill require the LDBCW to submit their draft reports to the new County Councils on or after 1 April 2020, which is the day they assume the full responsibilities and functions of Local Government. There is a case to bring this date forward to May 2019 when the Shadow Authorities come into existence, in order that the Commission may undertake the review, publish draft reports and consult in a more timely fashion. We would welcome your views.

The provisions require the LDBCW's recommendations to be implemented by the County Councils by order. Alternatively, this is a task the Commission could undertake. This would enable it to undertake and implement the reviews in a single smooth process. We would welcome your views.

Question 6.2: Should the Boundary Commission be required to submit their draft reports to Shadow Authorities from May 2019?

Question 6.3: Should the new County Councils implement the Boundary Commission's recommendations or should this be a responsibility of the Boundary Commission itself?

Chapter 2: Members of Community Councils to Complete Training

Community Councils and Principal Councils have formal relations on a number of matters. For example, Standards Committees of Principal Councils have oversight of code of conduct matters relating to Community Councillors, Community Councils are consultees on planning matters in their area, and Principal Councils may delegate functions to Community Councils. It is in the interest of both parties that Community Councillors are equipped with the knowledge and skills necessary to undertake their role effectively. Therefore, the Draft Bill includes provisions which require Principal Councils to consider if there are matters on which it should be compulsory for Community Councillors to complete appropriate training.

If the clerk to the Community Council receives a complaint that a Community Councillor has not completed compulsory training, the Draft Bill puts a duty on the clerk to look into the matter. If, having consulted the chair of the Community Council and considered any representations the Community Councillor makes about his or her failure to complete the training, the clerk's view is the Community Councillor does not have a good reason for not completing the training, the clerk must notify the Community Council at a public meeting. As the clerk is an employee of the Council, this has the potential to strain the good relations between the clerk and the Council. We welcome your views on whether this is the right approach to ensuring Community Councillors complete compulsory training or whether an alternative approach should be considered.

Question 6.4: Do you have any comments on our proposals relating to compulsory training for Community Councillors?

Chapter 3: Community Council Election Dates

The Draft Bill provides that elections to all Community Councils, including those in Powys, will be held in May 2017 and May 2023, in both cases coinciding with elections to Principal Councils. The terms of Community Councillors elected in 2017 would need to be extended so they serve a sixyear term. From 2023 Community Council elections will be held every five years, in line with the new cycle of elections to Principal Councils.

Question 6.5: Do you have any comments on our proposal to extend the term of Community Councillors elected in 2017 to six years?

Other matters

Capping the Community Council Precept

Further consideration has been given to the proposal in the White Paper to cap the amount by which non-competent Community Councils can raise their Council Tax precept. This issue will now be considered on a longer term basis following the review of Community Councils and alongside the wider financial reforms.

Annual Reports

The White Paper set out our intention to consider whether other aspects of the governance of Community Councils should be strengthened. It is our intention to require every Community Council to publish an annual report so that local people can understand what the Council has achieved during the previous year. Many already do so. Although we have not included provision in the Draft Bill, it is our intention to include provision to this effect in the Bill for introduction.

Training

Given the extra responsibilities larger Community Councils may take on, including the significant legal and financial responsibilities which may accrue from exercising the general power of competence, we believe it is right that Community Councils should consider their own training needs (both Councillors and employees) and, where appropriate, make that training compulsory. We have not previously consulted on this proposal and no provision is included in the Draft Bill, however, it is our intention to include provision to this effect in the Bill for introduction.

Question 6.6: Do you have any comments on our proposal that Community Councils should be required to consider and plan for the training needs of their own members and employees?

<u>Performance Management</u>

In the White Paper we consulted on a proposal that the chair of a Community Council should set objectives for, or otherwise manage the performance of, the clerk. The clerk is an employee of the Council and as a matter of good employment practice, they should have both a clear job description and be set annual objectives so they know what they are required to do. For very small Community Councils, this approach may be seen as too much of a burden. However, as Community Councils with bigger budgets take on more responsibilities, which is our intention through the review of Community Council arrangements, good employment practice will become more important. We would welcome further views on this proposal and, in particular, whether the duty should rest on the Council as a whole or its chair, and whether such a duty should apply to all Community Councils or only some.

Question 6.7: Do you have any comments in relation to the setting of objectives for a Community Council clerk?

Community Polls

In the White Paper we said 'we will also clarify the legislative framework governing community polls'. There were insufficient responses to draw any useful conclusions. We set out here our further thinking on community polls.

Community polls are governed by section 99 and Schedule 12 of the Local Government Act 1972, as amended by the Local Government (Wales) Measure 2011. A community poll may be called in respect of any matter raised at a community meeting, subject to the support of a sufficient number of electors. Community polls provide an indication of community feeling about the subject of the poll. However, they are non-binding and a Principal Council or Community Council does not have to act on the results of the poll, although they do have to bear the cost.

Community polls often relate to matters outside of the control of Local Government. For example, in recent years, community polls have been held about the future of Ffestiniog Memorial Hospital and Flint Community Hospital, but these are matters in which Local Government has no say. Community polls are costly and turn-out is generally low. Four recent polls in Cardiff cost in excess of £20,000 and in recent polls, turn out of those entitled to vote has ranged from 37% down to 9% in a 2012 poll relating to the closure of the paddling pool in Ynysangharad Park in Pontypridd.

The Welsh Government considers a modernising reform which strengthens community voice and progresses the digital agenda of Local Government should be considered, through the use of electronic petitions. Petitions are an established feature of Government at all levels in the UK and in recent years, there has been a significant increase in the use of e-petitions as a means of giving the public a greater say in decisions which affect them, by the National Assembly, the UK Government, the Scottish Parliament, and Local Government in England and Scotland. Most Councils in Wales have procedures for dealing with written petitions, but none is known to operate e-petitions.

We are, therefore, proposing that in the Bill for introduction the existing legislation which provides for community polls should be repealed in respect of Wales, and be replaced by an alternative approach which would require Principal Councils to establish a petitions scheme and an e-petitions facility. This would enable communities (of place or interest) to express their views on matters which concern them, without the restrictions and costs which currently apply to community polls, and there would be a requirement for Councils to set out how they would respond. We would welcome your views.

It is, therefore, our intention to put on hold the implementation of any changes to the existing Community Polls rules, following the consultation on a proposed amendment of the rules¹⁶ which concluded in February 2014. With regard to polls conducted for the purpose of forming or dissolving a

¹⁶ Welsh Government | Community Polls

Community Council, it is our intention to review the existing legislation, as set out in the following section.

Question 6.8: Do you have any comments on our proposal to repeal the legislation relating to community polls and to require instead that Local Authorities should implement a system of e-petitions?

Review of Community Council Legislation

The legislation governing the creation, grouping, merging and dissolution of Community Councils and the holding of community meetings etc. is found in the 1972 Local Government Act, as amended.

It is our intention to suspend the majority of the 1972 provisions about community applications for the duration of the LDBCW's review of Community Council arrangements and for a period afterwards. This is to ensure the Commission is able to undertake its review without Council arrangements changing during the course of the review. There is a question about whether to revert to the provisions in the 1972 Act from 2025 onwards. We intend to use the intervening period to consult further with the sector and consider a legislative framework for Community Councils which is appropriate for the mid 21st century.

Part 7: Workforce Matters

Part 7 of the Draft Bill deals with workforce matters. Chapter 1 enables the Welsh Ministers to publish guidance to public bodies on workforce matters. Chapter 2 provides for the establishment of a Public Services Staff Commission. We recognise our vision for reform, and our ambition for world class public services, can only be realised through a world class public service workforce with the right skills and support to deliver them. The dedication and excellence of the public service workforce is instrumental to transformation. Our non-statutory Public Services Staff Commission has been operating since September using the Workforce Partnership Council as its primary reference point. We are committed to continuing our strong model of social partnership through the Workforce Partnership Council as we take forward our programme of reform.

Question 7.1: Do you have any comments on any of the provisions in Part 7 of the Draft Bill?

We also welcome your views on the following matters.

Chapter 2: Public Services Staff Commission

Following consultation in late 2014 on the White Paper, *Devolution*, *Democracy and Delivery: Public Services Staff Commission*, the Minister for Public Services established a non-statutory Public Services Staff Commission in September 2015. The overwhelming majority of consultation responses were in favour of putting the Commission on a statutory footing and the Draft Bill includes such provision.

However, it may be the case that a statutory Commission would be more constrained in the matters on which it can offer workforce guidance in comparison with the non-statutory Commission. Whether this is the case or not will depend on the content of the UK government's proposed Wales Bill which will make changes to the devolution settlement in Wales.

The content of the Wales Bill and its implications in relation to the proposed statutory Public Services Staff Commission will not be fully known until the Bill is passed. This will not happen during the course of this consultation and the passage of the Wales Bill is not in the control of the Welsh Ministers. Nevertheless, the Welsh Government would welcome views at this stage on the best way forward if it turned out that a statutory Commission would be more constrained in the matters on which it can offer workforce guidance in comparison with the non-statutory Commission.

Question 7.2: Do you have any views on whether it would still be desirable to establish a statutory Public Services Staff Commission if it would be more constrained in the matters on which it could issue guidance than a non-statutory Commission?

Part 8: General and Schedules

Part 8 of the Draft Bill sets out the meaning of terms and of references to existing legislation in the Draft Bill, sets out how the provisions will apply to Local Authorities before 1 April 2020, gives the Welsh Ministers powers to make consequential and transitional provision, makes provision for orders and regulations, sets out when the different parts of the Draft Bill will come into force, and the short name by which the Draft Bill will be known when it is enacted.

Schedules

In addition, there are a number of Schedules to the Draft Bill.

Schedule 1 sets out the areas of the new Counties in Wales. Two alternative tables are provided (see Part 2, Chapter 1 for an extensive discussion).

Schedule 2 sets out the arrangements for holding the first ordinary elections and first meetings of the new Councils.

Schedule 3 sets out provisions relating to the finance of the new Councils.

Schedule 4 makes various transitional arrangements and provides for staff, property, rights and liabilities to be transferred from the existing County and County Borough Councils to the new County Councils.

Schedule 5 makes amendments to the Local Government Act 1972.

Schedule 6 makes amendments to existing legislation in relation to assistants to Local Authority executives which are provided for in Chapter 5 of Part 4 of the Draft Bill.

Schedule 7 makes consequential amendments to existing legislation in relation to Chief Executives of the new Councils.

Schedule 8 makes various amendments and repeals to existing legislation in relation to community area committees established by provisions in Chapter 3 of Part 3 of the Draft Bill.

Schedule 9 makes various amendments in respect of the renaming of County Council Audit Committees.

Further explanation of the content of the Schedules is provided in the Explanatory Memorandum and the Explanatory Notes.

Question 8.1: Do you have any comments on any of the provisions in Part 8 of the Draft Bill or on any of the Schedules?

Annex A: Peer assessment

Governance Themes

The focus of peer assessment under the Draft Bill is a Local Authority's compliance with its good governance duty. In essence, the assessment will look at the corporate capability and capacity of a Local Authority, within a framework of 'good governance themes', such as that set out in *Delivering Good Governance in Local Government: Framework* (CIPFA/SOLACE).¹⁷

Reviewers

To add maximum value, members of the peer assessment panel should not just be external to the Local Authority, but experienced and credible. This way the Local Authority knows that their findings can be trusted and will give them confidence that the changes proposed are necessary.

Peer assessment panels should be made up of a cross section of individuals (Elected Members, officers and those from other sectors) whose skills and expertise cover the breadth of the core themes and any additional areas that the Local Authority wishes to cover. Whilst it is important that a proportion of the assessment team have a good understanding of the complexity of working in a political environment and the impact which local politics can have on a Local Authority's organisational culture and priorities, those with experience of other sectors can also offer a valuable insight. We would expect a peer assessment panel to be made up of both those with direct experience of working in and with Local Government.

As a guide/ minimum, we would expect peer assessment panels to be made up of:

- A chair who will lead the peer assessment
- At least 1 Elected Member from a Welsh Local Authority that does not have a border with the host Authority
- At least 1 senior officer from a Welsh Local Authority that does not have a border with the host Authority
- At least 1 Elected Member or officer from another Local Government administrative area (e.g. England, Scotland, Northern Ireland, Ireland, or further afield)
- At least 1 individual with relevant expertise or knowledge from beyond Local Government (e.g. the Private or Third sector)

Local Authorities could also consider whether it would be appropriate to invite others to be members of the assessment panel, for example:

Tudalen 122

¹⁷ http://www.cipfa.org/Policy-and-Guidance/Publications/D/Delivering-Good-Governance-in-Local-Government-Framework

- A representative from the WLGA (or LGA)
- A representative from an Audit, Inspection and Regulation body

Example Process

Typically a peer assessment process will take a number of months, from start to finish and it is suggested that there may well be an ongoing relationship with the chair or the panel in a mentoring capacity in the longer term. The information below is a suggested example, recognising that there will need to be flexibility around the process, but the principle of peer assessment is that it is a 'short and sharp' process ensuring that any findings remain relevant, and allow the Local Authority to respond in a timely manner.

Pre-assessment

The pre-assessment process would typically take around 8 weeks. During this period the Local Authority, working with partners or an organisation paid to coordinate this activity, will scope the requirements of their peer assessment, identifying any particular challenges that have been highlighted in their self-assessment, or in recent reports of the relevant regulators, including a combined assessment. This will also help to identify the skills and expertise required to maximise the impact of the review.

Once scoped, the Local Authority, working with partners or an organisation paid to coordinate this activity, will be able to identify an appropriate assessment panel to undertake the assessment.

Once appointed, the chair (or lead assessor) will have the opportunity to meet with the Leader and Chief Executive of the host Local Authority to discuss the scope of the assessment in detail and amend the scope of the assessment as necessary. This is also an opportunity to discuss expectations and the approach to the assessment.

During this time the Local Authority will be able to gather information to share with peer assessment panel that is relevant to set the context for and inform the assessment. This information should rarely be required to be produced specifically for the assessment as, typically, the documents should be in existence. However, it might be useful for the host Authority to produce an overview document.

<u>Assessment</u>

The assessment itself will typically take place over a period of about 4 weeks. The initial period of the assessment is desk based for the assessment panel. The assessment panel will have time to consider the information provided to them, and have the opportunity to request any additional information that they believe would be pertinent to the assessment. This desk based review will allow them to develop areas for consideration for them to focus on during their site visit.

The site visit would normally last around 4 days over no more than a two week period, and ideally in a single week. The site visit is an opportunity for the assessment panel to question and discuss areas within the scope of the assessment with Elected Members, officers and other stakeholders of the host Authority.

At the end of each day, the panel members would come together to discuss their findings and consider if there are any other issues that should be considered or individuals or stakeholders that it would be useful to meet.

At the end of the site visit the assessment panel will present its findings to the Leaders and Chief Executive. The Local Authority may wish to consider if others should be invited to hear the assessment findings (for example, the whole Executive, senior management team, full Cabinet). This presentation is an opportunity for the peer assessment panel to set out their main findings and areas which they believe the Authority would benefit most from focusing in on. It is also an opportunity for the host Authority to clarify its understanding of the issues raised.

Post assessment

Following the site visit and presentation the assessment panel will produce a more detailed report for the Authority to consider in more detail. This report is not meant to be all encompassing, but it should give sufficient detail that it can be read and understood as a stand alone document, as the report will eventually be published on the Local Authority's website. We would expect that the Local Authority should receive the report within 2 weeks of the site visit. This ensures that the issues identified during the assessment and any recommendations remain relevant. As part of the report we would expect the assessment panel to recommend when the next peer assessment should be undertaken. Peer assessments must be undertaken at least once in every election cycle (i.e. 5 years from 2023) but it may be advisable to conduct them more frequently, for example, during periods of significant organisational change.

The Local Authority will be required to respond to the findings from the assessment. Whilst some Local Authorities may choose to produce an action plan as a result of the assessment, others will feed in the action they intend to take as a result of the assessment findings into other documents, for example, recruitment and retention strategies or the corporate plan. The Local Authority's response should be made within 4 weeks of the report being produced to ensure that early action can be taken.

The Local Authority will need to be clear what action it is taking as a result of the peer assessment and there will be a requirement that the proposed actions are considered by the Corporate Governance and Audit Committee. We would encourage the lead assessor to return to the host Authority within 3-6 months following the review to support the Authority to develop their improvement planning.

The Local Authority may choose to approach members of the peer assessment panel, or the lead assessor to establish a longer term 'mentoring' relationship to provide ongoing peer support to the host Authority.

<u>Timeline</u>

Activity	Scoping of assessment as well as identification of the assessment panel	Review Desk based review and research by assessment panel	Site Visit	Report write up	Published	Response to findings developed by Local Authority
Time scales	8 weeks	3 weeks	1 week	2 weeks	Report	4 weeks

Annex B: Consultation Questions

The Welsh Government would like to hear your views on the Draft Bill, Draft Explanatory Memorandum, Draft Regulatory Impact Assessment and the matters raised in this Consultation Paper. We would like your views on the practical application of the provisions contained within the Draft Bill.

The intention would be to introduce the Bill into the National Assembly for Wales following the Assembly elections in 2016, and we want to ensure we have addressed as many issues as possible before doing so. Your responses will help inform the Bill for introduction.

Please let us have your responses and comments on the questions set out in this Annex, based on the suite of documents that comprise this consultation.

PART 1

Question 1.1: Do you have any comments on any of the provisions in Part 1 of the Draft Bill?

Question 1.2: What are your views on the options for 2 or 3 Counties in North Wales, as set out in Schedule 1 to the Draft Bill?

Question 1.3: What are your views on the proposed configuration of Local Government areas in Wales?

Question 1.4: Do the Welsh Ministers need to seek any further powers to support the integration of Powys Teaching Health Board and Powys County Council?

Question 1.5: What are your views on the procedure for naming the new Counties?

Question 1.6: What are your views on the proposed changes to the Local Government election timetable?

Question 1.7: Do you have any general comments on the provisions in section 16 and Schedule 3 of the Draft Bill relating to Local Government finance?

Question 1.8: How could the Welsh Government measure the current level of avoidance of Non-Domestic Rates?

Question 1.9: Do you have any comments or suggestions on how future legislation could help to reduce instances of avoidance of Non-Domestic Rates?

Question 1.10: In what other ways could the Welsh Government enable Local Government to reduce the level of avoidance and fraud within the Non-Domestic Rates system?

Question 1.11: Do you agree that the preserved counties be abolished and that consequential amendments are made so that the appointments of Lord-Lieutenants and High Sheriffs are made in respect of the Counties in existence after 1 April 2020?

Question 1.12: Are there other matters of a technical nature which should also be considered?

PART 2

Question 2.1: Do you have any comments on any of the provisions in Part 2 of the Draft Bill?

Question 2.2: Do you have any comments on our proposals relating to Community Councils with competence?

PART 3

Question 3.1: Do you have any comments on any of the provisions in Part 3 of the Draft Bill?

Question 3.2: Do you have any comments on the proposed public participation duty and the requirement to consult on the annual budget?

Question 3.3: How should community representatives to sit on community area committees be sought and selected?

Question 3.4: Do you agree County Councils should be able to delegate functions to a community area committee? If yes, are there any functions that should or should not be capable of being delegated?

Question 3.5: Do you have any views on whether transitional arrangements need to be put in place for existing area committees, or is a good lead-in time sufficient?

Question 3.6: Do you have any comments on the revised provisions for 'improvement requests' or on the interaction between these provisions and those relating to the public participation duty (Part 3, Chapter 2) and community area committees (Part 3, Chapter 3)?

Question 3.7: Do you have any comments on any of our further proposals relating to access to meetings?

Question 3.8: Do you have any comments on our proposals to enhance participation by children and young people through the public participation duty?

PART 4

- Question 4.1: Do you have any comments on any of the provisions in Part 4 of the Draft Bill?
- Question 4.2: Do you have any comments on the proposed duty on leaders of political groups or the monitoring and reporting roles of the Standards Committee?
- Question 4.3: Do you have any comments on our proposals in relation to the delegation of functions by Local Authorities?
- Question 4.4: Do you have any comments on our proposal to give the Welsh Ministers a power to direct the IRPW to have regard to guidance when reviewing the remuneration framework for Councillors?
- Question 4.5: Do you agree the provisions relating to remote attendance in the 2011 Measure should be made more flexible?
- Question 4.6: Do you have any comments on our proposal that Shadow Authorities should be required to appoint interim Returning Officers?
- Question 4.7: Do you have any comments on the desirability of giving Councils the power to dismiss the Chief Executive, the Chief Finance Officer, the Monitoring Officer and the Head of Democratic Services through a vote?
- Question 4.8: Do you have any comments on our proposal to change the framework within which Councils and their Executive determine how their functions are to be allocated?
- Question 4.9: Do you have any comments on our proposals in relation to the disposal and transfer of Local Authority assets?

PART 5

- Question 5.1: Do you have any comments on any of the provisions in Part 5 of the Draft Bill?
- Question 5.2: Do you have any comments on our proposal to subject Local Authorities to a governance arrangements duty?
- Question 5.3: Do you have any comments on the model approach to peer assessment set out in Annex A?
- Question 5.4: Do you have any comments on the proposed role for the Corporate Governance and Audit Committee in relation to the Local Authority's response to the self assessment, peer assessment, combined assessment and governance review?

Question 5.5: Do you have any comments on our proposal to reject local public accounts committees?

Question 5.6: Are Public Services Boards the right bodies to examine the policy choices facing local public services?

Question 5.7: If so, would they benefit from additional legal powers?

Question 5.8: What legislative measures could be considered to enable Local Government to take a public sector-wide shared services role?

PART 6

Question 6.1: Do you have any comments on any of the provisions in Part 6 of the Draft Bill?

Question 6.2: Should the Boundary Commission be required to submit their draft reports to Shadow Authorities from May 2019?

Question 6.3: Should the new County Councils implement the Boundary Commission's recommendations or should this be a responsibility of the Boundary Commission itself?

Question 6.4: Do you have any comments on our proposals relating to compulsory training for Community Councillors?

Question 6.5: Do you have any comments on our proposal to extend the term of Community Councillors elected in 2017 to six years?

Question 6.6: Do you have any comments on our proposal that Community Councils should be required to consider and plan for the training needs of their own members and employees?

Question 6.7: Do you have any comments in relation to the setting of objectives for a Community Council clerk?

Question 6.8: Do you have any comments on our proposal to repeal the legislation relating to community polls and to require instead that Local Authorities should implement a system of e-petitions?

PART 7

Question 7.1: Do you have any comments on any of the provisions in Part 7 of the Draft Bill?

Question 7.2: Do you have any views on whether it would still be desirable to establish a statutory Public Services Staff Commission if it would be more constrained in the matters on which it could issue guidance than a non-statutory Commission?

PART 8

Question 8.1: Do you have any comments on any of the provisions in Part 8 of the Draft Bill or on any of the Schedules?

ADDITIONAL QUESTIONS

Question 9.1: Are you aware of any consequential amendments to legislation that will need to be made?

Question 9.2: Please provide feedback you think would be useful in relation to the supporting documents published alongside the Draft Bill i.e. Draft Explanatory Memorandum (including the Regulatory Impact Assessment) and specific Impact Assessments.

Question 9.3: We have asked a number of specific questions. If you have any related issues which we have not specifically addresses, please use this space to comment.

Draft Local Government (Wales) Bill Consultation Response

PART 1

Question 1.1: Do you have any comments on any of the provisions in Part 1 of the Draft Bill?

The Council's comments on specific questions relating to Part 1 are set out in the responses below. The council is concerned that the process of merger will be hugely expensive and place an intolerable burden on existing authorities in terms of the resources and capacity required to support and implement the work of transition committees. The amount of money and time spent by senior officers and members in the work of transition committees will severely restrict the capacity of those authorities to drive service improvements for their residents. There will need to be a substantial injection of resources from government to facilitate this process.

Question 1.2: What are your views on the options for 2 or 3 Counties in North Wales, as set out in Schedule 1 to the Draft Bill?

The Council's position is that it would prefer that mergers do not take place at all. If mergers of local authorities are to proceed then the Council's view is that Denbighshire County Council should merge with Conwy County Borough Council. The Council had prepared a joint submission with Conwy in 2014, which set out the argument for a voluntary merger in detail. As far as we are aware no other pair or group of authorities, consistent with the proposed maps, have produced the detailed argument for merger that our two authorities have. The Council's position on this hasn't changed and we present no further arguments.

Question 1.3: What are your views on the proposed configuration of Local **Government areas in Wales?**

The Council's position regarding its preferred option for North Wales is set out in the response to 1.2 above. The Council does not have a strong view about the rest of Wales, other than an argument about proportionality: the six councils in North Wales is currently between a third and a quarter of all councils. The Council would not support a model that worsened the current ratio.

Question 1.4: Do the Welsh Ministers need to seek any further powers to support the integration of Powys Teaching Health Board and Powys County Council?

The Council has no comment to make on this point.

Question 1.5: What are your views on the procedure for naming the new Counties?

It would appear appropriate that the Shadow Authorities determine the names of the new authorities.

Question 1.6: What are your views on the proposed changes to the Local Government election timetable?

The proposed changes to the election timetable appear to be sensible in order to achieve the transition to the new authorities and a new electoral cycle. The council is however concerned that two three year terms will limit the ability of either Council to be ambitious and to make significant improvements in services to residents

Question 1.7: Do you have any general comments on the provisions in section 16 and Schedule 3 of the Draft Bill relating to Local Government finance?

The Council has no comment to make.

<u>Question 1.8: How could the Welsh Government measure the current level of avoidance of Non-Domestic Rates?</u>

Avoidance is notoriously difficult to quantify. The values attached to such things are by definition hypothetical and will be based on a study in one area and results extrapolated. In other areas such as Housing Benefit fraud when actual instances were compared to studies, the results were much lower. The Council would be wary of introducing an overly bureaucratic and therefore costly administration process to address a problem that may or may not be significant. At the very least it considers that a Welsh (urban and rural) study or research project be commissioned to inform whether this may be an issue.

Question 1.9: Do you have any comments or suggestions on how future legislation could help to reduce instances of avoidance of Non-Domestic Rates?

The Council has no additional comment to make.

Question 1.10: In what other ways could the Welsh Government enable Local Government to reduce the level of avoidance and fraud within the Non-Domestic Rates system?

It could be that something as simple as designing forms to make positive responses mandatory (the example of having 'zero' as a response rather than leaving a box blank on returns etc.) would limit potential avoidance.

Question 1.11: Do you agree that the preserved counties be abolished and that consequential amendments are made so that the appointments of Lord-Lieutenants and High Sheriffs are made in respect of the Counties in existence after 1 April 2020?

This would appear to be a sensible proposal ensuring a consistent geographical area for all civic and administrative purposes.

Question 1.12: Are there other matters of a technical nature which should also be considered?

The Council has no comment to make.

PART 2

Question 2.1: Do you have any comments on any of the provisions in Part 2 of the Draft Bill?

The Council welcomes the introduction of the general power of competence. This power should not, however, be regarded as a panacea. The Draft Bill places constraints on the use of this power in that it will not allow Local Authorities to do anything which they are currently prohibited from doing by existing or future legislation. There are already many such prohibitions and restrictions. Local Authorities will also be constrained by the public law principles which apply to the exercise of any of their functions. The Council does not see the need for any further constraint and encourages Welsh Ministers to use their powers to make regulations in respect of the exercise of the general power in such a way that they reduce rather than increase the barriers to its use.

Question 2.2: Do you have any comments on our proposals relating to Community Councils with competence?

The Council welcomes the removal of the requirement for County Councils to monitor the competence of Community Councils and the £200k turnover test.

The test of competence set out in the Draft Bill would appear relatively meaningless for the following reasons.

The Community Council only has to meet the requirement at a given snapshot in time. It may then continue to exercise the power in respect of new activity, even though it is no longer "competent", for up to five years.

In respect of activity undertaken whilst it was "competent", (which includes any period within the five years that it did not meet the competency requirement) a Community Council may continue to exercise the power for what appears to be an indefinite period.

The Council's comments are similar to those made in response to the White Paper. It would seem sensible to create Community Councils of an appropriate size, impose minimum governance standards and then grant the same powers and responsibilities to them all.

The Council previously commented that the requirement to have a Clerk with a relevant professional qualification could impose a significant financial burden on some

Community Councils given the level of salary that may be needed to attract and retain such staff. In the absence of any detail as to the exact nature of the required qualification it is difficult to comment further. One question that arises is what will be the position of existing Clerks who do not meet this requirement? If this is to be introduced then it would seem sensible that it be implemented after the review of Community Councils has taken place.

PART 3

Question 3.1: Do you have any comments on any of the provisions in Part 3 of the Draft Bill?

The Council is very concerned about the bureaucratic burden that will be imposed on Councils by the establishment of Community Area Committees as proposed in the Draft Bill. In particular the requirement that these be full public committees under the terms of the Local Government Act 1972 with all of the consequential costs of servicing and support that this entails. If these committees are to be truly connected to their local areas it is assumed that their meetings will take place in those areas. If they are also to be subject to the requirement to be broadcast live then there will be additional capital costs of installing the necessary equipment in additional buildings as well as the revenue costs of additional capacity, as suppliers price webcasting services on the amount of hours broadcast. Little, if any, thought appears to have been given to the significant increased costs that will be placed upon councils by this and many other proposals in the draft bill.

The potential for these committees to appoint sub-committees and the requirement for various consultation exercises on behalf of each committee will only increase the burden of support placed on Councils.

The prescriptive nature of proposals for these committees seems to be a departure from the intention stated in the White Paper that Councils would be able to design their own system of community led governance.

There is also concern that unelected representatives will have the ability to vote on what may include functions of the county council that have been delegated to them. This concern is heightened by the reserve power for ministers to prescribe those functions that may be delegated. The Council agrees that a constitution guide be produced as proposed.

The Council already publishes on its website contact details including electronic and postal addresses for its elected members and agrees that an official address be provided as proposed.

Question 3.2: Do you have any comments on the proposed public participation duty and the requirement to consult on the annual budget?

The Council is in favour of increasing awareness and public participation in the work of the Council as proposed. The Council is concerned that the arrangements for the preparation of a strategy and any guidance issued by the Welsh Ministers are not too prescriptive and retain flexibility for Councils to make arrangements appropriate to their individual areas and circumstances.

The Council would make similar comments on the proposal to require consultation in respect of the budget. It will be extremely difficult to have meaningful consultation in the terms described in the draft bill.

The Council does not see why Councils should bear the burden of creating strategies for the increased participation of local people in the processes of other public bodies which have their own resources.

Question 3.3: How should community representatives to sit on community area committees be sought and selected?

It would seem appropriate that if councils have to publish a policy statement setting out how it proposes to exercise its functions in this regard that local discretion is maintained. It would seem sensible that the process of appointment should be similar to the arrangements that are in place for appointing independent members of the Standards Committee, involving some form of notice or advertisement to relevant bodies and the appointment of a panel to implement a selection process.

Question 3.4: Do you agree County Councils should be able to delegate functions to a community area committee? If yes, are there any functions that should or should not be capable of being delegated?

Whether or not a function should be delegated must be the decision of each Council. It is noted with concern that s57 grants Welsh Ministers the power to make regulations to require or restrict the delegation of functions. The Council believes this to be a serious erosion of local democracy and to be unnecessarily micro managing.

The exercise of executive functions by these committees will require scrutiny. There are no proposals for how this is to be done.

Question 3.5: Do you have any views on whether transitional arrangements need to be put in place for existing area committees, or is a good lead-in time sufficient?

The Council's view is that this is a provision which should come into force for the new councils envisaged by Part 1 of the Draft Bill and not for existing Councils. The capacity required following the elections in 2017 to run the existing authorities whilst also supporting transition to the new Shadow Authorities in 2019 will not allow sufficient capacity to set up these area committees and support them properly. The new councils will be able to set up these committees.

Question 3.6: Do you have any comments on the revised provisions for 'improvement requests' or on the interaction between these provisions and those relating to the public participation duty (Part 3, Chapter 2) and community area committees (Part 3, Chapter 3)?

In its response to the White Paper, the Council stated that it would be pleased to receive serious requests from serious bodies in respect of the development and delivery of services. The proposals as drafted appear to include safeguards against repeated vexatious and frivolous requests. The requirement to publish reports of the outcome of discussions, the creation and publication of annual reports and a specific complaints system appear to be overly onerous

As a general point it does appear that this Part of the Draft Bill is a little "crowded" and creates a large resource implication for Councils as set out above.

It may also be confusing to the public to the extent that they are put off from participating because there are a number of competing demands on their attention that could cause consultation fatigue.

Question 3.7: Do you have any comments on any of our further proposals relating to access to meetings?

The Council would expect that it should be able to make provision in its standing orders for the manner in which meetings conducted for the purpose of the Leader answering questions from the public be regulated.

The Council is unaware of any corresponding statutory requirement for Welsh Ministers.

This Council already broadcasts meetings of Council and Planning Committee both of which meetings take place in the same location where appropriate equipment is installed. The Council does hold other meetings such as scrutiny in other areas of the authority from time to time, depending on the subject matter being discussed, in order to make it easier for local people to attend. The Council is concerned that a requirement to broadcast all of its public meetings will either incur significant extra capital cost for equipment to be installed in several buildings, or that some of its meetings will be more remote from local people, who may wish to attend in person, than would otherwise be the case. This is especially the case when considering Community Area Committees. There is also an additional revenue cost of broadcasting additional hours of content.

The Council does not see why the Welsh Ministers may wish to make provision for members of the public to film meetings when they are already being broadcast by the Council. The Council is also concerned as to the extent of the facilities that it may be obliged to provide to persons wishing to do this, and is unclear as to the provision relating to live oral commentary.

Allowing the public a right to speak on all matters being discussed at meetings of Community Councils may be problematic for some Community Councils who may not have the resources to manage this. There are some Community Councils where

individuals within the community would monopolise and cause difficulties at such meetings.

With regards to the additional matters upon which views are sought under this part of the consultation document, the Council agrees that County Councils should have a website. The Council notes that these are already in place. It is difficult to imagine the circumstances in which a County Council would not have a website and it seems that this is legislation that is being introduced to solve a problem that does not exist.

The Council agrees with the suggestion that Councils should be able to send out notices of meetings by electronic means alone.

It is agreed that it is sensible to repeal legislation prohibiting Community Councils from meeting in licensed premises as explained in the consultation document. It seems somewhat patronising to require Community Councils to adopt standing orders prohibiting the consumption of alcohol during meetings. There is no similar provision for County Councils or the National Assembly.

Question 3.8: Do you have any comments on our proposals to enhance participation by children and young people through the public participation duty?

The Council agrees that the proposal for Youth Councils should not be included in the Draft Bill and that participation by young people and children should be incorporated under the public participation duty.

PART 4

Question 4.1: Do you have any comments on any of the provisions in Part 4 of the Draft Bill?

The Council wishes to comment on a number of matters included in this Part.

Chapters 2, 3 and 4 - Performance duties and the breach thereof

In its response to the White Paper, the Council confirmed that Members have previously expressed frustration at the lack of a statutory requirement to attend committees more frequently than the statutory minimum threshold for disqualification contained within the Local Government Act 1972, and notes that there is no proposal for this to be changed.

Members support an increased focus on attendance although there is no definition of what a good reason for non-attendance might be.

Members of this Council have agreed a programme of compulsory training, however, they did not agree any sanction for failure to attend. The Council welcomes the fact that Councils will determine what training is compulsory. Again there is no definition

or suggestion given as to what would amount to a good reason. There is a potential resource implication depending on the frequency with which training must be repeated in order to give Members an opportunity to avoid non-attendance due to other commitments.

Members do not agree with the requirement to hold surgeries or to complete annual reports. Members believe that it is up to them to communicate with their constituents in the manner that they and their constituents consider best and not to have any particular method dictated to them. Some Members also believe that the concept of surgeries is outdated and that many of them engage with their constituents by electronic means.

The proposals that any person may complain about a breach of these duties and that these complaints be investigated by the Monitoring Officer has the potential to substantially increase the workload of Monitoring Officers and Standards Committees and lead to a large number of vexatious, frivolous and politically motivated complaints particularly in the run up to election periods.

On a more general point, it is of concern that the Draft Bill appears to be creating a separate standards regime outside of the provisions of the Code of Conduct. The provisions in the Draft Bill give Standards Committees the same powers of sanction that currently exist for breaches of the Code of Conduct. The Draft Bill provides for regulations to be made about the procedure for investigations and hearings in respect of complaints. Regulations already exist in respect of complaints under the Code of Conduct. Is there to be a duplicate process?

There is the potential for duplication of effort in investigation and reporting to Standards Committees. If a complaint is made to the Monitoring Officer or to the Ombudsman about a Member, which contains an allegation that refers to a potential breach of the duties in the Draft Bill and a breach of the Code of Conduct, are there to be two investigations and reports to the Standards Committee? If not, who is to conduct the investigation and make the report?

Would it not be more sensible to amend the current model Code of Conduct to contain the duties that the Draft Bill proposes rather than create an additional system of complaint, investigation and hearing?

Chapter 5

Objectives to be met by Council Executive

The Council agrees that the Leader should set objectives for Cabinet Members. This is the practice in this Council.

Manifestos: Election of Leader

In its response to the White Paper the Council agreed that candidates for the position of Leader should set out their values and priorities and present this to Council before their election. The Council also confirmed that it currently follows this practice.

Assistants to the Executive

In its response to the White Paper the Council agreed that there should be an opportunity for Councillors who wished to gain experience to assist them in becoming Cabinet or Executive members in their own right. The proposals in the Draft Bill, rightly prevent such Members from being members of Scrutiny Committees, and restrict their appointment to Democratic Services Committee and Corporate Governance Committee. In order for this system to work, there have to be sufficient numbers of Members left available to perform the important task of scrutinising and holding the Cabinet/Executive to account. For smaller existing authorities this would be impractical and it is suggested that this proposal's implementation be deferred until the new authorities come into existence.

Chapter 6

Appointment etc. of certain Chief Officers

The Council agrees that there should be generic duties for all Chief Executives and that these should include the functions of the Head of Paid Service.

The Council agrees with the setting of objectives for Chief Executives, but believes, as stated in its White Paper response, that the setting of objectives and performance appraisal of the Chief Executive is conducted by a politically balanced panel of Members and is not the preserve of the Leader. This is the practice in this Council as the Chief Executive is accountable to the whole Council and not just the Leader.

There are certain employment law and personal data safeguards to be considered before deciding to make what is effectively a performance appraisal of the Chief Executive public.

The Council agrees that the Head of Democratic Services should be a Chief Officer and that the bar on the Monitoring Officer being designated as such be removed.

Chapter 7

The Council agrees that the question of whether or not the right to vote is extended to those co-opted members of scrutiny who do not currently have it should be determined locally by each Council. The Council considers that regulations to set minimum and maximum numbers of co-opted members is overly prescriptive and that such questions should be determined locally by each Council.

The Council also considers that a power for Welsh Ministers to make regulations setting out circumstances in which Councils must have joint scrutiny committees is overly prescriptive and an erosion of local democracy.

The Council agrees that Standards Committees should make an annual report to Council. The Standards Committee of this Council already does so.

Question 4.2: Do you have any comments on the proposed duty on leaders of political groups or the monitoring and reporting roles of the Standards Committee?

The Council agrees with a general duty for Group Leaders to take reasonable steps to promote good conduct amongst members of their group and co-operate with the Standards Committee. It should be noted that all Members are individually responsible under the Code of Conduct for their own actions and that a Group Leader can do no more than take reasonable steps to promote good conduct.

s98(3) suggests that Welsh Ministers will make regulations about the circumstances in which Members are to be treated as constituting a political group and in which a Member is treated as being a Group Leader. Regulations already exist in the form of the Local Authorities (Committees and Political Groups) Regulations 1990. Are these to be repealed, and, if so, what are the proposals for the new arrangements?

Question 4.3: Do you have any comments on our proposals in relation to the delegation of functions by Local Authorities?

The Council agrees with the proposal that Part 2 Deregulation and Contracting Out Act 1994 be repealed and replaced with a regime that will allow for the delegation of Council functions to third parties.

Question 4.4: Do you have any comments on our proposal to give the Welsh Ministers a power to direct the IRPW to have regard to guidance when reviewing the remuneration framework for Councillors?

The Council is concerned that the Independent Remuneration Panel for Wales which was set up to provide an independent and objective approach to the question of member remuneration is increasingly perceived as losing its independence and objectivity given the recent draft report which appeared to make findings inconsistent with previous reports and to have made recommendations in response to a ministerial letter. If the Panel is to retain its independence and objectivity it is suggested that it makes evidence based recommendations and is free from interference by Ministers and Councillors alike. If a more diverse pool of councillors is to be attracted to stand for election, particularly those of working age and from all socio-economic groups, it would appear sensible that they have confidence that questions of remuneration will be determined objectively on an evidence based approach. The panel should be restricted to questions of remuneration levels and should not make recommendations which have structural consequences.

Question 4.5: Do you agree the provisions relating to remote attendance in the 2011 Measure should be made more flexible?

It is very hard to provide an intelligent and considered response to such a vague question and in the absence of any information as to what the proposal will be. The Council's previously determined view is that it is not against remote attendance per se, however, the advice it has previously received is that there is not yet a technical solution that is sufficiently robust or reliable to facilitate remote attendance as prescribed in the 2011 Measure.

Question 4.6: Do you have any comments on our proposal that Shadow Authorities should be required to appoint interim Returning Officers?

The Council agrees that allowing shadow authorities to appoint Returning officers is a practical solution in respect of the 2020 General Election, however, it does not understand this to be a devolved matter.

Question 4.7: Do you have any comments on the desirability of giving Councils the power to dismiss the Chief Executive, the Chief Finance Officer, the Monitoring Officer and the Head of Democratic Services through a vote?

The officers described above have a measure of protection due to the nature of the advice that they may have to give to a Council in accordance with their statutory duties. This was designed to prevent them being summarily dismissed as a result of giving inconvenient advice in accordance with those duties. The protection amounts to a requirement that an independent person investigates any allegation brought against the officer and concludes that disciplinary action including dismissal is justified. The proposal is that the requirement for an investigation be dispensed with and a vote at full council be sufficient to dismiss. This raises the prospect of unfair dismissal and other potential legal claims. It seems inconsistent with the provisions of a Draft Bill that is concerned with, amongst other things, good governance, to suggest such a measure.

Question 4.8: Do you have any comments on our proposal to change the framework within which Councils and their Executive determine how their functions are to be allocated?

The Council agrees with the proposal that the current complicated regime of determining allocation of functions between the Council and Executive be amended and that there be a more liberal approach subject to clear guiding principles. There will need to be some form of consistency amongst Councils particularly where they may wish to collaborate or operate joint committee arrangements in the discharge of their functions.

Question 4.9: Do you have any comments on our proposals in relation to the disposal and transfer of Local Authority assets?

The Council welcomes a less rigid approach than that proposed in the White paper. The Council does not object to notice of intended disposal being given, but, in the absence of detailed proposals for how these arrangements would work, the Council would reiterate the point made in response to the White Paper that Councils must be able to retain the right to make the final decision over whether or not to transfer an asset. Care will need to be taken when determining the values above which details of assets are to be published, particularly where the value of assets may be close to the threshold, in order that the commercial interests of the local authority are not prejudiced.

PART 5

Question 5.1: Do you have any comments on any of the provisions in Part 5 of the Draft Bill?

Whilst the intention is clearly to link corporate planning with the new Wellbeing plans required under the Well-Being of Future Generations (Wales) Act, the read across could be better, for example in reporting schedules, or in the identification of stakeholder groups. They could refer directly to each other or even be integrated.

The strategic equality planning should also be included in the integrated reporting proposed, but it is not mentioned. This will be done in practice.

The 3 months given to publish a corporate plan after an election is too short. Councils will need time to work with the new administration and then consult on the plan before publishing it. They will also need to ensure that the plan aligns with the Public Service Board's Wellbeing Plan. If all this is to be meaningful then at least 6 months will be required not 3.

The Bill does not seem to address the question of proliferating indicator sets. There could be better read across with the Well-Being of Future Generations Act, Programme for Government, Social Services Act etc.

There is the danger of significant overlap and duplication of reporting and other requirements between the Draft Bill and these other pieces of legislation.

Whilst the intervention powers for Welsh Ministers are similar to those already in existence, they do not appear to be linked to any evidence based threshold or criteria.

The requirement for at least one third of the Corporate Governance and Audit Committee to be made up of independent lay members appears to be overly prescriptive and contrary to local democracy. In its White Paper response the Council agreed that independent members can bring a fresh perspective but their number should be a matter for local determination. The Council also believes that the Committee should select its Chair and that there should be no prescription in respect of this, either in favour of, or against, independent members.

The Council agrees that regulators should be required to co-ordinate their activities.

There are other matters to which the consultation document refers but which are not included in the Draft Bill even though it is intended that they be introduced into the final Bill. The Council agrees with the proposal that regulators be required to share their reports with scrutiny and attend to present them when invited.

There is insufficient detail in respect of an online information portal to enable meaningful consultation. Clearly, Councils would be concerned if the number and complexity of the information sets required to be published were unduly onerous or a duplication of other reporting requirements.

Question 5.2: Do you have any comments on our proposal to subject Local Authorities to a governance arrangements duty?

The Council agrees with the suggestion that County Councils should be under a duty to make, implement and comply with arrangements for good governance, accountability, economy, efficiency and effectiveness of its use of resources. The Council does however note that there are several proposals contained in the Draft Bill which will significantly increase the bureaucratic resource required to support them.

The Council is concerned that there are reserved to Welsh Ministers powers to make regulations that have the potential to be extremely prescriptive in nature which appears to be inconsistent with the intention expressed in the White Paper that local authorities should become more responsible and accountable for their activities.

Councils should be required to have good governance arrangements but a one size fits all approach should not be imposed upon them.

Question 5.3: Do you have any comments on the model approach to peer assessment set out in Annex A?

The Council would be concerned if the proposals for Peer Assessment were to be too prescriptive.

Question 5.4: Do you have any comments on the proposed role for the Corporate Governance and Audit Committee in relation to the Local Authority's response to the self-assessment, peer assessment, combined assessment and governance review?

The Council welcomed the introduction of self and peer assessment and combined assessments by regulators in the White Paper. The Council already operates a system of self-assessment by means of its service challenge process.

The Council agrees that the Corporate Governance and Audit Committee should have the role described in the Draft Bill in respect of such assessments.

Question 5.5: Do you have any comments on our proposal to reject local public accounts committees?

The Council agrees with the proposal to reject public accounts committees which it considers would have been an unnecessary potential duplication of Councils' own audit and financial management processes.

Question 5.6: Are Public Services Boards the right bodies to examine the policy choices facing local public services?

It is not clear what role is being suggested. Are the Public Services Boards being asked to scrutinise decisions made by councils? Under the Well-Being of Future Generations (Wales) Act 2015 local authorities are to scrutinise the Boards.

There is in the governance duty a requirement of a Council to make, implement and comply with arrangements for economy, efficiency and effectiveness of its use of resources. There is a system of self and peer assessment and combined assessment by regulators. There is a system of internal scrutiny and challenge through Scrutiny Committees.

What would an additional function for Public Services Boards add?

Question 5.7: If so, would they benefit from additional legal powers?

The Council does not believe that additional powers are necessary

Question 5.8: What legislative measures could be considered to enable Local Government to take a public sector-wide shared services role?

Local Government finance officers have previously commented on the limitations of the KPMG study and noted that its conclusions are overly simplistic and the costbenefit conclusion is flawed.

At a time of huge change and potential reorganisation of local authorities it seems a very high risk approach to try and create an all Wales shared support service just when local authorities will need to rely on responsive, stable support services. Larger units of support will be created by virtue of the merger of Councils. It would seem more sensible to establish the new councils before considering further major changes in the provision of support.

PART 6

Question 6.1: Do you have any comments on any of the provisions in Part 6 of the Draft Bill?

The Council's comments are contained in the responses set out below.

Question 6.2: Should the Boundary Commission be required to submit their draft reports to Shadow Authorities from May 2019?

The Council welcomes the fact that it is no longer proposed that Councils conduct these reviews. Whilst it appears sensible to enable these reports to be submitted in a timely fashion, it is the Council's view that the Shadow Authorities will have more than enough to do in establishing the governance and other arrangements for the new councils and will not have the resource or capacity to undertake the work required of it by s153 of the Draft Bill.

Question 6.3: Should the new County Councils implement the Boundary Commission's recommendations or should this be a responsibility of the Boundary Commission itself?

It would appear that the new County Councils will have a significant workload in establishing themselves and the efficient delivery of their services and may not have the resource or capacity to implement these reviews. Allowing the Commission to implement the reviews may assist Councils in this respect.

Question 6.4: Do you have any comments on our proposals relating to compulsory training for Community Councillors?

It is not clear why County Councils should be responsible for identifying and securing the provision of compulsory training to Community Councillors who have been elected in their own right to public authorities that are independent of County Councils. This is an additional burden for County Councils.

Question 6.5: Do you have any comments on our proposal to extend the term of Community Councillors elected in 2017 to six years?

It would appear sensible to ensure that election dates for Community Councils are consistent with those for County Councils.

Question 6.6: Do you have any comments on our proposal that Community Councils should be required to consider and plan for the training needs of their own members and employees?

This appears to be more sensible than, but contradictory to, the proposal that County Councils identify and provide compulsory training for Community Councillors.

Question 6.7: Do you have any comments in relation to the setting of objectives for a Community Council clerk?

If Community Councils are to become bigger, it seems appropriate and in accordance with good practice that the Clerk have objectives set. The Council considers that this should be done by the Council rather than the Chair, as the Clerk owes a duty to the Council as a whole, and since the Chair changes annually it will provide some consistency of approach. The Council considers that the same approach should be taken for all Community Councils.

Question 6.8: Do you have any comments on our proposal to repeal the legislation relating to community polls and to require instead that Local Authorities should implement a system of e-petitions?

The legislation for community polls is cumbersome and outdated. There will need to be safeguards against frivolous or vexatious petitions and the thresholds to be crossed before Councils are required to respond.

PART 7

Question 7.1: Do you have any comments on any of the provisions in Part 7 of the Draft Bill?

Any guidance issued on workforce matters by Welsh Ministers should not be so prescriptive that it hinders local authorities' ability to shape service delivery in accordance with local requirements. Councils will have a duty to operate governance arrangements that delivery economy, efficiency and effectiveness of the use of resources including their workforce. Their ability to do this should not be unnecessarily hindered.

Question 7.2: Do you have any views on whether it would still be desirable to establish a statutory Public Services Staff Commission if it would be more constrained in the matters on which it could issue guidance than a non-statutory Commission?

What purpose would there be in establishing a statutory commission in such circumstances?

PART 8

Question 8.1: Do you have any comments on any of the provisions in Part 8 of the Draft Bill or on any of the Schedules?

Part 8 refers to the fact that the term "county council" is to be read as a reference to an existing authority until April 2020. The Government should make it clear which provisions of the Draft Bill will be implemented before the creation of the new councils. Given the nature and extent of change that is envisaged in the structures of councils in this Draft Bill, it would appear sensible to defer the implementation of major changes to the way in which councils are structured until after April 2020. Councils will have little capacity to manage and implement these changes whilst also supporting transition committees and shadow authorities. It would appear wasteful to commit significant resources to changes to committee and other structures which may themselves be changed by shadow authorities and new councils.

ACCOMPANYING DOCUMENTS

Explanatory Notes and an Explanatory Memorandum are printed separately.

Local Government (Wales) Bill

DRAFT FOR CONSULTATION

CONTENTS

PART 1

LOCAL GOVERNMENT AREAS AND COUNTY COUNCILS

CHAPTER 1

LOCAL GOVERNMENT AREAS

1 The local government areas

CHAPTER 2

COUNTY COUNCILS

- 2 Constitution of county councils
- 3 The chair of a county council
- 4 Election of the chair
- 5 The vice-chair of a county council
- 6 Chair entitled to title of "mayor"
- 7 Presiding member
- 8 Deputy presiding member
- 9 Title of chair etc. where council has determined to have a presiding member
- 10 Ordinary elections of councillors and term of office
- 11 Electoral wards
- 12 Change of county name by county council

CHAPTER 3

ESTABLISHING THE COUNCILS FOR THE NEW COUNTIES

Interpretation

13 New counties and new councils

First ordinary elections of councillors

14 First ordinary elections of councillors of the new councils

Functions of new councils

15	New	councils:	shadow	authorities

- 16 New councils: finance
- 17 Transfer of functions: application of other legislation in relation to new councils

Transitional provision

18 Transfer of staff, property and liabilities, and other transitional provision

CHAPTER 4

MISCELLANEOUS PROVISION

- 19 Welsh Ministers' power to make consequential, supplementary etc. provision
- 20 Amendments of dates specified in certain Acts
- 21 Application of 1972 Act to the county councils in Schedule 1
- 22 Interpretation of Part

PART 2

GENERAL POWER OF COMPETENCE

CHAPTER 1

THE GENERAL POWER

- 23 Local authority's general power of competence
- 24 Boundaries of the general power
- 25 Limits on charging in exercise of general power
- 26 Limit on doing things for commercial purpose in exercise of general power
- 27 Power to make supplemental provision
- 28 General power of competence: repeal of local authority powers relating to promotion of well-being
- 29 General power of competence: other minor and consequential amendments
- 30 Interpretation

CHAPTER 2

COMMUNITY COUNCILS WITH COMPETENCE

- 31 Community councils with competence
- 32 Continuing or ceasing to be a community council with competence
- Community councils that cease to be community councils with competence
- 34 Power to amend this Part
- 35 Guidance

PART 3

PROMOTING ACCESS TO LOCAL GOVERNMENT

CHAPTER 1

OVERVIEW OF PART

36 Overview

CHAPTER 2

PUBLIC PARTICIPATION IN LOCAL GOVERNMENT

Duty to encourage participation in decision making

- Duty to encourage local people to participate in local government
 Strategy on encouraging participation
 Public participation strategy: consultation and review
 - Budget calculations of county councils under Local Government Finance Act 1992
- 40 Consultation by county councils in respect of budget requirement

General

- 41 Guidance on exercise of functions under this Chapter
- 42 Meaning of "local people"

CHAPTER 3

COMMUNITY AREA COMMITTEES

Appointment and membership

- Duty of county council to appoint community area committees
 County council members of committees
- 45 Community council members of committees
- 46 Community nominee members of committees
- 47 Policy for appointments to community area committees
- 48 Term of office and termination of membership

Proceedings and publicity

- 49 Proceedings of community area committees
- Meetings of community area committees: access and attendance
- 51 Arrangements for public participation

Functions

52 53 54 55 56 57 58	Duty of community area committee to prepare statement of priorities and objectives Further provision about preparation of statement of priorities and objectives Power to confer additional advisory functions on community area committee Delegation of functions of county council to community area committee Request to delegate functions to community area committee Power to require or restrict delegation of functions to community area committee Delegation of functions by community area committee
	General
59 60 61 62 63	Guidance Functions of county councils under this Chapter Power to apply enactments to community area committees Interpretation of this Chapter Further amendments and repeals
	CHAPTER 4
	IMPROVEMENT REQUESTS
	Requests to discuss how outcomes might be improved
64 65 66	Definitions Improvement requests Requirements for improvement requests
	Decisions about improvement requests
67 68 69	Deciding whether to agree to an improvement request Informing interested bodies whether an authority has agreed to an improvement request Entering into discussions with the interested body
	Complaints
70	Complaints
	Reporting
71 72	Reporting on the outcome of improvement requests Annual reports
	General
73 74	Guidance Regulations

CHAPTER 5

ACC	FSS TO	MEETING	S FTC	. OF LOCA	A AIITE	JORITIES
Λ	LOGIO		10 1710	. OF LOCA	$\mathbf{u}_{\perp} \wedge \mathbf{u}_{\perp}$	

- 75 Duty of elected mayors and executive leaders to attend public meetings
- 76 Electronic broadcasts of meetings of certain local authorities
- 77 Public access to meetings of certain local authorities
- 78 Participation at meetings of community councils

CHAPTER 6

FURTHER DUTIES OF COUNTY COUNCILS

- 79 Duty to publish constitution guide
- 80 Duty on county councils to publish official addresses

PART 4

FUNCTIONS OF COUNTY COUNCILS AND THEIR MEMBERS

CHAPTER 1

OVERVIEW OF PART

81 Overview

CHAPTER 2

DUTIES ON MEMBERS OF COUNTY COUNCILS

Performance duties

- 82 Members of county councils to attend meetings
- 83 Members of county councils to hold surgeries
- Members of county councils to answer correspondence
- 85 Members of county councils to complete training

Duty to make an annual report

86 Members of county councils to make annual reports

CHAPTER 3

BREACHES OF DUTIES UNDER CHAPTER 2

Possible breach of performance duties

- 87 Complaints about a member's non-compliance with duties
- 88 Deciding whether to investigate possible breach of performance duties

107

	Possible breach of duty to make annual report
89	Deciding whether to investigate possible breach of duty to make annual report
	Investigations, reports and sanctions
90	Investigation by monitoring officer
91	Monitoring officer's report
92 93	Consideration of matter by standards committee Standards committee's determination
	CHAPTER 4
	FURTHER PROVISION ABOUT DUTIES ON MEMBERS
	Regulations and guidance relating to Chapters 2 and 3
94 95	Regulations Guidance
	Related amendments
96 97	Standards committee to provide training Amendments to the 2011 Measure
	Role of leaders of political groups on county councils
98	Duties of leaders of political groups in relation to standards of conduct
	CHAPTER 5
	EXECUTIVES OF COUNTY COUNCILS
99	Objectives to be met by council executives
100 101	Manifestos: election of executive leader Appointment of assistants to executive
102	Guidance
	CHAPTER 6
	APPOINTMENT ETC. OF CERTAIN CHIEF OFFICERS
103	Appointment and reports of chief executive
104 105	Council leaders to set objectives for chief executives etc. Head of democratic services
	CHAPTER 7
	OVERVIEW AND SCRUTINY COMMITTEES AND STANDARDS COMMITTEES

Voting rights of co-opted members of overview and scrutiny committees

Making information available to overview and scrutiny committees

108 109	Power to require authorities to appoint joint overview and scrutiny committees Duty of standards committee to make annual report
	CHAPTER 8
110	MINOR AMENDMENTS TO OTHER LEGISLATION Minor amendments to other legislation
	PART 5
	COUNTY COUNCILS: IMPROVEMENT OF GOVERNANCE
	CHAPTER 1
111	DUTY TO MAKE ARRANGEMENTS TO SECURE GOOD GOVERNANCE ETG Governance arrangements
	CHAPTER 2
	CORPORATE PLANS
	Preparation, publication and review of corporate plans
112 113 114 115	Duty of county councils to prepare and publish a corporate plan When corporate plan is to be published Review and revision of corporate plan Consultation about corporate plan
	CHAPTER 3
	ASSESSMENTS OF GOVERNANCE ARRANGEMENTS
	Self assessment
116 117	Duty of county council to carry out self assessment Self assessment report
	Peer assessment
118 119 120	Duty of county council to arrange peer assessments Frequency of peer assessments Peer assessment report
121 122	Council's response to peer assessment report Further provision about peer assessment report and response
	Combined regulators' assessment
123	Definitions of "relevant regulator" and "relevant functions"

124 125 126 127	Duty to carry out combined assessment of governance arrangements of county council Report of combined assessment Council's response to combined assessment report Further provision about combined assessments
	CHAPTER 4
	GOVERNANCE REVIEWS AND INTERVENTION
	Governance reviews
128 129 130 131 132 133 134	Welsh Ministers' power to arrange governance review Consultation to be carried out before making arrangements for governance review Duty to notify council of arrangements for governance review Council's duty to co-operate with governance reviewer Provision of information by other persons Reviewer's duty to prepare report Council's response to governance review report
	Intervention directions
135 136 137 138 139 140 141 142	Welsh Ministers' power to give intervention directions Direction to obtain support services Direction to take specified steps Direction that functions be performed by Welsh Ministers or their nominee General provision about intervention directions Duty to co-operate Powers of entry and inspection Interpretation of Chapter
	CHAPTER 5
143	CO-ORDINATION BETWEEN REGULATORS Co-ordination of relevant functions of relevant regulators
	CHAPTER 6
	MISCELLANEOUS PROVISION ABOUT CHAPTERS 1 TO 5
144 145 146 147	Guidance Exercise of functions under this Part Interpretation of Part Amendment of Local Government (Wales) Measure 2009
	CHAPTER 7
	CORPORATE GOVERNANCE AND AUDIT COMMITTEES
148 149 150	Renaming of audit committees Functions of corporate governance and audit committee Membership of corporate governance and audit committee

- 151 Meaning of lay person
- 152 Proceedings etc.

PART 6

COMMUNITY COUNCILS

CHAPTER 1

REVIEW	OF	COMMUNITY	COUNCIL	ARRANGEN	MENTS

153	Duty to review community council arrangements
154	Conduct of community council review
155	Pre-review notification
156	Consultation, investigation and report
157	Final report on review
158	Implementation of Commission recommendations by county council
159	Community council arrangements order if Commission does not submit report
160	Implementation by Welsh Ministers if county council does not implement
161	Implementation orders: transitional, consequential etc. provision
162	Variation and revocation of implementation orders
163	General consequential, transitional etc. provision
164	Transfer of staff
165	Directions and guidance
166	Modification and amendment of section 30 of the 1972 Act

CHAPTER 2

MEMBERS OF COMMUNITY COUNCILS TO COMPLETE TRAINING

Training for members of community councils

167 County councils to consider the training needs of community councillors and to provide training

Duty to complete training

Members of community councils to complete training
 Complaints about a member's failure to complete training
 Notifying the community council that a member has failed to complete training

CHAPTER 3

COMMUNITY COUNCIL ELECTION DATES

171 Change to date of community council elections and extension of term of office

PART 7

WORKFORCE MATTERS

CHAPTER 1

	ON WORKFORCE	$\mathbf{X} \mathbf{I} \mathbf{X} \mathbf{T} \mathbf{T} \mathbf{T} \mathbf{D} \mathbf{C}$
1 PLIBLIC BUILDING		$N/I \Delta + I + I + I + I + I + I + I + I + I $

172	Ministerial guidance to public bodies on workforce matters
173	Meaning of "workforce matters"
174	Meaning of "public body"
175	Power to amend definition of "public body"

CHAPTER 2

PUBLIC SERVICES STAFF COMMISSION

176	Public services staff commission
177	Commission's functions etc.
178	Repeal of this Chapter

PART 8

GENERAL

179 180 181 182 183 184	Interpretation Application of Act in relation to counties and county boroughs until 1 April 2020 Power to make consequential and transitional provision etc. Regulations and orders under this Act Coming into force Short title		
Scher Scher Scher Scher Scher Scher	dule 2 dule 3 dule 4 Part 1 Part 2 dule 5 Part 1 Part 2 dule 6 dule 7 dule 8		Counties in Wales New councils: first ordinary elections and meetings New councils: finance Transitional etc. provision Miscellaneous transitional provision Transfer of staff, property, rights and liabilities The 1972 Act: amendments and transitional provision Minor and consequential amendments Provision related to repeal of section 26 of the 1972 Act (elections) Assistants to local authority executives: further amendments Consequential amendments: chief executive Community area committees: further amendments and repeals Renaming of county council audit committees

Local Government (Wales) Bill

DRAFT FOR CONSULTATION

An Act of the National Assembly for Wales to make provision about local government.

Having been passed by the National Assembly for Wales and having received the assent of Her Majesty, it is enacted as follows:

PART 1

LOCAL GOVERNMENT AREAS AND COUNTY COUNCILS

CHAPTER 1

LOCAL GOVERNMENT AREAS

1 The local government areas

5

10

15

20

25

30

- (1) For the administration of local government on and after 1 April 2020 the local government areas in Wales are
 - (a) the counties set out in the table in Schedule 1, and
 - (b) the communities established under section 20 of the 1972 Act as originally enacted (subject to any changes to those communities which have occurred since that section came into force or which may occur after this section comes into force).
- (2) Each of the counties has the name given to it in the first column of the table in Schedule 1.
- (3) The areas of the counties are set out in the second column of the table in Part 1 of Schedule 1, and are determined by reference to the counties and county boroughs which, immediately before 1 April 2020, were local government areas under the 1972 Act.
- (4) The counties and county boroughs established by the 1972 Act (as amended by the Local Government (Wales) Act 1994 (c.19)) as counties and county boroughs in Wales, other than the county of Powys, cease to exist on 1 April 2020.
- (5) The councils of the counties and county boroughs abolished by subsection (4) also cease to exist on 1 April 2020.
- (6) Nothing in this Act, or done under this Act, affects the establishment or continued existence of the county of Powys, or its council.

CHAPTER 2

COUNTY COUNCILS

2 Constitution of county councils

- (1) Each of the counties has a council consisting of
 - (a) a chair and councillors, or

10

15

20

30

35

- (b) where a council is operating executive arrangements which involve a mayor and cabinet executive, an elected mayor, a chair, and councillors.
- (2) In this Act, "county council" means a council for a county set out in Schedule 1.
- (3) Each county council is a body corporate and has the functions given to it by this Act, the 1972 Act, or otherwise.
- (4) Each county council has the name of the county with the addition of the words "County Council" or, if the council decides, the word "Council".

3 The chair of a county council

- (1) The chair of a county council must be elected annually by the council from among the councillors.
- (2) A member of the executive of a county council, or an assistant to the executive, may not be elected as the chair of the council.
- (3) The chair, unless the chair resigns or becomes disqualified, continues in office until a successor becomes entitled to act as chair.
- (4) During the chair's term of office the chair continues to be a member of the council, despite the provisions of this Act and the 1972 Act relating to the retirement of councillors.
- (5) The chair of a county council has precedence in the area of that council, but not so as to affect Her Majesty's prerogative prejudicially.
- (6) Subsection (5) has effect in relation to a county council which is operating executive arrangements which involve a mayor and cabinet executive as if it provided for the elected mayor of the council to have precedence in the area of that council.
- (7) Subsection (6) does not apply if the executive arrangements of the county council provide for it not to apply.

25 4 Election of the chair

- (1) The election of the chair must be the first business transacted at the annual meeting of a county council.
- (2) If, apart from section 3(4) or section 5(3), the person presiding at the meeting would have ceased to be a member of the council, that person is not entitled to vote in the election except in accordance with subsection (3).
- (3) In the case of an equality of votes the person presiding at the meeting must give a casting vote in addition to any other vote that person may have.

5 The vice-chair of a county council

- (1) A county council must appoint a member of the council to be vice-chair of the council.
- (2) A member of the executive of a county council, or an assistant to the executive, may not be appointed as the vice-chair of the council.
- (3) A vice-chair, unless the vice-chair resigns or becomes disqualified, holds office until immediately after the election of a chair at the first annual meeting of the council following the annual meeting at which the vice-chair was appointed; and until the vice-

- chair ceases to hold office, the vice-chair continues to be a member of the council despite the provisions of this Act and the 1972 Act relating to the retirement of councillors.
- (4) Subject to any standing orders made by the council, anything authorised or required to be done by, to or before the chair may be done by, to or before the vice-chair.

5 6 Chair entitled to title of "mayor"

- (1) A county council may determine that its chair is to have the title of "mayor".
- (2) If a county council makes such a determination, its vice-chair has the title of "deputy mayor".
- (3) This section does not apply where a county council is operating executive arrangements which involve a mayor and cabinet executive.

7 Presiding member

10

15

20

25

30

35

- (1) A county council may determine to have a presiding member.
- (2) A presiding member is elected by the county council from among the councillors.
- (3) A member of the executive of a county council, or an assistant to the executive, may not be elected as its presiding member.
- (4) The county council may determine
 - (a) the functions of the presiding member, and
 - (b) the term of office of the member (subject to the limits in subsection (6)).
- (5) The functions of the presiding member may, in particular, include any function of the chair of the county council in relation to its meetings and proceedings.
- (6) A presiding member continues in office until—
 - (a) the presiding member's resignation or disqualification,
 - (b) a successor becomes entitled to act as presiding member,
 - (c) the county council determines not to have an office of presiding member, or
 - (d) an ordinary council election under section 10.

8 Deputy presiding member

- (1) This section applies where a county council has determined to have a presiding member.
- (2) The county council must appoint a member of the council to act as deputy to the presiding member ("the deputy presiding member").
- (3) A member of the executive of a county council, or an assistant to the executive, may not be appointed as the deputy presiding member.
- (4) Subject to any standing orders made by the council, anything authorised or required to be done by, to or before the presiding member may be done by, to or before the deputy presiding member.
- (5) A deputy presiding member continues in office until
 - (a) the deputy presiding member's resignation or disqualification,
 - (b) a successor becomes entitled to act as deputy presiding member,

10

15

20

25

30

35

- (c) the council determines not to have an office of presiding member, or
- (d) an ordinary council election under section 10.

9 Title of chair etc. where council has determined to have a presiding member

- (1) This section applies where a county council has determined under section 7 to have a presiding member.
- (2) The chair of the council must have
 - (a) the title of "mayor" (if the chair does not already have that title), or
 - (b) the title of "civic chair".
- (3) The vice-chair of the council must have—
 - (a) the title of "deputy mayor" (if the vice-chair does not already have that title), or
 - (b) the title of "civic vice-chair".
- (4) Subsections (2)(a) and (3)(a) do not apply where the county council is operating executive arrangements which involve a mayor and cabinet executive.

10 Ordinary elections of councillors and term of office

- (1) Councillors of a council of a county are to be elected by the local government electors for the county in accordance with this Act, the 1972 Act and Part 1 of the Representation of the People Act 1983 (c.2).
- (2) The ordinary elections of councillors of county councils take place in 2023 and in every fifth year after 2023.
- (3) The term of office of every such councillor is five years.
- (4) On the fourth day after an ordinary election of councillors of a county council—
 - (a) the persons who were councillors immediately before the election retire, and
 - (b) the newly elected councillors assume office.
- (5) In this section "local government elector" means a person registered as a local government elector in the register of electors in accordance with the provisions of the Representation of the People Acts.

11 Electoral wards

- (1) An "electoral ward" is an area in a county for which members are elected to the county council.
- (2) Each electoral ward returns such number of councillors as is provided for under or by virtue of Part 3 of the Local Government (Democracy) (Wales) Act 2013 (anaw 4), section 22 or 23 of the Local Government (Wales) Act 2015 or Part 4 of the 1972 Act.
- (3) There must be a separate election for each electoral ward.

12 Change of county name by county council

- (1) A county council may by passing a resolution change the name of the county for which it is the council.
- (2) A resolution under subsection (1) must be passed by not less than two-thirds of the

10

15

20

30

35

members of the council voting on the change of name at a meeting of the council specially convened for that purpose, with notice given to the members of the purpose of the meeting.

- (3) Notice of any change of name made by virtue of this section
 - (a) must be sent by the county council to—
 - (i) the Welsh Ministers,
 - (ii) the Director General of the Ordnance Survey,
 - (iii) the Registrar General for England and Wales, and
 - (iv) the Local Democracy and Boundary Commission for Wales, and
 - (b) must be published by the council in such manner as the Welsh Ministers may direct.
- (4) A change of name made by virtue of this section does not affect any rights or obligations of any county council, authority or other person, or render defective any legal proceedings; and any legal proceedings may be commenced or continued as if there had been no change of name.

CHAPTER 3

ESTABLISHING THE COUNCILS FOR THE NEW COUNTIES

Interpretation

13 New counties and new councils

In this Part —

"new council" ("cyngor newydd") means the council for a new county;

"new county" ("sir newydd") means a county set out in the table in Schedule 1, other than the county of Powys.

First ordinary elections of councillors

25 14 First ordinary elections of councillors of the new councils

- (1) The first ordinary elections of councillors of the new councils are to be held on 2 May 2019.
- (2) Schedule 2 makes further provision about the first ordinary elections of councillors of the new councils.

Functions of new councils

15 New councils: shadow authorities

- (1) After the first ordinary elections of councillors of the new councils, each new council is a shadow authority in relation to the county for which it will be the council from 1 April 2020.
- (2) A new council is a shadow authority in relation to a new county until 1 April 2020 (after which, the new council will assume all of the functions of a county council).

Tudalen 161

10

15

20

25

30

35

- (3) Shadow authorities have such functions, and are subject to such arrangements as to funding, as are specified in regulations made by the Welsh Ministers (and section 2(3) is to be read accordingly).
- (4) Subsection (3) does not limit the power of the Welsh Ministers to make regulations under section 19 (consequential, incidental, transitional etc. provision) as regards shadow authorities, or any other authority or other person.
- (5) The Welsh Ministers may issue guidance in relation to the exercise of functions by shadow authorities; and shadow authorities must have regard to guidance issued under this subsection in the exercise of their functions.
- (6) The Local Government (Wales) Act 2015 contains further provision in relation to shadow authorities (see sections 13, 25 to 30 and 38 of that Act).

16 New councils: finance

Schedule 3 makes provision in relation to the new councils about —

- (a) council tax and non-domestic rating,
- (b) valuation lists,
- (c) local non-domestic rating lists, and
- (d) council funds.

17 Transfer of functions: application of other legislation in relation to new councils

- (1) This section has effect for the purpose of modifying relevant legislative provisions; in particular for the purpose of providing for the functions conferred by relevant legislative provisions to be exercisable by, and in relation to, the new councils.
- (2) For the purposes of this section, "relevant legislative provision" means
 - (a) a provision of any primary legislation which received Royal Assent or Royal Approval on or before the day on which this Act received Royal Assent;
 - (b) an instrument which
 - (i) was made under primary legislation on or before the day on which this Act received Royal Assent, and
 - (ii) is of a legislative character but is not in the nature of a local enactment.
- (3) This section is subject to the other provisions of this Part, and to any provision contained in regulations made under this Part.
- (4) In any relevant legislative provision
 - (a) any reference to an area which is the area of a county council or the area of a county borough council, and
 - (b) any reference which is to be construed as a reference to such an area,

is on and after 1 April 2020 to be read as a reference to the area of a new council.

- (5) In any relevant legislative provision
 - (a) any reference to a council of a county or county borough, and
 - (b) any reference which is to be construed as a reference to such a council,

is on and after 1 April 2020 to be read as a reference to a new council.

- (6) In this section "primary legislation" means
 - (a) an Act of Parliament;
 - (b) an Act or Measure of the National Assembly for Wales.

Transitional provision

18 Transfer of staff, property and liabilities, and other transitional provision

Schedule 4-

5

10

15

20

25

30

35

- (a) makes miscellaneous transitional etc. provision in relation to the new councils and abolished councils, and
- (b) provides for the transfer of staff, property, liabilities etc. from abolished councils.

CHAPTER 4

MISCELLANEOUS PROVISION

19 Welsh Ministers' power to make consequential, supplementary etc. provision

- (1) The Welsh Ministers may by regulations make supplementary, incidental, consequential, transitional or saving provision for the general purposes, or any particular purpose, of this Part or in consequence of any of its provisions or for giving full effect to it.
- (2) Regulations under subsection (1) may, in particular, make provision
 - (a) for enabling any person to take any steps which are necessary as a preliminary to the exercise of any powers which the person will be able to exercise by virtue of any provision made by or under this Act;
 - (b) for arrangements to be made before any date specified by or under this Act for securing the satisfactory operation from that date of any provision made by or under this Act;
 - (c) about which person or persons is or are to bear the cost of any arrangements (whether or not made by virtue of this section) for securing the satisfactory operation of any provision made by or under this Act;
 - (d) for or in connection with the transfer of functions and property, rights and liabilities (including criminal liabilities) from an abolished council to a new council or other public body;
 - (e) for or in connection with the transfer of staff, for compensation for loss of office, and with respect to pensions and other staffing matters;
 - (f) with respect to the management or custody of transferred property (whether real or personal) transferred to a new council;
 - (g) in respect of charter trustees;
 - (h) in respect of port health districts and port health authorities;
 - (i) in relation to preserved counties (within the meaning given by section 270(1) of the 1972 Act).

10

15

20

25

30

35

- (3) Regulations under subsection (1) may make provision with respect to
 - (a) the establishment or membership of public bodies and the election or appointment of members of the public bodies;
 - (b) the abolition or establishment, or the restriction or extension, of the jurisdiction of any public body.
- (4) Regulations under subsection (1) may also make provision
 - (a) applying (with or without modification), amending, repealing or revoking (with or without savings) any enactment;
 - (b) for making savings from the effect of any repeal or revocation made under this Act.
- (5) In subsection (4)(a) "enactment" includes a charter, whenever granted.
- (6) Provision under subsection (2)(d) may, among other things, make provision disapplying paragraph 6 or 7 of Schedule 4 in relation to a transfer specified in regulations made under this section.
- (7) Provision under subsection (2)(e) may, among other things
 - (a) disapply paragraph 5 of Schedule 4 in relation to a transfer specified in regulations made under this section;
 - (b) provide that members of staff of an abolished council are to be transferred to the employment of a new council other than the new council for the area which contains the area of the abolished council.

20 Amendments of dates specified in certain Acts

- (1) The Welsh Ministers may by regulations amend
 - (a) any provision of this Part or Chapter 1 of Part 6 (community council reviews) which specifies a date, and
 - (b) sections 2(8) and 39(2) of the Local Government (Wales) Act 2015,

by substituting a later date for the date for the time being specified in that provision.

(2) Regulations under subsection (1) may make consequential amendments to other provisions of this Part.

21 Application of 1972 Act to the county councils in Schedule 1

- (1) In section 270(1) of the 1972 Act (definitions)
 - (a) in the definition of "county"
 - (i) after "means" insert "—

(a)", and

- (ii) after "non-metropolitan county only" insert ", and
 - (b) in relation to Wales, a county set out in Schedule 1 to the Local Government (Wales) Act 2017";
- (b) in the definition of "local authority", omit ", county borough council";
- (c) in the definition of "local government area", in paragraph (b) omit ", county

borough";

- (d) in the definition of "new" after "the Local Government (Wales) Act 1994" insert "or the Local Government (Wales) Act 2017";
- (e) in the definition of "principal area", omit "or county borough".
- (2) In Schedule 5 to this Act
 - (a) Part 1 contains further amendments to the 1972 Act, and
 - (b) Part 2 makes transitional provision relating to the repeal of section 26 of that Act (timing of elections and terms of office), including provision for an ordinary election of councillors of the council for the county of Powys in 2020.

10 22 Interpretation of Part

5

15

20

25

30

35

In this Part –

"abolished council" ("cyngor a ddiddymir") means a council of a county or county borough which is abolished under this Act;

"assistant to the executive" ("cynorthwyydd i'r weithrediaeth") has the same meaning as in Schedule 1 to the Local Government Act 2000 (c.22) (see paragraph 3A of that Schedule);

"local government area" ("ardal llywodraeth leol") means a county or community;

"member" ("aelod"), in relation to a county council which is operating executive arrangements which involve a mayor and cabinet executive, means—

- (a) the elected mayor of the council;
- (b) the chair of the council;
- (c) a councillor of the council;

"new council" ("cyngor newydd") and "new county" ("sir newydd") have the meanings given in section 13.

PART 2

GENERAL POWER OF COMPETENCE

CHAPTER 1

THE GENERAL POWER

23 Local authority's general power of competence

- (1) A qualifying local authority has power to do anything that individuals generally may do, even if that thing is, in nature or extent or otherwise
 - (a) unlike anything a qualifying local authority may do apart from this section;
 - (b) unlike anything that other public bodies may do.
- (2) Where subsection (1) confers power on an authority to do something, it confers power to do it in any way whatsoever, including –

Tudalen 165

10

15

20

25

30

35

- (a) power to do it anywhere in Wales or elsewhere;
- (b) power to do it for a commercial purpose or otherwise for a charge, or without charge;
- (c) power to do it for, or otherwise than for, the benefit of the authority, its area or persons resident or present in its area.
- (3) The generality of the power conferred by subsection (1) on a qualifying local authority is not limited by the existence of any other power of the authority; and any other power of the authority is not limited by the existence of the general power.
- (4) For the purposes of this Chapter, each of the following is a qualifying local authority
 - (a) a county council;
 - (b) a community council with competence (for which see Chapter 2).
- (5) In this section, "individual" means an individual with full capacity.
- (6) References in this Part to the general power are to the power conferred by subsection (1).
- (7) This section is subject to sections 24 to 26 and to any provision made under section 27(3) or (4).

24 Boundaries of the general power

- (1) The general power does not enable a qualifying local authority to do anything that the authority is unable to do by virtue of a pre-commencement limitation.
- (2) Nor does the general power enable a qualifying local authority to do anything that the authority is unable to do by virtue of a post-commencement limitation that is expressed to apply
 - (a) to the general power,
 - (b) to all of the authority's powers, or
 - (c) to all of the authority's powers but with exceptions that do not include the general power.
- (3) The general power does not confer power to make or alter
 - (a) arrangements of a kind that are made, or may be made, by or under Part 6 of the 1972 Act (discharge of functions by local authorities);
 - (b) arrangements of a kind that are made, or may be made, by or under Part 2 of the 2000 Act (arrangements with respect to executives etc.);
 - (c) contracting out arrangements, or any other arrangements that are not arrangements within paragraph (a) or (b), that authorise a person to exercise a function of a qualifying local authority.
- (4) In this section –

"post-commencement limitation" means a prohibition, restriction or other limitation expressly imposed by any statutory provision coming into force after the Assembly year in which this Act was passed;

"pre-commencement limitation" means a prohibition, restriction or other limitation expressly imposed by any statutory provision coming into force before

10

15

20

25

30

35

or in the same Assembly year as the Assembly year in which this Act was passed.

(5) For the purposes of subsection (1), section 111(3) of the 1972 Act (subsidiary powers of local authorities not to include power to raise money) is to be disregarded.

25 Limits on charging in exercise of general power

- (1) The general power confers power on a qualifying local authority to charge for providing a service to a person only if the following conditions are met.
- (2) The first condition is that the service is not one that a statutory provision requires the authority to provide to the person.
- (3) The second condition is that the person has agreed to the service being provided.
- (4) Except in relation to a service provided for a commercial purpose, to the extent that the general power confers a power on a qualifying local authority to charge for the provision of a service, the power is subject to a duty to secure that, taking one financial year with another, the income from charges imposed under it does not exceed the costs of provision.
- (5) The duty under subsection (4) applies separately in relation to each kind of service.
- (6) Subject to the duty under subsection (4), in exercising the power conferred by the general power to charge for providing a service, a qualifying local authority may set its charges as it thinks fit, and may among other things—
 - (a) charge only some persons for providing a service;
 - (b) charge different persons, or different descriptions of persons, different amounts for the provision of a service.

26 Limit on doing things for commercial purpose in exercise of general power

- (1) The general power confers power on a qualifying local authority to do things for a commercial purpose only if they are things that the authority may, in exercise of the general power, do otherwise than for a commercial purpose.
- (2) Where, in exercise of the general power, an authority does things for a commercial purpose, the authority must do them through a company.
- (3) A qualifying local authority may not, in exercise of the general power, do things for a commercial purpose in relation to a person if a statutory provision requires the authority to do those things in relation to the person.
- (4) In this section, "company" means—
 - (a) a company within the meaning given by section 1(1) of the Companies Act 2006 (c.46), or
 - (b) a registered society within the meaning of the Co-operative and Community Benefit Societies Act 2014 (c.14) or a society registered or deemed to be registered under the Industrial and Provident Societies Act (Northern Ireland) 1969 (c.24 (NI)).

27 Power to make supplemental provision

(1) If the Welsh Ministers think that a statutory provision (whenever passed or made)

10

15

20

25

30

35

- prevents qualifying local authorities from exercising the general power, or obstructs them in exercising the general power, the Welsh Ministers may by regulations amend, repeal, revoke or disapply that statutory provision.
- (2) If the Welsh Ministers think that any other power overlaps (to any extent) the general power, then, for the purpose of reducing or removing that overlap, the Welsh Ministers may by regulations amend, repeal, revoke or disapply any statutory provision (whenever passed or made).
- (3) The Welsh Ministers may by regulations make provision preventing qualifying local authorities from doing, in exercise of the general power, anything that is specified, or is of a description specified, in the regulations.
- (4) The Welsh Ministers may by regulations provide for the exercise of the general power to be subject to conditions, either generally or in relation to doing anything that is specified, or is of a description specified, in the regulations.
- (5) The power under subsection (1), (2), (3) or (4) may be exercised in relation to
 - (a) all qualifying local authorities,
 - (b) a particular authority that is a qualifying local authority, or
 - (c) particular descriptions of local authorities that are qualifying local authorities.
- (6) Except as provided for in subsection (7), before making regulations under subsection (1), (2), (3) or (4) the Welsh Ministers must consult
 - (a) whatever qualifying local authorities (if any) they think appropriate,
 - (b) whatever representative persons (if any) they think appropriate, and
 - (c) whatever other persons (if any) they think appropriate.
- (7) The duty imposed by subsection (6) does not apply in the case of regulations made by the Welsh Ministers only for the purpose of amending earlier regulations
 - (a) so as to extend the earlier regulations, or any provision of the earlier regulations, to a particular authority or authorities of a particular description, or
 - (b) so that the earlier regulations, or any provision of the earlier regulations, ceases to apply to a particular authority or to authorities of a particular description.
- (8) Representative persons, for the purposes of subsection (6), are persons who appear to the Welsh Ministers to be representative of local government in Wales.
- (9) This section does not confer power to make provision
 - (a) that amends, repeals, revokes or disapplies a statutory provision contained in this Act;
 - (b) for the delegation or transfer of any function of legislating by order, rules, regulations, or other subordinate instrument.

General power of competence: repeal of local authority powers relating to promotion of well-being

- (1) The 2000 Act is amended as follows.
- (2) In section 1 (meaning of "local authority"), in subsection (1), omit paragraph (b).

- (3) Omit sections 2 and 3.
- (4) Omit section 5.
- (5) In section 6 (power to modify enactments concerning plans etc.)
 - (a) in subsection (1), omit "so far as that enactment has effect in relation to England";
 - (b) in subsection (2), omit "in England" in each place where those words occur.
- (6) Omit section 7.

10

15

20

25

30

35

- (7) In section 9 (procedure for orders)
 - (a) in subsection (1), omit "5 or";
 - (b) omit subsection (2);
 - (c) in subsection (3)
 - (i) for "the preceding provisions of this section" substitute "subsection (1)";
 - (ii) omit "5 or";
 - (iii) omit paragraph (d) (and the word" and" immediately preceding it);
 - (d) in subsections (4) and (6) to (8) omit "5 or" in each place where it occurs.
- (8) Omit section 9A.
- (9) In section 105 (orders and regulations)
 - (a) in subsection (6), omit "5,";
 - (b) in subsection (6A), omit ", 5".
- (10) In the Well-being of Future Generations (Wales) Act 2015 (anaw 2), in Schedule 4 (public services boards: consequential amendments and repeals), omit paragraphs 3 and 4.

29 General power of competence: other minor and consequential amendments

- (1) In the 1972 Act, section 137 (power of local authorities to incur expenditure for certain purposes not otherwise authorised) is amended as follows.
- (2) In subsection (9)
 - (a) after "Subject" insert "(in relation to England)";
 - (b) in paragraph (b), after "council" insert "that is not a community council with competence for the purposes of Part 2 of the Local Government (Wales) Act 2017 (general power of competence)".
- (3) In subsection (10), for the words from ""local authority"" to the end of the subsection, substitute ", in its application to England, "local authority" means a county council, a district council, a London borough council, the Common Council or a parish council."
- (4) In the Local Government Act 2003 (c.26), in section 93 (power to charge for discretionary services), in subsection (9), in paragraph (d), after "council" insert "that is not a community council with competence for the purposes of Part 2 of the Local Government (Wales) Act 2017".
- (5) In section 95 of the Local Government Act 2003 (c.26) (power to trade in function-related activities through a company)
 - (a) in subsection (4), after "section" insert "made by the Secretary of State";

- (b) after subsection (4), insert
 - "(4A) Power conferred by an order under this section made by the Welsh Ministers is only exercisable through a company within the meaning given in section 26 of the Local Government (Wales) Act 2017.";
- (c) in subsection (7), in the definition of "relevant authority", in paragraph (d), after "council" insert "that is not a community council with competence for the purposes of Part 2 of the Local Government (Wales) Act 2017".
- (6) In the Local Government and Public Involvement in Health Act 2007, in Schedule 14 (consequential amendments relating to entities controlled etc. by local authorities), in paragraph 5 (amendments to section 95 of the Local Government Act 2003), in subparagraph (4)(c), after "company"" insert ", for the purposes of subsection (4), ".

30 Interpretation

5

10

15

20

25

30

35

In this Chapter -

"general power" ("pŵer cyffredinol") has the meaning given by section 23;

"qualifying local authority" ("awdurdod lleol cymwys") has the meaning given by section 23;

"statutory provision" ("darpariaeth statudol") means a provision of —

- (a) a Measure or Act of the National Assembly for Wales;
- (b) an Act of Parliament;
- (c) an instrument made under a Measure or Act of the National Assembly for Wales, or under an Act of Parliament.

CHAPTER 2

COMMUNITY COUNCILS WITH COMPETENCE

31 Community councils with competence

- (1) A community council becomes a community council with competence if it passes a resolution
 - (a) that it meets the competency requirements, and
 - (b) that it is a community council with competence.
- (2) The competency requirements are that
 - (a) at least two-thirds of the total number of members of the council have been declared to be elected (whether at an ordinary election or at a by-election),
 - (b) the clerk to the council holds a relevant professional qualification, and
 - (c) the council satisfies the audit condition.
- (3) The audit condition is satisfied if
 - (a) the council has received an unqualified Auditor's opinion on the council's accounts for two consecutive financial years, and
 - (b) one of those opinions has been received by the council within the period of twelve

months ending on the day on which the resolution is passed.

- (4) An Auditor's opinion is unqualified if the Auditor General for Wales has not, in the opinion, expressed in any way that he is not satisfied in relation to the matters set out in section 17 of the Public Audit (Wales) Act 2004 (c.23).
- (5) In this Part —

5

10

15

20

25

30

35

"Auditor's opinion" ("barn Archwilydd") means an opinion provided by the Auditor General for Wales under section 23 of the Public Audit (Wales) Act 2004, having concluded an audit of a community council's accounts for a financial year;

"relevant professional qualification" ("cymhwyster proffesiynol perthnasol") means any qualification prescribed as such by the Welsh Ministers in regulations.

32 Continuing or ceasing to be a community council with competence

- (1) A community council with competence must pass a resolution at each relevant annual meeting of the council—
 - (a) that it meets the competency requirements and continues to be a community council with competence,
 - (b) that it meets the competency requirements but ceases to be a community council with competence, or
 - (c) that it does not meet the competency requirements and ceases to be a community council with competence.
- (2) A community council with competence may pass a resolution at any meeting of the council that it ceases to be a community council with competence.
- (3) A community council that passes a resolution of the kind mentioned in subsection (1)(b) or (c) or (2) ceases to be a community council with competence on the day on which the resolution is passed.
- (4) A community council that fails to pass a resolution under subsection (1) ceases to be a community council with competence on the day of the relevant annual meeting in which it fails to pass the resolution.
- (5) For the purposes of this Part, a "relevant annual meeting" means a meeting of the community council held under paragraph 23 of Schedule 12 to the 1972 Act in a year in which ordinary elections of community councillors to the council occur.

Community councils that cease to be community councils with competence

A community council that ceases to be a community council with competence may continue to exercise the general power of competence in relation to any activity undertaken whilst it was a community council with competence.

34 Power to amend this Part

- (1) The Welsh Ministers may, by regulations, amend this Part for the purposes of
 - (a) adding a competency requirement,
 - (b) removing a competency requirement,
 - (c) changing any of the competency requirements, or

- (d) making provision for a community council to cease to be a community council with competence in circumstances other than those specified in section 32 if any of the competency requirements stop being met.
- (2) Before making regulations under paragraphs (a) to (c) of subsection (1), the Welsh Ministers must consult any bodies representing the interests of community councils they consider appropriate.

35 Guidance

5

10

15

20

25

30

35

A community council must have regard to any guidance issued by the Welsh Ministers about how to exercise functions under this Chapter.

PART 3

PROMOTING ACCESS TO LOCAL GOVERNMENT

CHAPTER 1

OVERVIEW OF PART

36 Overview

In this Part -

- (a) Chapter 2—
 - (i) requires a county council to encourage local people to participate in decision making by the council, and in decision making by other local authorities in the council's area;
 - (ii) provides for a council to prepare and publish a strategy setting out how it is to comply with its duty to encourage participation in decision making;
 - (iii) requires a county council to consult when estimating expenditure for the purpose of calculating its budget requirement for a financial year;
- (b) Chapter 3 provides for the establishment of community area committees to operate within the area of a county council for the purpose of ensuring that community interests and priorities are taken into account by the council in exercising its functions;
- (c) Chapter 4 requires a county council and any authority prescribed by regulations to enter into discussions, in certain circumstances, with other bodies for the purpose of improving local outcomes;
- (d) Chapter 5 makes provision for
 - (i) holding meetings where members of the public may scrutinise the exercise of functions by councils operating executive arrangements;
 - (ii) broadcasting proceedings at meetings of county councils and other local authorities which are open to the public;
 - (iii) giving members of the public the opportunity to speak at meetings of community councils open to the public;

10

15

20

25

35

- (e) Chapter 6 requires a county council—
 - (i) to publish a guide to accompany its constitution and make copies of the guide available on request;
 - (ii) to publish an electronic and postal address for each of its members.

CHAPTER 2

PUBLIC PARTICIPATION IN LOCAL GOVERNMENT

Duty to encourage participation in decision making

37 Duty to encourage local people to participate in local government

- (1) A county council must encourage local people to participate in the making of decisions by the council (including the making of decisions in partnership or in conjunction with any other person).
- (2) A county council must also encourage local people to participate in the making of decisions by authorities connected with the council (including the making of decisions in partnership or in conjunction with any other person).
- (3) For the purposes of subsection (2) each of the following is an authority connected with a county council—
 - (a) a community council for an area in the county council's area;
 - (b) a fire and rescue authority for the county council's area;
 - (c) a National Park authority for a National Park any part of which is in the county council's area.
- (4) The Welsh Ministers may by regulations amend subsection (3) for the purpose of
 - (a) adding a reference to any body or other person, or description of body or person, with functions of a public nature as an authority which is connected with a county council;
 - (b) removing a reference to any body or other person, or description of body or person, from that subsection.
- (5) In this section, a reference to the making of decisions includes a reference to the making of decisions by a person in relation to the exercise of a function delegated to that person by a county council or an authority connected with a council.

30 **Strategy on encouraging participation**

- (1) A county council must prepare and publish a strategy ("a public participation strategy") specifying how it proposes to comply with the duty in section 37(1) and (2).
- (2) A public participation strategy must, in particular, address
 - (a) ways of promoting awareness among local people of the county council's functions and the functions of authorities connected with the council;
 - (b) ways of promoting awareness among local people of how to become a member of the county council or of an authority connected with the council, and what membership entails;

10

15

20

25

30

35

- (c) ways of facilitating access for local people to information about decisions made, or to be made, by the county council or by authorities connected to the council;
- (d) ways of promoting and facilitating processes by which local people may make representations to the county council, or authorities connected with the council, about a decision before, and after, it is made;
- (e) arrangements made, or to be made, for the purpose of the council's duty in section 62 of the 2011 Measure (bringing views of the public to attention of overview and scrutiny committees);
- (f) ways of promoting awareness among members of the county council, and members of authorities connected with the council, of the benefits of using social media to communicate with local people.
- (3) A public participation strategy may address how a county council proposes to comply with a duty imposed by another enactment, including an enactment contained in or under any other provision of this Act.
- (4) In the case of a county council operating executive arrangements which involve a mayor and cabinet executive, a reference in subsection (2) to a member of a county council includes a reference to the elected mayor of the council.

39 Public participation strategy: consultation and review

- A county council's first public participation strategy must be published as soon as is practicable after the coming into force of section 38, and when preparing that strategy the council must consult —
 - (a) local people, and
 - (b) such other persons as it considers appropriate.
- (2) A county council—
 - (a) must review its public participation strategy as soon as is practicable following each ordinary election of councillors to the council, and
 - (b) may review its strategy at any other time.
- (3) In conducting a review of a public participation strategy under subsection (2)(a) a county council must consult
 - (a) local people, and
 - (b) such other persons as it considers appropriate.
- (4) Following a review under subsection (2) a county council may revise its public participation strategy, or replace it with a new strategy.
- (5) But a county council may not revise or replace its public participation strategy following a review under subsection (2)(b)without first having consulted
 - (a) local people, and
 - (b) such other persons as it considers appropriate.
- (6) If a county council revises or replaces a public participation strategy, it must publish the revised or new strategy as soon as is practicable.

Budget calculations of county councils under Local Government Finance Act 1992

40 Consultation by county councils in respect of budget requirement

- (1) A county council must take reasonable steps to consult the persons listed in subsection (2) in relation to the estimate of expenditure to be made by it for a financial year under section 32(2)(a) of the 1992 Act (estimate of expenditure to be incurred in performing functions).
- (2) The persons are —

5

10

15

20

25

30

- (a) local people;
- (b) persons carrying on a business in the council's area;
- (c) representatives of businesses in the council's area;
- (d) persons who provide services to local people on the council's behalf;
- (e) persons who carry out activities in the council's area for the benefit of local people, for which the council provides financial assistance;
- (f) voluntary bodies carrying out activities in the council's area;
- (g) every trade union which is recognised (within the meaning of the Trade Union and Labour Relations (Consolidation) Act 1992 (c.52)) by the council;
- (h) such other persons as the council considers appropriate.
- (3) Calculations made under section 32 of the 1992 Act are not invalid merely because of a failure to comply with subsection (1).
- (4) In subsection (1) the reference to calculations does not include substitute calculations made by virtue of section 37 of the 1992 Act.
- (5) In this section "1992 Act" means the Local Government Finance Act 1992 (c.14).

General

41 Guidance on exercise of functions under this Chapter

- (1) The Welsh Ministers may issue guidance to a county council about the exercise by the council of its functions under this Chapter.
- (2) In exercising those functions a county council must have regard to any guidance issued to it under subsection (1).

42 Meaning of "local people"

In this Chapter, "local people", in relation to a county council, means people who live, work or study in the council's area.

20

25

30

35

CHAPTER 3

COMMUNITY AREA COMMITTEES

Appointment and membership

5 43 Duty of county council to appoint community area committees

- (1) A county council must appoint a committee, to be known as a community area committee, for each community area in the council's area.
- (2) The members of a community area committee are to be appointed in accordance with sections 44 to 46.
- (3) A community area committee is to exercise the functions conferred on it by or under—
 - (a) this Chapter, and
 - (b) Part 2 of the 2000 Act (executive arrangements).
- (4) In subsection (1), "community area" means a community area determined under section 37(5) of the Well-being of Future Generations (Wales) Act 2015 (anaw 2).

15 44 County council members of committees

A community area committee must include every member of the county council elected for an electoral ward all or part of which is in the area of the committee (the "county council members" of the committee).

45 Community council members of committees

- (1) A county council must invite each relevant community council to nominate a person for appointment as a member of a community area committee.
- (2) A "relevant community council", in relation to a community area committee, means a council for a community all or part of which is in the area of the committee.
- (3) In response to an invitation, a relevant community council must either
 - (a) nominate one of its members for appointment to the community area committee, or
 - (b) join with another relevant community council in nominating one of that council's members for appointment.
- (4) If a community is divided into community wards, a member of the council for that community who is nominated for appointment to a community area committee must represent a ward all or part of which is in the area of the committee.
- (5) A county council may appoint people nominated under this section to be community council members of a community area committee.

46 Community nominee members of committees

(1) A county council must invite at least one body in each of the following categories to nominate a person for appointment as a member of a community area committee—

10

15

20

25

30

35

- (a) bodies other than community councils which exercise functions of a public nature in relation to the whole or any part of the area of the committee (whether or not they also exercise functions in relation to any other area);
- (b) voluntary bodies carrying on activities which directly or indirectly benefit the whole or any part of the area of the committee (whether or not they also benefit any other area).
- (2) A county council may also invite any other body with a connection to the area of a community area committee to nominate a person for appointment as a member of the committee.
- (3) A county council may appoint people nominated in response to invitations under this section to be community nominee members of a community area committee.
- (4) The community nominee members of a community area committee may not include more than one person nominated for appointment by any particular body.
- (5) A member of a county council may not be a community nominee member of a community area committee of the council.
- (6) In subsection (5), the reference to a member of a county council includes an elected mayor or an elected executive member of the council.

47 Policy for appointments to community area committees

- (1) A county council must prepare and publish a statement setting out how it proposes to exercise its functions under sections 45 and 46.
- (2) The statement must, in particular, specify the numbers of community council members and community nominee members that the council wishes to appoint to each of its community area committees.
- (3) A county council must exercise its functions under sections 45 and 46 with a view to appointing to each of its community area committees the numbers of community council members and community nominee members specified in the statement.
- (4) But a community area committee is not improperly constituted merely because it does not include the numbers of community council members and community nominee members specified in the statement.

48 Term of office and termination of membership

- (1) The term of office of a community council member or community nominee member of a community area committee is to be determined by the county council.
- (2) If a person who is a county council member of a community area committee ceases to be a member of the county council for an electoral ward all or part of which is in the area of the committee, the person also ceases to be a member of the committee.
- (3) But subsection (2) does not apply to a person who
 - (a) ceases to be a member of the county council by reason of retirement, and
 - (b) not later than the day of retirement, is re-elected a member of the council for an electoral ward all or part of which is in the area of the committee.

10

15

20

25

35

Proceedings and publicity

49 Proceedings of community area committees

- (1) A community area committee must appoint one of its members to chair the committee.
- (2) The person appointed to chair a community area committee may not be a member of the county council's executive (unless all of the members of the committee are members of the executive).
- (3) A community area committee may appoint sub-committees.
- (4) Every member of a community area committee may vote on any question to be decided by the committee.
- (5) Every member of a sub-committee of a community area committee may vote on any question to be decided by the sub-committee.

50 Meetings of community area committees: access and attendance

- (1) The meetings of a community area committee, or of a sub-committee of a community area committee, must be held in the area of the committee.
- (2) A failure to comply with subsection (1) does not affect the validity of anything done by the community area committee or sub-committee.
- (3) A community area committee, or a sub-committee of a community area committee
 - (a) may require officers of the county council to attend before it to answer questions, and
 - (b) may invite other people to attend meetings of the committee or sub-committee.
- (4) An officer of a county council must comply with any requirement imposed under subsection (3)(a).
- (5) A person is not obliged by subsection (4) to answer any question which the person would be entitled to refuse to answer in, or for the purposes of, proceedings in a court in England and Wales.
- (6) A community area committee, or a sub-committee of a community area committee, is a committee or sub-committee of a principal council for the purposes of Part 5A of the 1972 Act (access to meetings and documents of certain authorities, committees and subcommittees).

30 51 Arrangements for public participation

- (1) A community area committee must make arrangements for informing local people about the committee and the exercise of its functions.
- (2) A county council must make arrangements in relation to each of its community area committees that enable local people to bring to the attention of the committee their views on any matter in respect of which the committee is exercising a function.
- (3) A community area committee must, when exercising a function, take into account any views brought to its attention in accordance with arrangements under subsection (2).

10

15

20

25

30

35

Functions

Duty of community area committee to prepare statement of priorities and objectives

- (1) A community area committee must, no later than 6 months after the date of each ordinary election of members of the county council, submit a statement of priorities and objectives to the council.
- (2) The statement must set out what the community area committee considers should be the county council's priorities and objectives in exercising its functions in relation to the area of the committee.
- (3) The statement may also contain recommendations as to steps which the committee considers that the council should take in respect of those priorities and objectives.
- (4) A community area committee must review its statement of priorities and objectives before the end of
 - (a) the period of 12 months starting with the date on which it submitted the statement to the county council, and
 - (b) each period of 12 months starting with the date on which the previous review was completed.
- (5) A community area committee may revise its statement of priorities and objectives at any time, and must submit the revised statement to the county council.
- (6) A community area committee must arrange for its statement of priorities and objectives to be published.
- (7) The Welsh Ministers may by regulations make further provision about the form and content of a statement of priorities and objectives.

Further provision about preparation of statement of priorities and objectives

- (1) In preparing and revising a statement of priorities and objectives, a community area committee must have regard to the most recent assessment published under section 37 of the Well-being of Future Generations (Wales) Act 2015 (anaw 2) (assessments of local well-being) for the area of the county council.
- (2) Before submitting the statement to the county council, the community area committee must consult the following on a draft of the statement
 - (a) such local people, or organisations representing local people, as the committee considers appropriate;
 - (b) such voluntary bodies carrying on activities which directly or indirectly benefit the whole or any part of the area of the committee, or organisations representing such bodies, as the committee considers appropriate;
 - (c) such other persons as the committee considers appropriate.
- (3) The statement that the committee submits to the council must include
 - (a) a description of the consultation carried out under subsection (2), and
 - (b) a summary of the views expressed in response to the consultation.

Power to confer additional advisory functions on community area committee

10

15

20

25

30

35

A county council may confer on a community area committee of the council a function of making reports or recommendations to the council or its executive on any matter relating to the area of the committee.

Delegation of functions of county council to community area committee

- (1) A county council may arrange for a community area committee of the council to exercise a function of the council to which this section applies.
- (2) This section applies to a function of a county council if
 - (a) the function is not the responsibility of an executive of the council, and
 - (b) section 101(1)(a) of the 1972 Act (power to arrange for functions to be discharged by committees, sub-committees or officers) applies to the function.
- (3) Arrangements made by a county council under this section
 - (a) may not provide for a community area committee to exercise a function otherwise than in relation to the area of the committee;
 - (b) do not prevent a function being exercised by the council.
- (4) In this section, a reference to exercising a function includes doing anything which is calculated to facilitate, or is conducive or incidental to, the exercise of the function.
- (5) Sections 14 and 15 of the 2000 Act (discharge of functions under executive arrangements) make provision by virtue of which a community area committee may exercise a function which is the responsibility of an executive of the county council.

Request to delegate functions to community area committee

- (1) A community area committee may
 - (a) request the county council to make arrangements under section 55 for the committee to exercise a function of the council;
 - (b) request the county council's executive, or a member or committee of the executive, to make arrangements under section 14 or 15 of the 2000 Act for the community area committee to exercise a function.
- (2) The recipient of the request must give the community area committee notice of its decision on whether to make the requested arrangements.
- (3) If the recipient decides not to make the requested arrangements, the notice must include the reasons for the decision.

57 Power to require or restrict delegation of functions to community area committee

- (1) The Welsh Ministers may by regulations
 - (a) require arrangements within subsection (2) to be made in relation to a function of a county council;
 - (b) provide that a function may not be the subject of such arrangements;
 - (c) restrict a power to make such arrangements in relation to a function;
 - (d) make provision about the manner in which such arrangements must be made.
- (2) The arrangements are –

- (a) arrangements under section 55;
- (b) arrangements under section 14 or 15 of the 2000 Act for a function to be exercised by a community area committee.

Delegation of functions by community area committee

- (1) A community area committee may arrange for any function that is exercisable by the committee under this Chapter to be exercised by a sub-committee of the committee or by an officer of the county council.
- (2) Where by virtue of subsection (1) a function is exercisable by a sub-committee of a community area committee, the sub-committee may arrange for the function to be exercised by an officer of the council.
- (3) A county council which arranges for a function to be exercised by a community area committee under section 55 may direct that subsection (1) or (2) is not to apply to the function or is not to apply to it in particular cases or circumstances.
- (4) A community area committee which arranges for a function to be exercised by a sub-committee under subsection (1) may direct that subsection (2) is not to apply to the function or is not to apply to it in particular cases or circumstances.
- (5) Arrangements made under this section for the exercise of a function do not prevent its exercise by any person by whom it is exercisable in the absence of such arrangements.

General

20 **59** Guidance

5

10

15

25

30

In exercising functions under this Chapter, each of the following must have regard to any guidance issued to it by the Welsh Ministers —

- (a) a county council;
- (b) an executive of a county council;
- (c) a member or committee of an executive;
- (d) a community area committee;
- (e) a sub-committee of a community area committee;
- (f) a community council.

60 Functions of county councils under this Chapter

The functions conferred on a county council by this Chapter –

- (a) are not to be the responsibility of an executive of the council under executive arrangements;
- (b) are not functions to which section 101 of the 1972 Act (arrangements for discharge of functions by local authorities) applies.

35 61 Power to apply enactments to community area committees

(1) The Welsh Ministers may by regulations provide that for the purposes of an enactment specified in the regulations —

10

15

20

25

30

- (a) a community area committee is to be treated as if it were a committee appointed by a county council under section 102 of the 1972 Act;
- (b) a sub-committee of a community area committee is to be treated as if it were a sub-committee appointed under that section;
- (c) arrangements under section 55 are to be treated as if they were arrangements under section 101(1)(a) of the 1972 Act;
- (d) arrangements under section 58 are to be treated as if they were arrangements under section 101(2) of that Act.
- (2) The regulations may include provision modifying the application of an enactment.
- (3) In this section, "enactment" means a provision contained in any of the following
 - (a) an Act of Parliament;
 - (b) an Act or Measure of the National Assembly for Wales;
 - (c) subordinate legislation within the meaning of the Interpretation Act 1978 (c.30) (including subordinate legislation made under an Act of Parliament or an Act or Measure of the National Assembly for Wales).

62 Interpretation of this Chapter

(1) In this Chapter —

"community council member" ("aelod cyngor cymuned") means a member of a community area committee appointed under section 45;

"community nominee member" ("aelod enwebedig o'r gymuned") means a member of a community area committee appointed under section 46;

"county council member" ("aelod cyngor sir") has the meaning given by section 44;

"local people" ("pobl leol"), in relation to a community area committee, means people who live, work or study in the area of the committee;

"statement of priorities and objectives" ("datganiad o flaenoriaethau ac amcanion") means a statement submitted under section 52(1).

- (2) In this Chapter
 - (a) a reference to the county council, in relation to a community area committee, is a reference to the county council that appointed the committee;
 - (b) a reference to a function which is the responsibility of an executive of a county county is to be interpreted in accordance with section 13 of the 2000 Act.

63 Further amendments and repeals

Schedule 8 contains further amendments and repeals relating to this Chapter.

CHAPTER 4

IMPROVEMENT REQUESTS

Requests to discuss how outcomes might be improved

64 Definitions

- (1) This section defines various terms used in this Chapter.
- (2) An "improvement request" is a request made by an interested body to be permitted to enter into discussions with a relevant authority with a view to improving an outcome or outcomes in the authority's area.
- (3) "Interested body" means
 - (a) a body, other than a body exercising functions of a public nature, that carries out activities primarily for the benefit of a relevant authority's area, part of a relevant authority's area, or any group of persons within a relevant authority's area;
 - (b) a voluntary body;
 - (c) a body of persons or a trust which is established for charitable purposes only;
 - (d) a community council for an area in a relevant authority's area;
 - (e) any other person or body specified in regulations made by the Welsh Ministers.
- (4) An "outcome", in relation to a relevant authority's area, is an outcome
 - (a) that results from, or is contributed to by, the provision of a public service by or on behalf of the authority, or
 - (b) that could be improved by the provision of a public service by or on behalf of the authority.
- (5) "Public service", in relation to a relevant authority, means a service that is provided, or could be provided, to the public by or on behalf of the authority.
- (6) "Relevant authority" means
 - (a) a county council, or
 - (b) any other person or body exercising functions of a public nature that is specified in regulations by the Welsh Ministers.
- (7) "Voluntary body" means a body, other than a body exercising functions of a public nature, the activities of which are not carried out for profit.
- (8) For the purposes of subsection (7), the fact that a body's activities generate a surplus does not prevent it from being a voluntary body so long as that surplus is used for the purposes of its activities or is invested in a relevant authority's area.

65 Improvement requests

A relevant authority must consider an improvement request if —

- (a) it is submitted to the authority in writing by an interested body,
- (b) it complies with the requirements specified in section 66, and

Tudalen 183

20

5

10

15

25

30

35

.5

10

15

20

25

30

35

(c) it complies with any other requirements about improvement requests specified in regulations by the Welsh Ministers.

66 Requirements for improvement requests

- (1) An improvement request must—
 - (a) specify the outcome that might be improved,
 - (b) explain how the outcome might be improved,
 - (c) provide details of any knowledge, expertise or experience the interested body making the request has in relation to the outcome, and
 - (d) set out the reasons why the interested body considers that it should enter into discussions with the relevant authority with a view to improving the outcome.
- (2) An improvement request may be made jointly by two or more interested bodies.
- (3) An improvement request may include a request that one or more relevant authorities other than the authority to which the request is made participate in discussions along with the authority to which the request is made.

Decisions about improvement requests

67 Deciding whether to agree to an improvement request

- (1) A relevant authority must agree to an improvement request unless
 - (a) there are reasonable grounds for refusing the request, or
 - (b) subsection (2) applies.
- (2) This subsection applies where
 - (a) an improvement request (a "new request") is submitted to a relevant authority,
 - (b) the new request relates to matters that are the same, or substantially the same, as matters contained in a previous improvement request (a "previous request"), and
 - (c) the new request was submitted to the authority before the end of the period of two years starting with the date on which the previous request was submitted.
- (3) For the purposes of subsection (2)
 - (a) a new request only relates to matters that are the same, or substantially the same, as matters contained in a previous request if both requests relate to the improvement of the same, or substantially the same, outcome;
 - (b) it is irrelevant whether the body making a new request is the same body as made the previous request.

Informing interested bodies whether an authority has agreed to an improvement request

- (1) A relevant authority must, before the end of the period set out in subsection (2), give an interested body that has made an improvement request a notice setting out—
 - (a) whether it has decided to agree to or to refuse the request, and
 - (b) if it has decided to refuse the request, the reasons for the decision.

(2) The period is—

- (a) 45 days starting with the day on which the relevant authority received the improvement request, or
- (b) such longer period as is agreed between the authority and the interested body.

5 69 Entering into discussions with the interested body

- (1) A relevant authority that agrees to an improvement request must, before the end of the period of 45 days starting with the day on which the notice is given under section 68(1), enter into discussions with the interested body about the request.
- (2) A relevant authority must publish any information of a description specified in regulations by the Welsh Ministers about decisions on whether to enter into discussions with interested bodies under this Chapter.

Complaints

70 Complaints

10

15

20

25

30

35

- (1) A relevant authority must establish a procedure for dealing with complaints relating to improvement requests.
- (2) The Welsh Ministers may make provision by regulations about matters that must be addressed in the complaints procedure.

Reporting

71 Reporting on the outcome of improvement requests

- (1) This section applies where
 - (a) an interested body has submitted an improvement request,
 - (b) the relevant authority has entered into discussions with the interested body as a result of that request, and
 - (c) those discussions have concluded.
- (2) The relevant authority must, as soon as reasonably practicable after the discussions have concluded, publish a report
 - (a) summarising the subject matter of the discussions,
 - (b) recording the outcome of the discussions (including any decisions that were taken), and
 - (c) explaining how the authority intends to keep the interested body informed about any further action taken as a result of the discussions.
- (3) In preparing the report, the relevant authority must seek the views of
 - (a) any interested body that took part in the discussions, and
 - (b) any other person it thinks appropriate.
- (4) The relevant authority must publish the report mentioned in subsection (2) on its website and in any other manner it thinks appropriate.

72 Annual reports

5

10

15

25

30

35

- (1) A relevant authority must publish an annual improvement request report for each financial year.
- (2) An annual improvement request report is a report setting out, in respect of the financial year to which it relates
 - (a) the number of improvement requests the authority received,
 - (b) the number of such requests which the authority
 - (i) agreed to, and
 - (ii) refused,
 - (c) the number of such requests which resulted in a change to a public service, and
 - (d) any action taken by the authority
 - (i) to promote the use of improvement requests, or
 - (ii) to support bodies in making improvement requests.
- (3) An annual improvement request report must be published no later than 30 June following the end of the financial year to which it relates.

General

73 Guidance

A relevant authority must have regard to any guidance issued by the Welsh Ministers about how it should carry out its functions under this Chapter.

20 74 Regulations

- (1) The Welsh Ministers may by regulations make further provision about improvement requests.
- (2) Regulations under subsection (1) may, in particular, make provision about
 - (a) the manner in which requests must be made;
 - (b) the procedure to be followed by relevant authorities in relation to requests;
 - (c) information that must be provided to relevant authorities in connection with improvement requests (in addition to that required by section 66);
 - (d) the procedure to be followed by relevant authorities in relation to requests that include a request of the type mentioned in section 66(3);
 - (e) information that must be included in reports (in addition to that required by sections 71 and 72);
 - (f) ways in which relevant authorities are to promote the use of improvement requests;
 - (g) support that relevant authorities must make available to interested bodies to enable them to make improvement requests and to participate in discussions with relevant authorities:
 - (h) additional support that must be made available to persons or groups of persons in

10

15

20

25

30

35

order to assist them to form interested bodies, to make improvement requests and to participate in discussions with relevant authorities.

- (3) The Welsh Ministers may by regulations
 - (a) amend or repeal any of paragraphs (a) to (d) of section 64(3);
 - (b) amend or repeal section 64(7) or (8);
 - (c) make any amendments to this Chapter (including amendments to any power to make regulations) in consequence of provision made under section 64(3)(e) or (6) (b) or under paragraph (a) or (b) of this subsection.

CHAPTER 5

ACCESS TO MEETINGS ETC. OF LOCAL AUTHORITIES

75 Duty of elected mayors and executive leaders to attend public meetings

In the 2000 Act, at the appropriate place insert the following section –

"15B Public meetings in relation to discharge of functions

- (1) The standing orders of a local authority operating executive arrangements must include provision for requiring the elected mayor, or as the case may be, executive leader of the authority to attend at least one meeting per calendar year at which local people are given an opportunity to put questions to the elected mayor or executive leader about the discharge of functions which, under executive arrangements, are the responsibility of the executive.
- (2) "Local people", in relation to a local authority, means people who live, work or study in the authority's area."

76 Electronic broadcasts of meetings of certain local authorities

- (1) A county council must put in place arrangements for the purpose of ensuring that
 - (a) a broadcast of proceedings at a meeting to which subsection (2) applies is available electronically so that members of the public not in attendance at the meeting can see and hear the proceedings as they take place;
 - (b) the broadcast is available electronically for a reasonable period after the meeting.
- (2) This subsection applies to proceedings at a meeting, or any part of a meeting, of a county council or of a committee or sub-committee of a county council which is open to the public.
- (3) The reference in subsection (2) to a county council, in relation to a council with an executive (within the meaning of Part 2 of the 2000 Act), includes a reference to the executive.
- (4) The Welsh Ministers may by regulations specify proceedings to which subsection (2) does not apply.
- (5) A county council putting into place any arrangements required by subsection (1) must have regard to any guidance issued by the Welsh Ministers.

10

20

25

30

35

- (6) The validity of any proceedings to which subsection (2) applies is not affected by the availability of a broadcast (whether as the proceedings take place or afterwards).
- (7) The Welsh Ministers may by regulations make provision for and in connection with ensuring that proceedings at a meeting of an authority listed in subsection (8), or at a meeting of a committee or sub-committee of such an authority, are broadcast electronically.
- (8) The authorities are
 - (a) a fire and rescue authority for an area in Wales;
 - (b) a National Park authority for a National Park in Wales.
- (9) Regulations under subsection (7) may amend or repeal any provision of
 - (a) the Public Bodies (Admission to Meetings) Act 1960 (c.67);
 - (b) Part 5A of the 1972 Act (access to meetings and documents of local authorities).
- (10) A reference in this section to a committee or sub-committee includes a reference to a joint committee, or a sub-committee of a joint committee.

15 77 Public access to meetings of certain local authorities

- (1) The Welsh Ministers may by regulations make provision for and in connection with allowing persons—
 - (a) to film, photograph or make sound recordings of proceedings at a meeting of an authority to which this section applies, or of a committee or sub-committee of such an authority;
 - (b) to use other means for enabling persons not present at such a meeting to see or hear proceedings at the meeting, as it takes place or later;
 - (c) to report or provide commentary on the proceedings at such a meeting, orally or in writing, so that the report or commentary is available, as the meeting takes place or later, to persons not present at the meeting.
- (2) The regulations may, in particular, make provision
 - (a) for allowing persons to make available to the public or a section of the public using any medium (including the internet) things produced as a result of activities within subsection (1);
 - (b) about the facilities to be made available by authorities to which the regulations apply to enable persons to carry on such activities;
 - (c) about the steps to be taken by persons before carrying on such activities;
 - (d) about the circumstances in which persons may not carry on such activities, including for enabling a person specified in the regulations to prevent them from doing so in the circumstances specified in the regulations.
- (3) The regulations may amend or repeal any provision of
 - (a) the Public Bodies (Admission to Meetings) Act 1960 (c.67);
 - (b) Part 5A of the 1972 Act (access to meetings and documents of local authorities and committees of such authorities).

10

15

20

25

30

35

- (4) This section applies to—
 - (a) a county council;
 - (b) a community council;
 - (c) a fire and rescue authority for an area in Wales;
 - (d) a National Park authority for a National Park in Wales.
- (5) In subsection (1) a reference to a committee or sub-committee includes a reference to a joint committee or sub-committee.
- (6) The reference in subsection (4)(a) to a "county council", in relation to a council with an executive (within the meaning of Part 2 of the 2000 Act), includes a reference to the executive.

78 Participation at meetings of community councils

In Part 4 of Schedule 12 to the 1972 Act (meetings and proceedings of community councils), after paragraph 27 insert –

- "27A(1) This paragraph applies in respect of a meeting or part of a meeting of a community council which is open to the public.
 - (2) The person presiding over the meeting must give members of the public in attendance a reasonable opportunity to make representations about any business to be transacted at the meeting, unless that person considers that doing so is likely to prejudice the effective conduct of the meeting.
 - (3) In complying with sub-paragraph (2), the person presiding over the meeting must have regard to any guidance issued by the Welsh Ministers."

CHAPTER 6

FURTHER DUTIES OF COUNTY COUNCILS

79 Duty to publish constitution guide

- (1) Section 37 of the 2000 Act (local authority constitution) is amended as follows.
- (2) After subsection (1) insert
 - "(1A) A local authority must prepare and keep up to date a document (referred to in this section as their constitution guide) which explains, in ordinary language, the content of their constitution."
- (3) In subsection (2)
 - (a) after the word "must" insert "-
 - (a) publish their constitution and their constitution guide on their website, and
 - (b) ";
 - (b) after "copies of their constitution" insert "and their constitution guide".

10

15

20

25

30

35

- (4) In subsection (3)
 - (a) after "constitution" insert "or, as the case may be, their constitution guide";
 - (b) for the words "who requests" to the end of the subsection, substitute "on request, either free of charge or at a charge representing no more than the cost of providing the copy".

80 Duty on county councils to publish official addresses

A county council must publish on its website, and in any other manner it considers appropriate, an official electronic and postal address for each member of the council, to which correspondence for the member may be sent.

PART 4

FUNCTIONS OF COUNTY COUNCILS AND THEIR MEMBERS

CHAPTER 1

OVERVIEW OF PART

81 Overview

In this Part -

- (a) Chapters 2 to 4 impose duties upon members of a county council which relate to their performance, and make connected provision, including for the enforcement of those duties;
- (b) Chapter 5 makes provision
 - (i) for objectives to be set relating to the performance of an executive of a council operating executive arrangements;
 - (ii) for candidates for election as executive leader of a council to produce written manifestos;
 - (iii) for the appointment of assistants to executives where a council is operating executive arrangements;
 - (iv) for issuing guidance to elected mayors and executive leaders on equality and diversity;
- (c) Chapter 6 makes provision
 - (i) removing the requirement for a county council to designate a head of paid service and requiring instead that a chief executive be appointed for the council, whose functions will include duties imposed under that Chapter;
 - (ii) relating to the appointment of the head of democratic services of a county council, and a council's pay policy as it relates to that post;
- (d) Chapter 7 makes provision
 - (i) in respect of the meetings and functions of overview and scrutiny committees;

10

15

20

25

30

35

- (ii) requiring standards committees to produce annual reports on the exercise of their functions;
- (e) Chapter 8 makes minor amendments to the 2000 Act and the Local Government (Democracy) (Wales) Act 2013 (anaw 4).

CHAPTER 2

DUTIES ON MEMBERS OF COUNTY COUNCILS

Performance duties

82 Members of county councils to attend meetings

- (1) A member of a county council must attend all relevant meetings.
- (2) But a member is not in breach of the duty under subsection (1) if the member has a good reason for not complying with the duty.
- (3) For the purposes of subsection (1), each of the following is a relevant meeting
 - (a) a meeting of the county council of which the person is a member;
 - (b) a meeting of a committee or sub-committee of the county council, if the person is a member of that committee or sub-committee;
 - (c) a meeting of any joint committee, joint board or other body of which the person is a member, if that committee, board or body is discharging any of the functions of the county council, or has been appointed to advise the council on any matter relating to the discharge of its functions;
 - (d) if a county council is operating executive arrangements and the person is a member of the executive, a meeting of the executive or of a committee of the executive of which the person is a member;
 - (e) any other meeting that the person would reasonably be expected to attend in the exercise of his or her functions as a member of the county council.
- (4) Subsection (1) does not apply to a member of a county council who is exercising a right to a family absence under Part 2 of the 2011 Measure.
- (5) Nothing in this section affects the operation of section 85 of the 1972 Act (vacation of office because of failure to attend meetings for six months).

83 Members of county councils to hold surgeries

- (1) A member of a county council must hold a surgery at least four times in every relevant 12 month period.
- (2) But a member is not in breach of the duty under subsection (1) if the member has a good reason for not complying with the duty.
- (3) A member holds a surgery if the member makes himself or herself available for at least one hour to meet members of the public to discuss matters in private.
- (4) In complying with subsection (1), the member must ensure that the location, date and time of the surgery is published on the council's website and in any other manner he or she thinks appropriate more than seven days before the day of the surgery.

10

15

25

30

35

- (5) For the purposes of this section, a relevant 12 month period is
 - (a) the period of 12 months starting with the day on which the member assumes office as member of the council, and
 - (b) each subsequent period of 12 months.
- (6) For the purposes of subsection (5), any period during which a member is exercising a right to a family absence under Part 2 of the 2011 Measure is to be disregarded; accordingly, the following two periods are to be treated as consecutive—
 - (a) any part of a 12 month period that falls immediately before the period of family absence, and
 - (b) the period that begins immediately after the period of family absence.

84 Members of county councils to answer correspondence

- (1) A member of a county council must respond to all correspondence sent to his or her official address within 14 days of receipt.
- (2) But a member is not in breach of the duty under subsection (1) if the member has a good reason for not complying with the duty.
- (3) A member's "official address" is any address (whether postal or electronic) which is published by the council under section 80.
- (4) This section does not apply to a member of a county council who is exercising a right to a family absence under Part 2 of the 2011 Measure.

20 85 Members of county councils to complete training

- (1) A member of a county council must complete all compulsory training courses.
- (2) But a member is not in breach of the duty under subsection (1) if the member has a good reason for not complying with the duty.
- (3) Training is compulsory if the county council of which a person is a member has notified the person that he or she must complete it.
- (4) This section does not apply to a member of a county council who is exercising a right to a family absence under Part 2 of the 2011 Measure.

Duty to make an annual report

86 Members of county councils to make annual reports

- (1) A member of a county council must make a report
 - (a) about his or her activities as a member of the council during the period of 12 months to which the report relates;
 - (b) setting out the results of those activities, if known to the member.
- (2) A member of a county council must submit his or her report made under subsection (1) (an "annual report") to the county council's head of democratic services in accordance with the council's standing orders made by virtue of subsection (3)(b).
- (3) A county council must specify in its standing orders –

10

15

20

25

30

35

- (a) the period to which annual reports must relate, and
- (b) when annual reports must be submitted to the head of democratic services.
- (4) Standing orders made under subsection (3)(b) must not permit an annual report to be submitted after the end of the period of three months starting with the day of the annual meeting of the county council which follows the period to which the annual report relates.
- (5) A county council must publish the annual reports submitted by its members.

CHAPTER 3

BREACHES OF DUTIES UNDER CHAPTER 2

Possible breach of performance duties

87 Complaints about a member's non-compliance with duties

- (1) A person may make a complaint about a possible breach by a member of a county council of a duty imposed by section 82, 83, 84 or 85 to the monitoring officer of that council.
- (2) A complaint under subsection (1) must be made in writing.

88 Deciding whether to investigate possible breach of performance duties

- (1) If the monitoring officer of a county council—
 - (a) receives a complaint under section 87, or
 - (b) otherwise has reason to believe that a member has breached a duty under section 82, 83, 84 or 85,

he or she must refer the matter to the chair of the standards committee of the county council.

- (2) The monitoring officer and the chair of the standards committee must together consider whether or not the matter should be investigated.
- (3) When considering whether a matter should be investigated, the monitoring officer and the chair of the standards committee must consider—
 - (a) in cases where the monitoring officer has received a complaint under section 87
 - (i) whether the complaint is frivolous or vexatious, and
 - (ii) whether the complaint relates to the same, or substantially the same, facts as a complaint that has already been disposed of, and
 - (b) in all cases, whether it is appropriate for the matter to be investigated.
- (4) If either the monitoring officer or the chair of the standards committee consider that a matter should be investigated, the monitoring officer must investigate the matter.
- (5) If the monitoring officer and the chair of the standards committee both consider that a matter should not be investigated, no investigation may take place; and, in cases where the monitoring officer has received a complaint under section 87, the monitoring officer must as soon as practicable take reasonable steps to—
 - (a) notify the person who made the complaint that the matter will not be investigated,

10

15

20

25

30

35

and

- (b) give reasons for that decision.
- (6) The monitoring officer and the chair of the standards committee may, for the purpose of deciding whether to investigate a matter, make such enquiries as they think necessary.

Possible breach of duty to make annual report

89 Deciding whether to investigate possible breach of duty to make annual report

- (1) This section applies where the head of democratic services for a county council considers that a member of the county council may have breached the duty under section 86.
- (2) The head of democratic services may
 - (a) notify the monitoring officer of the council, who may investigate the matter;
 - (b) if the head of democratic services is also the monitoring officer of the council, investigate the matter (in the exercise of his or her functions as monitoring officer).
- (3) When considering whether to conduct an investigation under this section, the monitoring officer must consult the chair of the standards committee of the county council on whether it is appropriate for the matter to be investigated.

Investigations, reports and sanctions

90 Investigation by monitoring officer

- (1) This section applies if the monitoring officer of a county council investigates a matter under section 88 (possible breach of performance duties) or 89 (possible breach of duty to make annual report).
- (2) The monitoring officer must give the member who is the subject of the investigation an opportunity to respond to the matter being investigated.
- (3) The monitoring officer may
 - (a) ask any person for information relating to the matter being investigated, and
 - (b) ask any person to assist him or her in conducting the investigation.

91 Monitoring officer's report

- (1) After conducting an investigation under section 88 or 89, the monitoring officer of a county council must—
 - (a) provide the standards committee of the county council with a report on the investigation,
 - (b) make any recommendations to the standards committee that the monitoring officer considers appropriate,
 - (c) send a copy of the report, and any recommendations made, to the member of the county council who was the subject of the investigation, and
 - (d) if the investigation results from a complaint under section 87, take reasonable steps to send a copy of the report, and any recommendations made, to the person who made the complaint.

15

20

25

30

35

- (2) The standards committee must allow the monitoring officer to appear before it for the purpose of presenting the report and any recommendations made.
- (3) The Welsh Ministers may by regulations make provision about the publicity to be given to reports and recommendations provided under this section.

5 92 Consideration of matter by standards committee

- (1) This section applies if the standards committee of a county council receives a report from a monitoring officer under section 91.
- (2) The standards committee must determine whether there is any evidence of a breach of the duty in question.
- (3) If the standards committee determines that there is no evidence of a breach of the duty, it must
 - (a) notify the member who is the subject of the report, and
 - (b) if the report results from a complaint under section 87, take reasonable steps to notify the person who made the complaint.
- (4) If the standards committee determines that there is evidence of a breach of the duty, it must give the member who is the subject of the report written notice that he or she may make representations to the standards committee, either orally or in writing, in respect of the report.
- (5) The standards committee may, for the purposes of exercising its functions under this section
 - (a) ask the monitoring officer to attend before it for the purposes of assisting it, and
 - (b) ask any person for information, or invite any person to make representations to it.
- (6) The monitoring officer may not unreasonably refuse a request made under subsection (5).
- (7) The standards committee must determine the period within which a member may make representations under subsection (4); but the period must not be less than 28 days starting with the day on which the member is given notice under that subsection.

93 Standards committee's determination

- (1) After the period determined under section 92(7) for making representations has ended, the standards committee must determine whether the member who is the subject of the report under section 91 has breached the duty in question.
- (2) If the standards committee determines that the member has breached the duty, it may decide to—
 - (a) censure the member,
 - (b) suspend or partially suspend the member from being a member of the county council for a period not exceeding six months, or
 - (c) take no further action.
- (3) The standards committee must
 - (a) notify the member of its determination under this section, and of any action that will follow, and

- (b) if the report results from a complaint under section 87, take reasonable steps to notify the person who made the complaint of those matters.
- (4) The Welsh Ministers may by regulations make provision about the publicity that is to be given to any determination under this section and to any action taken against a member.
- (5) The Welsh Ministers must by regulations make provision conferring rights of appeal on members of a county council against any determination of a standards committee under this section.

CHAPTER 4

FURTHER PROVISION ABOUT DUTIES ON MEMBERS

Regulations and guidance relating to Chapters 2 and 3

94 Regulations

5

10

15

20

25

30

35

- (1) The Welsh Ministers may by regulations make further provision about the way in which a possible breach of a duty imposed by Chapter 2 is to be dealt with.
- (2) Regulations under subsection (1) may, in particular, make provision
 - (a) about matters to be taken into account in considering whether a possible breach of a duty should be investigated;
 - (b) about matters to be taken into account in determining whether a member has breached a duty;
 - (c) enabling a monitoring officer or a standards committee to refer a possible breach of a duty, or any other matter that comes to the monitoring officer or the standards committee's attention in the course of an investigation, to the monitoring officer or the standards committee of another relevant authority;
 - (d) about procedures to be followed by a monitoring officer or a standards committee;
 - (e) about the disclosure of information;
 - (f) enabling sums to be paid to persons by way of expenses and allowances or costs;
 - (g) about how appeals are to be made, considered and determined.
- (3) For the purposes of subsection (2)(c), a "relevant authority" means—
 - (a) a county council;
 - (b) a fire and rescue authority;
 - (c) a National Park authority for a National Park.

95 Guidance

A person exercising functions under Chapter 2 or 3 must have regard to any guidance issued by the Welsh Ministers.

Related amendments

96 Standards committee to provide training

In section 54 of the 2000 Act (functions of standards committees), before subsection (3)

insert-

5

10

15

20

25

"(2B) A standards committee of a county council in Wales also has the specific function of advising, training or arranging to train members of the council on matters relating to the duties imposed by sections 82 to 86 of the Local Government (Wales) Act 2017 (duties to attend meetings, hold surgeries, answer correspondence, complete training and make annual reports)."

97 Amendments to the 2011 Measure

- (1) The 2011 Measure is amended as follows.
- (2) Section 5 is repealed.
- (3) In section 7, after subsection (1), insert
 - "(1A) A local authority must
 - (a) consider whether it should be compulsory for its members to complete any training under section 85 of the Local Government (Wales) Act 2017, and
 - (b) in exercising its functions under subsection (1), secure the provision of any training which it is compulsory for a member to attend under section 85 of that Act."

Role of leaders of political groups on county councils

98 Duties of leaders of political groups in relation to standards of conduct

- (1) The 2000 Act is amended as follows.
- (2) After section 52 insert –

"52A County councils: duties of leaders of political groups in relation to standards of conduct

- (1) A leader of a political group consisting of members of a county council in Wales
 - (a) must take reasonable steps to promote and maintain high standards of conduct by the members of the group, and
 - (b) must co-operate with the council's standards committee (and any sub-committee of the committee) in the exercise of the standards committee's functions.
- (2) In complying with subsection (1), a leader of a political group must have regard to any guidance issued by the Welsh Ministers.
- (3) The Welsh Ministers may by regulations make provision for the purposes of this section about the circumstances in which—
 - (a) members of a county council in Wales are to be treated as constituting a political group;
 - (b) a member of a political group is to be treated as a leader of the

35

30

group.

- (4) Before making regulations under subsection (3), the Welsh Ministers must consult such persons as they consider appropriate."
- (3) In section 54 (functions of standards committees), after subsection (2) insert
 - "(2A) A standards committee of a county council in Wales also has the specific functions of
 - (a) monitoring compliance by leaders of political groups on the council with their duties under section 52A(1), and
 - (b) advising, training or arranging to train leaders of political groups on the council about matters relating to those duties."

CHAPTER 5

EXECUTIVES OF COUNTY COUNCILS

99 Objectives to be met by council executives

In the 2000 Act, at the appropriate place insert the following section –

"15A Duty to set objectives etc. in relation to discharge of functions

- (1) As soon as practicable after an elected mayor or, as the case may be, executive leader of a local authority is elected, he or she must set objectives to be met by the executive in the discharge of functions which, under executive arrangements, are the responsibility of the executive.
- (2) An elected mayor or executive leader must keep objectives set under this section under review, and may revise the objectives at any time.
- (3) As soon as is reasonably practicable after 31 March in each year, the elected mayor or the executive leader of a local authority must prepare a report on the extent to which objectives set under this section have been achieved.
- (4) An elected mayor or executive leader must arrange for the publication of
 - (a) objectives set under this section, and
 - (b) reports prepared under subsection (3)."

100 Manifestos: election of executive leader

- (1) In paragraph 2 of Schedule 1 (leader and cabinet executives, Wales) to the 2000 Act—
 - (a) insert
 - "(1A) The executive arrangements
 - (a) must include provision requiring a candidate for election as the executive leader to prepare a written manifesto, and provision for ensuring that copies of the manifesto are made available before the election to the other members of the authority;

Tudalen 198

15

5

10

20

25

30

35

10

15

20

25

30

35

- (b) may include other provision with respect to the election of the executive leader.";
- (b) in sub-paragraph (2)
 - (i) after "may" insert "also", and
 - (ii) in paragraph (a) omit "election and".
- (2) After paragraph 2 of Schedule 1 insert
 - "2A The standing orders of a local authority operating executive arrangements which involve a leader and cabinet executive (Wales) must include provision for the purpose of ensuring that a candidate for election as the executive leader is given the opportunity to promote his or her written manifesto at a meeting of the authority held before the election."

101 Appointment of assistants to executive

- (1) Schedule 1 to the 2000 Act (executive arrangements in Wales: further provision) is amended as follows.
- (2) Before paragraph 4 insert –

"Assistants to the executive

- 3A (1) Executive arrangements by a local authority may make provision for councillors of the authority to be appointed to assist the executive in discharging functions which are the responsibility of the executive.
 - (2) Such a councillor is referred to in this Schedule as an assistant to the executive of the authority.
 - (3) Assistants to the executive of an authority are to be appointed
 - (a) in the case of an authority operating a mayor and cabinet executive, by the elected mayor;
 - (b) in the case of an authority operating a leader and cabinet executive (Wales), by the executive leader or the authority.
 - (4) Executive arrangements which make provision for the appointment of assistants to an executive may include provision about—
 - (a) the number of assistants that may be appointed,
 - (b) their term of office, and
 - (c) their responsibilities.
 - (5) The assistants to the executive of a local authority may not include
 - (a) the chairman and vice-chairman of the authority;
 - (b) the presiding member and deputy presiding member of the authority (if the authority has a presiding member).
 - (6) An assistant to the executive of an authority is not a member of the executive of the authority.

- (7) Section 101 of the Local Government Act 1972 (arrangements for discharge of functions by local authorities) does not apply to a local authority's function of making appointments under sub-paragraph (3) (b)."
- (3) In paragraph 5
 - (a) at the beginning insert
 - "(1) An assistant to the executive of a local authority is entitled to attend, and speak at, any meeting of the executive or of a committee of the executive.
 - (2) ";
 - (b) for "not a member of the authority's executive" substitute "neither a member of the authority's executive nor an assistant to the executive".
- (4) Schedule 6 contains further amendments relating to assistants to local authority executives.

15 102 Guidance

- (1) Section 38 of the 2000 Act (guidance) is amended as follows.
- (2) In subsection (1), after "local authority" insert ", an elected mayor or an executive leader".
- (3) After subsection (1) insert
 - "(1A) The guidance may, among other things, include provision designed to encourage good practice in relation to equality and diversity (within the meaning of section 8(2) of the Equality Act 2006)."

CHAPTER 6

APPOINTMENT ETC. OF CERTAIN CHIEF OFFICERS

103 Appointment and reports of chief executive

- (1) A county council must appoint a chief executive.
- (2) The chief executive of a county council must
 - (a) keep each of the matters specified in subsection (3) under review, and
 - (b) where he or she considers it appropriate to do so, prepare a report to the council setting out his or her proposals in respect of any of those matters.
- (3) Those matters are
 - (a) the manner in which the exercise by the council of its different functions is coordinated;
 - (b) the number and grades of staff required by the council for the exercise of its functions;
 - (c) the organisation of the council's staff;
 - (d) the appointment of the council's staff;

Tudalen 200

25

20

5

10

30

35

10

15

20

25

30

35

- (e) the arrangements for the management of the council's staff (including arrangements for training and development).
- (4) As soon as practicable after preparing a report under subsection (2)(b), the chief executive of a county council must arrange for a copy of the report to be sent to each member of the council.
- (5) A county council must consider a report prepared under subsection (2)(b) at a meeting held not more than three months after copies of the report are first sent to members of the council; and section 101 of the 1972 Act does not apply to the duty imposed by this subsection.
- (6) A county council must provide its chief executive with such staff, accommodation and other resources as are, in the chief executive's opinion, sufficient to allow the chief executive's duties under this section to be carried out.
- (7) Schedule 7 makes amendments to other legislation in consequence of this section.

104 Council leaders to set objectives for chief executives etc.

- (1) This section applies where a county council is operating executive arrangements.
- (2) The council's senior executive member must, as soon as is practicable after a person is appointed as chief executive of the council, set objectives to be met by the chief executive.
- (3) The council's senior executive member must annually
 - (a) review the extent to which the chief executive has met objectives set under this section, and
 - (b) review and, if the senior executive member considers it appropriate, revise the objectives.
- (4) The council's senior executive member must arrange for objectives set under this section to be published.
- (5) The council's senior executive member must
 - (a) prepare a report on each review under subsection (3)(a),
 - (b) arrange for the report to be published, and
 - (c) arrange for a copy of the report to be sent to each member of the council.
- (6) The council must consider a report on a review under subsection (3)(a) at a meeting held not more than three months after the date on which the report is published; and section 101 of the 1972 Act does not apply to the duty imposed by this subsection.
- (7) The council's senior executive member may not do anything mentioned in subsection (2), (3), (4) or (5) without consulting its chief executive.
- (8) The Welsh Ministers may issue guidance to—
 - (a) a senior executive member of a county council;
 - (b) a county council,

on the exercise of functions under this section, and the person to whom any guidance is issued must have regard to it.

(9) In this section "senior executive member" means—

- (a) in the case of a council operating a leader and cabinet executive, the executive leader;
- (b) in the case of a council operating a mayor and cabinet executive, the elected mayor.
- (10) In subsection (9) –

"executive leader" ("arweinydd gweithrediaeth") has the same meaning as in section 11(3)(a) of the 2000 Act;

"leader and cabinet executive" ("gweithrediaeth arweinydd a chabinet") means a leader and cabinet executive (Wales) within the meaning of section 11(3) of the 2000 Act.

105 Head of democratic services

- (1) In section 8(4) of the 2011 Measure
 - (a) omit paragraph (b), and
 - (b) in paragraph (c), for "that section" substitute "section 5 of the Local Government and Housing Act 1989".
- (2) In section 43(2) of the Localism Act 2011 (c.20) (meaning of "chief officer" for the purposes of pay policy statements), after paragraph (e) insert—
 - "(f) its head of democratic services designated under section 8(1) of the Local Government (Wales) Measure 2011 (designation by council of a county or country borough in Wales)."

CHAPTER 7

OVERVIEW AND SCRUTINY COMMITTEES AND STANDARDS COMMITTEES

Voting rights of co-opted members of overview and scrutiny committees

- (1) The 2000 Act is amended in accordance with subsections (2) and (3).
- (2) In Schedule 1 (further provision about executive arrangements), at the end insert —

"Overview and scrutiny committees: voting rights of co-opted members

- 15 (1) A local authority may
 - (a) permit a co-opted member of an overview and scrutiny committee of the authority to vote at meetings of the committee:
 - (b) permit a co-opted member of a sub-committee of an overview and scrutiny committee of the authority to vote at meetings of the sub-committee.
 - (2) Permission to vote at meetings of an overview and scrutiny committee may include permission to vote at meetings of its sub-committees (including sub-committees it may appoint in the future).
 - (3) Permission under this paragraph may only be given with the

Tudalen 202

20

106

15

5

10

30

25

35

10

15

20

25

agreement of the overview and scrutiny committee.

- (4) Permission must relate to a specific co-opted member.
- (5) Permission may be given subject to limitations.
- (6) A limitation may, in particular, relate to
 - (a) the questions on which the co-opted member is entitled to vote;
 - (b) the duration of the co-opted member's entitlement to vote.
- (7) The Welsh Ministers may by regulations make provision about the exercise of the functions of local authorities and overview and scrutiny committees under this paragraph.
- (8) Regulations under sub-paragraph (7) may, in particular, include
 - (a) provision for a maximum or minimum number of co-opted members of an overview and scrutiny committee or subcommittee that may be permitted to vote at meetings;
 - (b) provision about the procedure for giving permission.
- (9) Section 101 of the Local Government Act 1972 (c.70) (arrangements for discharge of functions by local authorities) does not apply to the functions of a local authority under this paragraph.
- (10) In this paragraph, "co-opted member", in relation to an overview and scrutiny committee of a local authority or a sub-committee of such a committee, means a member of the committee or sub-committee who is not a member of the authority."
- (3) In section 21(10) (co-opted members of overview and scrutiny committees), after "that meeting" insert "unless permitted to do so under paragraph 15 of Schedule 1".
- (4) In section 58(3)(c)(i) of the 2011 Measure (provisions that may be applied to joint overview and scrutiny committees), after "section 21 of" insert ", or paragraphs 8 to 15 of Schedule 1 to,".
- (5) In section 185(3)(b)(i) of the National Health Service (Wales) Act 2006 (c.42) (provisions that may be applied to joint overview and scrutiny committees with functions relating to the health service), after "(6) to (17) of" insert ", and paragraph 15 of Schedule 1 to,".

30 107 Making information available to overview and scrutiny committees

In section 22(10) of the 2000 Act (power to require information to be made available about decisions), for "or members of the authority" substitute ", members of the authority, an overview and scrutiny committee of the authority or a sub-committee of such a committee".

35 108 Power to require authorities to appoint joint overview and scrutiny committees

- (1) Section 58 of the 2011 Measure (joint overview and scrutiny committees) is amended as follows.
- (2) In subsection (1), after "authorities may" insert "or must".
- (3) In subsection (3), for paragraph (a) substitute –

- "(a) provision about the circumstances in which arrangements may be made;
- (aa) provision prescribing circumstances in which arrangements must be made;
- (ab) provision for arrangements to be made subject to prescribed conditions or limitations;".

109 Duty of standards committee to make annual report

In Part 3 of the 2000 Act, at the end of Chapter 1 insert –

"56B Annual reports by standards committees

- (1) As soon as reasonably practicable after the end of each financial year, a standards committee of a relevant authority must make an annual report to the authority in respect of that year.
- (2) The annual report must describe how the committee's functions have been discharged during the financial year.
- (3) In particular, it must include a summary of
 - (a) what has been done to discharge the general and specific functions conferred on the committee by section 54 or 56;
 - (b) reports and recommendations made or referred to the committee under Chapter 3 of this Part;
 - (c) action taken by the committee following its consideration of such reports and recommendations;
 - (d) notices given to the committee under Chapter 4 of this Part.
- (4) An annual report by a standards committee of a county council in Wales must include
 - (a) the committee's assessment of the extent to which leaders of political groups on the council have complied with their duties under section 52A(1) during the financial year;
 - (b) a summary of
 - (i) matters referred to the chair of the committee under section 88 (possible breach of performance duties by members) of the Local Government (Wales) Act 2017, and
 - (ii) matters considered under section 92 of that Act (consideration by standards committee of possible breaches of duties), and determinations made under section 93 of that Act (standards committee's determination in relation to possible breaches of duties).
- (5) An annual report by a standards committee of a relevant authority may include recommendations to the authority about any matter in respect of which the committee has functions.

15

10

5

20

25

30

35

40

10

15

20

25

30

35

- (6) A relevant authority must consider each annual report made by its standards committee before the end of 3 months beginning with the day on which the authority receives the report.
- (7) The function of considering the report may be discharged only by the relevant authority (and is accordingly not a function to which section 101 of the Local Government Act 1972 applies).
- (8) In this section "financial year" means a period of 12 months ending with 31 March."

CHAPTER 8

MINOR AMENDMENTS TO OTHER LEGISLATION

110 Minor amendments to other legislation

- (1) Section 106 of the 2000 Act (orders and regulations) is amended in accordance with subsections (2) and (3).
- (2) Omit subsection (5).
- (3) In subsection (6)
 - (a) after "made under" insert "section 52A(3),";
 - (b) after "54(5)" insert "or under paragraph 15(7) of Schedule 1".
- (4) In the Local Government (Democracy) (Wales) Act 2013 (anaw 4), omit section 68(4)(a).

PART 5

COUNTY COUNCILS: IMPROVEMENT OF GOVERNANCE

CHAPTER 1

DUTY TO MAKE ARRANGEMENTS TO SECURE GOOD GOVERNANCE ETC.

111 Governance arrangements

- (1) A county council must, in relation to the exercise of its functions, make, implement and comply with arrangements ("governance arrangements") for the purpose of securing
 - (a) good governance;
 - (b) accountability;
 - (c) economy, efficiency and effectiveness in its use of resources.
- (2) In making governance arrangements, a county council must comply with the principles, processes and practices set out or otherwise prescribed by the Welsh Ministers in regulations made under subsections (3) and (4).
- (3) The Welsh Ministers must by regulations make provision about principles, processes and practices relating to good governance, and they may do so by prescribing in the regulations any code of practice or guidance (whether or not issued by the Welsh Ministers).

10

15

20

25

30

35

- (4) The Welsh Ministers may by regulations make provision about principles, processes and practices relating to accountability and the economy, efficiency and effectiveness of a county council in its use of resources, and they may do so by prescribing any code of practice or guidance (whether or not issued by the Welsh Ministers).
- (5) The Welsh Ministers may by regulations make further provision about governance arrangements

CHAPTER 2

CORPORATE PLANS

Preparation, publication and review of corporate plans

Duty of county councils to prepare and publish a corporate plan

- (1) A county council must prepare and publish a corporate plan in accordance with this section.
- (2) The plan must include a statement of the council's priorities in relation to the exercise of its functions (including its priorities in relation to its performance in the short-term, medium-term and long-term).
- (3) The plan must include a statement identifying matters in relation to which the council anticipates making decisions of particular significance.
- (4) The plan must set out information about the council's strategies relating to
 - (a) workforce and workforce planning;
 - (b) financial affairs and financial planning;
 - (c) complaints handling;
 - (d) risk management;
 - (e) overview and scrutiny of the exercise of its functions;
 - (f) management of assets;
 - (g) public participation (as to which, see section 38).
- (5) The plan must, when it is first published, include
 - (a) the documents mentioned in subsection (6), as most recently published, or
 - (b) information on how to obtain those documents (including in electronic form).
- (6) The documents are
 - (a) the self assessment report published by the county council under section 117,
 - (b) the well-being objectives, and the statement relating to them, published by the county council in accordance with sections 3 and 7 of the Well-being of Future Generations (Wales) Act 2015 (anaw 2),
 - (c) the annual report published by the county council under section 13 of that Act, and
 - (d) the assessment of local well-being published by the public services board for the county under section 37 of that Act.

10

15

20

25

30

35

(7) The Welsh Ministers may by regulations make further provision about what must be included in a corporate plan.

113 When corporate plan is to be published

- (1) A county council must publish its first corporate plan no later than three months after the date of the first ordinary election of councillors of the council following the coming into force of this section.
- (2) Subsequently, a county council must publish corporate plans no later than three months after the date of each ordinary election of councillors of the council.
- (3) The Welsh Ministers may by regulations make
 - (a) provision about how a corporate plan is to be published, and
 - (b) further provision about when a corporate plan must be published, including provision amending subsection (1) or (2) for the purpose of changing the deadline by which publication must take place.

114 Review and revision of corporate plan

- (1) A county council must review its corporate plan from time to time (but not less frequently than at intervals of one year).
- (2) As part of each review, the council must assess its progress in relation to the priorities set out in the statement included in the plan under section 112(2).
- (3) On completing the review, the council must publish a revised corporate plan that includes a report of the assessment under subsection (2).
- (4) The council must ensure that the revised plan includes
 - (a) the documents mentioned in section 112(6), as most recently published, or
 - (b) information on how to obtain those documents (including in electronic form).
- (5) The council must also publish a revised plan if it thinks it is appropriate to do so for any other reason.

115 Consultation about corporate plan

Before preparing or revising its corporate plan, a county council must consult –

- (a) each Local Health Board for an area any part of which is in the council's area,
- (b) the public services board for the council's area, and
- (c) any other persons the council thinks appropriate.

CHAPTER 3

ASSESSMENTS OF GOVERNANCE ARRANGEMENTS

Self assessment

Duty of county council to carry out self assessment

(1) A county council must carry out a self assessment from time to time (but not less

frequently than at intervals of one year).

(2) A self assessment is an assessment by the council of its compliance with its duties under section 111.

117 Self assessment report

5

10

15

20

25

30

35

- (1) A county council must prepare a report of each self assessment that it carries out (a "self assessment report").
- (2) A self assessment report must—
 - (a) set out the council's assessment of its compliance with its duties under section 111, and
 - (b) identify what action (if any) the council proposes to take in the light of that assessment.
- (3) The county council must make a draft of the self assessment report available to its corporate governance and audit committee.
- (4) The corporate governance and audit committee must review the draft of the self assessment report and may make recommendations for changes to the draft.
- (5) The county council must, in finalising the self assessment report
 - (a) have regard to any recommendations for changes made by the corporate governance and audit committee, and
 - (b) if it does not make a change recommended by the committee, set out in the self assessment report the recommendation and the reasons why it did not make the change.
- (6) The county council must publish the finalised self assessment report.
- (7) The Welsh Ministers may by regulations make provision about
 - (a) the form and content of a self assessment report;
 - (b) when and how it must be published.

Peer assessment

Duty of county council to arrange peer assessments

- (1) A county council must arrange for peer assessments to be carried out.
- (2) A peer assessment is an assessment, by a panel appointed by the county council, of the council's compliance with its duties under section 111.
- (3) The Welsh Ministers may by regulations make provision about the appointment of panels to carry out peer assessments.
- (4) The regulations may, in particular, include provision about
 - (a) the number of people who may or must be appointed to a panel;
 - (b) descriptions of people who may or must be appointed to a panel;
 - (c) circumstances in which a person may not be appointed to a panel.

119 Frequency of peer assessments

Tudalen 208

- (1) Arrangements made by a county council under section 118 must ensure that a peer assessment is carried out in relation to the council—
 - (a) no later than 31 March 2023, and
 - (b) at least once during each electoral cycle after that date.
- (2) In subsection (1)(b), an "electoral cycle" means a period that
 - (a) begins with an ordinary election of members of a county council, and
 - (b) ends with the next ordinary election of members of the council.
- (3) The Welsh Ministers may by regulations amend subsections (1) and (2) for the purpose of changing when and how often a peer assessment must be carried out.

10 120 Peer assessment report

5

15

25

30

35

- (1) Arrangements made by a county council under section 118 must require a panel that carries out a peer assessment in relation to the council to—
 - (a) prepare a report of the assessment (a "peer assessment report"), and
 - (b) submit the report to the council.
- (2) The peer assessment report—
 - (a) must set out the panel's assessment of the council's compliance with its duties under section 111, and
 - (b) may recommend action to be taken by the council in the light of that assessment.
- (3) The county council must publish the peer assessment report.

20 121 Council's response to peer assessment report

- (1) A county council must prepare a response to each peer assessment report submitted to it.
- (2) The response must
 - (a) state whether the council accepts the assessment and recommendations (if any) set out in the report, and
 - (b) identify what action (if any) the council proposes to take in the light of the report.
- (3) The county council must make a draft of the response to the peer assessment report available to its corporate governance and audit committee.
- (4) The corporate governance and audit committee must review the draft of the response to the peer assessment report and may make recommendations for changes to the draft.
- (5) The county council must, in finalising the response to the peer assessment report
 - (a) have regard to any recommendations for changes made by the corporate governance and audit committee, and
 - (b) if it does not make a change recommended by the committee, set out in the response the recommendation and the reasons why it did not make the change.
- (6) The county council must publish the finalised response to the peer assessment report.

122 Further provision about peer assessment report and response

The Welsh Ministers may by regulations make provision about —

- (a) the form and content of a peer assessment report and response;
- (b) when and how the report and response must be published.

Combined regulators' assessment

5 123 Definitions of "relevant regulator" and "relevant functions"

(1) In this Part —

10

15

20

30

35

- (a) "relevant regulator" means a person listed in subsection (2);
- (b) the "relevant functions" of a relevant regulator means the functions specified in respect of the regulator in that subsection.
- (2) The relevant regulators and their relevant functions are
 - (a) Her Majesty's Chief Inspector of Education and Training in Wales exercising functions under section 38 of the Education Act 1997 (c.44);
 - (b) the Welsh Ministers exercising functions under sections 149A and 149B of the Social Services and Well-being (Wales) Act 2014 (anaw 4);
 - (c) the Auditor General for Wales
 - (i) auditing the accounts of a county council under Chapter 1 of Part 2 of the Public Audit (Wales) Act 2004 (c.23),
 - (ii) undertaking studies under Chapter 2 of Part 2 of that Act in relation to a county council, or
 - (iii) carrying out examinations of a county council under section 15 of the Wellbeing of Future Generations (Wales) Act 2015 (anaw 2).
- (3) The Welsh Ministers may by regulations amend subsection (2) by
 - (a) adding or removing a person or function, or
 - (b) varying the description of a person or function.

25 124 Duty to carry out combined assessment of governance arrangements of county council

- (1) The relevant regulators must, at such intervals as they think fit, carry out a combined assessment in relation to each county council.
- (2) A combined assessment is an assessment of a county council's compliance with its duties under section 111 carried out jointly by the relevant regulators, based on a review by the relevant regulators of
 - (a) reports relating to the council made by each of the relevant regulators in the exercise of their relevant functions,
 - (b) information available to the relevant regulators as a result of exercising their relevant functions in relation to the council, and
 - (c) documents published in relation to the council under this Part.
- (3) A relevant regulator may share reports and information within subsection (2)(a) and (b) with the other relevant regulators for the purposes of a combined assessment.

10

15

20

25

30

35

(4) In carrying out a combined assessment, the relevant regulators may consult any other person who exercises functions of a public nature in relation to the council.

125 Report of combined assessment

- (1) The relevant regulators must prepare a report of each combined assessment (a "combined assessment report").
- (2) The combined assessment report must set out the relevant regulators' assessment of the county council's compliance with its duties under section 111.
- (3) The combined assessment report may also include
 - (a) recommendations to the council about action that it should take in the light of that assessment;
 - (b) recommendations to the Welsh Ministers about the exercise of their functions under this Part in relation to the council.
- (4) The relevant regulators must try to reach agreement about the matters to be set out in the combined assessment report, but where they are unable to do so, the report may set out the views of each of them separately.
- (5) The relevant regulators must send a copy of the combined assessment report
 - (a) to the county council to which the combined assessment relates, and
 - (b) if the report includes recommendations to the Welsh Ministers, to the Welsh Ministers.
- (6) The county council must publish the combined assessment report.

126 Council's response to combined assessment report

- (1) A county council that receives a combined assessment report must prepare a response to the report.
- (2) The response must—
 - (a) state whether the council accepts the assessment and recommendations (if any) set out in the report, and
 - (b) identify what action (if any) the council proposes to take in the light of the report.
- (3) The county council must make a draft of the response to the combined assessment report available to its corporate governance and audit committee.
- (4) The corporate governance and audit committee must review the draft of the response to the combined assessment report and may make recommendations for changes to the draft.
- (5) The county council must, in finalising the response to the combined assessment report
 - (a) have regard to any recommendations for changes made by the corporate governance and audit committee, and
 - (b) if it does not make a change recommended by the committee, set out in the response the recommendation and the reasons why it did not make the change.
- (6) The county council must send a copy of the finalised response to the combined assessment report—

15

20

25

30

35

- (a) to the relevant regulators, and
- (b) if the combined assessment report includes recommendations to the Welsh Ministers, to the Welsh Ministers.
- (7) The county council must publish the response to the combined assessment report.

5 127 Further provision about combined assessments

The Welsh Ministers may by regulations make provision about –

- (a) when a combined assessment must be carried out;
- (b) the form and content of a combined assessment report and response;
- (c) when and how the report and response must be published.

CHAPTER 4

GOVERNANCE REVIEWS AND INTERVENTION

Governance reviews

128 Welsh Ministers' power to arrange governance review

- (1) If the Welsh Ministers consider that a county council is, or may be, failing to comply with any of its duties under section 111, they may arrange for a governance review to be carried out in respect of the council.
- (2) A governance review is a review of a county council's compliance with its duties under section 111, with a view to identifying any action that should be taken to ensure its compliance with those duties.
- (3) Arrangements made by the Welsh Ministers for a governance review to be carried out must
 - (a) describe the scope of the review,
 - (b) identify the person by whom the review is to be carried out (the "reviewer"), and
 - (c) specify a period within which the reviewer is to send a report to the Welsh Ministers under section 133(4).
- (4) The arrangements may include any other provision that the Welsh Ministers think is appropriate.

129 Consultation to be carried out before making arrangements for governance review

Before making arrangements for a governance review to be carried out in respect of a county council, the Welsh Ministers must consult—

- (a) the council, and
- (b) whatever other persons (if any) the Welsh Ministers think appropriate.

Duty to notify council of arrangements for governance review

(1) As soon as reasonably practicable after making arrangements for a governance review, the Welsh Ministers must notify the council concerned of the arrangements, and of the

10

15

20

25

30

35

reasons for making the arrangements.

- (2) The notification must
 - (a) describe the scope of the review,
 - (b) identify the reviewer, and
 - (c) state the period specified under section 128(3)(c).

131 Council's duty to co-operate with governance reviewer

A county council that is the subject of a governance review must provide the reviewer with whatever facilities and assistance the reviewer may reasonably require for the purpose of carrying out the review, including among other things by providing documents, records and other information.

132 Provision of information by other persons

- (1) A reviewer carrying out a governance review in respect of a county council may require a person within subsection (3) to provide the reviewer with whatever facilities and assistance the reviewer may reasonably require for the purpose of carrying out the review, including among other things by providing documents, records and other information.
- (2) That person must take all reasonable steps to comply with the requirement.
- (3) The persons are
 - (a) the public services board for the county council's area;
 - (b) the Local Health Board for an area any part of which is in the county council's area;
 - (c) a community council in the county council's area;
 - (d) Her Majesty's Chief Inspector of Education and Training in Wales;
 - (e) the Auditor General for Wales.
- (4) A person is not required by this section to provide documents, records or other information if the person is prohibited from providing them by any enactment or other rule of law.
- (5) The power in subsection (1) includes power to require documents or records to be produced in a form which is legible and portable.
- (6) The Welsh Ministers may by regulations amend subsection (3) by
 - (a) adding or removing a person, or
 - (b) varying the description of a person.

133 Reviewer's duty to prepare report

- (1) Where a governance review has been carried out in respect of a county council, the reviewer must prepare a report on the review (a "governance review report").
- (2) The governance review report must—
 - (a) set out the reviewer's assessment of the council's compliance with its duties under

10

15

20

25

30

35

section 111,

- (b) identify any action that the reviewer recommends be taken in order to ensure that the council complies with those duties,
- (c) specify the person who the reviewer considers should take the action,
- (d) specify the period within which the reviewer considers that the action should be taken, and
- (e) set out whether the reviewer considers that the action is likely to be taken within that period.
- (3) The report must also contain any other information about the review, or the governance arrangements of the council, that is requested by the Welsh Ministers.
- (4) The reviewer must send a copy of the governance review report to—
 - (a) the county council, and
 - (b) the Welsh Ministers.
- (5) The county council must publish the report within the period of 14 days starting with the date on which a copy of it is sent to the council under subsection (4).

134 Council's response to governance review report

- (1) A county council must prepare a response to any governance review report sent to it under section 133(4).
- (2) The response must identify what action (if any) the council proposes to take in the light of the report.
- (3) The county council must make a draft of the response to the governance review report available to its corporate governance and audit committee.
- (4) The corporate governance and audit committee must review the draft of the response to the governance review report and may make recommendations for changes to the draft.
- (5) The county council must, in finalising the response to the governance review report
 - (a) have regard to any recommendations for changes made by the corporate governance and audit committee, and
 - (b) if it does not make a change recommended by the committee, set out in the response the recommendation and the reasons why it did not make the change.
- (6) The county council must
 - (a) send the finalised response to the governance review report to the Welsh Ministers before the end of the period of 31 days starting with the date on which the copy of the report is sent to the council under section 133(4), and
 - (b) publish the response.
- (7) The Welsh Ministers may extend the period in subsection (6)(a) in any particular case if satisfied that the extension is reasonable to allow the council to discharge its duties under this section.
- (8) A period may be extended under subsection (7) more than once.

10

15

25

30

35

Intervention directions

Welsh Ministers' power to give intervention directions

- (1) The Welsh Ministers may give an intervention direction to a county council if
 - (a) they have received a copy of a governance review report in respect of the council, and the council's response to the report, and
 - (b) they consider that giving an intervention direction is necessary or appropriate in order to ensure that the council complies with its duties under section 111.
- (2) An intervention direction is a direction within section 136, 137 or 138.

136 Direction to obtain support services

- A direction within this section is a direction to a county council to enter into a contract or other arrangement with a specified person, or with a person of a specified description, for the provision to the council of specified support services.
- (2) "Support services" are services that the Welsh Ministers consider to provide information, advice, training, or other support.
- (3) A direction within this section may require the contract or other arrangement to contain specified terms and conditions.

137 Direction to take specified steps

A direction within this section is a direction to a county council to take specified steps, within a specified period.

20 138 Direction that functions be performed by Welsh Ministers or their nominee

- (1) A direction within this section is a direction that specified functions of a county council be exercised by the Welsh Ministers or a person nominated by them.
- (2) If a direction within this section is in force
 - (a) the council to which it relates must comply with the instructions of the Welsh Ministers or their nominee in relation to the exercise of the functions, and
 - (b) the functions of the council to which it relates are to be treated for all purposes as being exercisable by the Welsh Ministers or their nominee.
- (3) The Welsh Ministers may by regulations make provision amending, repealing, revoking or applying any statutory provision (whenever passed or made, including a statutory provision contained in or made under this Act), for the purposes of its application in respect of functions exercisable by the Welsh Ministers or their nominee by virtue of a direction within this section.
- (4) In this section, "statutory provision" means a provision of
 - (a) a Measure or Act of the National Assembly for Wales;
 - (b) an Act of Parliament;
 - (c) an instrument made under a Measure or Act of the National Assembly for Wales, or under an Act of Parliament.

139 General provision about intervention directions

- (1) A county council must comply with an intervention direction given to it.
- (2) An intervention direction
 - (a) must be in writing;
 - (b) may be varied or revoked by a later direction.

140 Duty to co-operate

5

10

15

20

25

30

35

- (1) A county council must provide a person within subsection (2) with whatever facilities and assistance the person may reasonably require in connection with the exercise of functions by virtue of an intervention direction, including among other things by providing documents, records and other information.
- (2) The persons are
 - (a) the Welsh Ministers;
 - (b) any person authorised for the purposes of this section by the Welsh Ministers;
 - (c) any person acting under an intervention direction;
 - (d) any person assisting
 - (i) the Welsh Ministers, or
 - (ii) a person mentioned in paragraph (b) or (c).

141 Powers of entry and inspection

- (1) A person within subsection (2) has at all reasonable times
 - (a) a right of entry to the premises of the county council concerned;
 - (b) a right to inspect, and take copies of, any records or other documents kept by that council, and any documents containing information relating to that council, that the person considers relevant to the exercise of the person's functions under or by virtue of this Chapter.
- (2) The following persons are within this subsection—
 - (a) a person specified in a direction within section 136 given to a county council, or, where a direction within section 136 given to a county council specifies a description of persons, the person with whom the county council enters into the contract or other arrangement required by the direction;
 - (b) the Welsh Ministers, where they have given a county council a direction within section 138 that specified functions of the council be exercised by the Welsh Ministers;
 - (c) a person nominated by a direction within section 138 given to a county council.
- (3) In exercising the right under subsection (1)(b) to inspect records or other documents, a person ("P")
 - (a) is entitled to have access to, and inspect and check the operation of, any computer and any associated apparatus or material which is or has been in use in connection with the records or other documents in question, and

10

15

20

25

30

35

- (b) may require the following persons to provide any assistance P may reasonably require (including, among other things, the making of information available for inspection or copying in a legible form)—
 - (i) a person by whom or on whose behalf the computer is or has been so used;
 - (ii) a person having charge of, or being otherwise concerned with the operation of, the computer, apparatus, or material.
- (4) Any reference in this section to a person falling within subsection (2) includes a reference to any person assisting that person.

142 Interpretation of Chapter

- (1) In this Chapter, "specified", in relation to a direction, means specified in the direction.
- (2) References in this Chapter to "documents" and "records" each include information recorded in any form.

CHAPTER 5

CO-ORDINATION BETWEEN REGULATORS

143 Co-ordination of relevant functions of relevant regulators

- (1) The relevant regulators must have regard to the need for co-ordination in the exercise of relevant functions.
- (2) The Welsh Ministers may by regulations make further provision for and in connection with the co-ordination of the exercise by relevant regulators of relevant functions.
- (3) Regulations made under subsection (2) may, among other things, make provision for
 - (a) a timetable for the exercise of relevant functions to be prepared and published;
 - (b) relevant regulators to be required to adhere to the timetable;
 - (c) information to be shared between relevant regulators in connection with the exercise of relevant functions;
 - (d) relevant regulators to consult one another in relation to the exercise of relevant functions.

CHAPTER 6

MISCELLANEOUS PROVISION ABOUT CHAPTERS 1 TO 5

144 Guidance

A person exercising functions under this Part must have regard to any guidance issued by the Welsh Ministers.

145 Exercise of functions under this Part

(1) The functions conferred on a county council by this Part (other than functions expressly conferred on a committee of a council) may be exercised by the council or by its executive, as the council may determine.

- (2) Section 101 of the 1972 Act (arrangements for discharge of functions by local authorities) does not apply in relation to any function of a county council under this Part (including the function in subsection (1)).
- (3) If a county council determines that any function under this Part is to be exercised by its executive, section 14 or (as the case may be) 15 of the 2000 Act (discharge of functions by executives) does not apply in relation to that function.

146 Interpretation of Part

5

10

15

20

25

35

In this Part –

"combined assessment" ("asesiad cyfun") has the meaning given in section 124 and "combined assessment report" has the meaning given in section 125;

"corporate plan" ("cynllun corfforaethol") is to be interpreted in accordance with section 112;

"governance arrangements" ("trefniadau llywodraethu") has the meaning given in section 111;

"governance review" ("adolygiad llywodraethu") has the meaning given in section 128 and "governance review report" ("adroddiad ar adolygiad llywodraethu") has the meaning given in section 133;

"intervention direction" ("cyfarwyddyd ymyrryd") has the meaning given in section 135:

"peer assessment" ("asesiad gan gymheiriaid") has the meaning given in section 118 and "peer assessment report" ("adroddiad ar asesiad gan gymheiriaid") has the meaning given in section 120;

"public services board" ("bwrdd gwasanaethau cyhoeddus") has the same meaning as in the Well-being of Future Generations (Wales) Act 2015 (anaw 2);

"relevant regulator" ("rheoleiddiwr perthnasol") and "relevant functions" ("swyddogaethau perthnasol") have the meanings given in section 123;

"reviewer" ("adolygydd") has the meaning given in section 128(3)(b);

"self assessment" ("hunanasesu") has the meaning given in section 116 and "self assessment report" ("adroddiad hunanasesu") has the meaning given in section 117.

147 Amendment of Local Government (Wales) Measure 2009

- (1) In the Local Government (Wales) Measure 2009 (nawm 2), in section 1 (meaning of "Welsh improvement authority"), omit paragraph (a).
- (2) In consequence of the amendment in subsection (1), in Part 1 of that Measure
 - (a) in section 4 (aspects of improvement), omit subsection (3)(a);
 - (b) in section 11 (meaning of "powers of collaboration"), omit subsections (1)(b) and (2);
 - (c) in section 16 (meaning of "relevant regulators" and "relevant functions"), omit subsection (2)(a) and (b);
 - (d) in section 22 (reports of special inspections), omit subsection (5).

10

15

20

25

30

35

CHAPTER 7

CORPORATE GOVERNANCE AND AUDIT COMMITTEES

148 Renaming of audit committees

- (1) In section 81 of the 2011 Measure (local authorities to appoint audit committees), in subsection (1), for "(an "audit committee")" substitute "(a "corporate governance and audit committee")".
- (2) Schedule 9 makes consequential amendments to the 2011 Measure relating to the renaming of audit committees.

149 Functions of corporate governance and audit committee

(1) After section 81 of the 2011 Measure insert –

"81A Further provision about functions of committees

- (1) In exercising its functions under section 81(1) the corporate governance and audit committee of a local authority must, in particular, assess—
 - (a) the authority's ability to comply with its duties under section 111 of the Local Government (Wales) Act 2017;
 - (b) the authority's ability to make progress in relation to the priorities set out in the statement included in its corporate plan under section 112(2) of that Act;
 - (c) the arrangements made by the authority
 - (i) in relation to self assessments under section 116 of that Act;
 - (ii) under section 118 of that Act (peer assessments), and
 - (d) the local authority's ability to handle complaints effectively.
- (2) The corporate governance and audit committee must report to the local authority on an assessment under subsection (1).
- (3) A corporate governance and audit committee also has functions under
 - (a) section 117 of the Local Government (Wales) Act 2017 (self assessment reports);
 - (b) section 121 of that Act (responses to peer assessment reports);
 - (c) section 126 of that Act (responses to combined assessment reports);
 - (d) section 134 of that Act (responses to governance review reports)."

150 Membership of corporate governance and audit committee

(1) Section 82 of the 2011 Measure (membership) is amended as follows.

- (2) In subsection (2)
 - (a) in paragraph (a) omit the words "at least";
 - (b) for paragraph (b) substitute
 - "(b) one-third of the members of that committee are lay persons;".
- (3) After subsection (5) insert
 - "(5A) A corporate governance and audit committee is to appoint
 - (a) a member of the committee as its chair ("the committee chair"), and
 - (b) a member of the committee as the deputy to the committee chair ("the deputy chair").
 - (5B) The member appointed as the committee chair must be a lay person.
 - (5C) The member appointed as the deputy chair must not be a member of the local authority's executive."
- (4) Omit subsection (6).

5 151 Meaning of lay person

In section 87 of the 2011 Measure (interpretation), omit the definition of "lay member" and insert –

""lay person" ("person lleyg") means a person who –

- (a) is not a member or an officer of any local authority,
- (b) has not at any time in the period of twelve months ending with the date of that person's appointment been a member or an officer of any local authority, and
- (c) is not the spouse or civil partner of a member or an officer of any local authority;".

25 **152 Proceedings etc.**

- (1) Section 83 of the 2011 Measure (proceedings) is amended as follows.
- (2) For subsections (1) and (2) substitute
 - "(1) A meeting of a corporate governance and audit committee is to be chaired—
 - (a) by the committee chair, or
 - (b) if the committee chair is absent, by the deputy chair.
 - (2) If both the committee chair and the deputy chair are absent the committee may appoint another member of the committee (who may not be a member of the local authority's executive) to chair the meeting."
- (3) Omit subsection (8).

35

30

5

10

15

20

25

30

35

PART 6

COMMUNITY COUNCILS

CHAPTER 1

REVIEW OF COMMUNITY COUNCIL ARRANGEMENTS

5 153 Duty to review community council arrangements

- (1) The Local Democracy and Boundary Commission for Wales ("the Commission") must conduct a community council review in relation to each of the counties in Wales, as set out in Schedule 1 (but the Commission may begin to conduct the reviews at any time before 1 April 2020).
- (2) In this Chapter, "community council review", in relation to a county, means a review of the communities in the area of the county which is conducted for the purpose of recommending that the council for the county makes changes in respect of
 - (a) councils for communities and common community councils in the area of the county, and
 - (b) the electoral arrangements for communities in the area of the county.
- (3) For the purposes of subsection (2)(a), the Commission may recommend that the council for a county under review should, in relation to communities in the area of the county and their councils
 - (a) establish a separate council for a community;
 - (b) dissolve a separate council for a community;
 - (c) group two or more communities together under a new common community council;
 - (d) add one or more communities to a group of communities for which there is a common community council;
 - (e) separate one or more communities from a group of communities;
 - (f) dissolve a group of communities and its common community council;
 - (g) give a name to a new, or change the name of an existing, council for a community or common community council.
- (4) In subsection (2)(b) "electoral arrangements", in relation to a community, means
 - (a) the number of members of a council for the community or a common community council for a group which includes the community;
 - (b) the division of the community into community wards for the purpose of the election of members to a council for the community or a common community council for a group which includes the community;
 - (c) the apportionment of members of a council for the community, or of a common community council for a group which includes the community, among community wards;
 - (d) the number, type and boundaries of any community wards into which the

10

15

20

25

30

35

40

community is to be divided for the purpose of the election of members to a council for the community or a common community council for a group which includes the community;

- (e) the number of members to be elected for a community ward;
- (f) the name of a community ward.
- (5) In subsection (4)(d) the reference to the type of a community ward is a reference to whether the community ward is a single member ward or a multiple member ward; and for this purpose—

"multiple member ward" ("ward amlaelod") means a ward in respect of which a specified number (greater than one) of members are to be elected for the ward;

"single member ward" ("ward un aelod") means a ward in respect of which only one member is to be elected.

- (6) The Commission may, in consequence of recommendations it makes in respect of matters set out in subsection (2)(a) and (b), recommend that a county council should
 - (a) change the boundaries of communities in the area of the county;
 - (b) constitute a new community within the area of the county (whether by merging existing communities in whole or in part, or otherwise);
 - (c) abolish an existing community in the area of the county,

and the Commission may also make recommendations as to the name of an altered or new community.

154 Conduct of community council review

- (1) In conducting a community council review, the Commission must seek to ensure effective and convenient local government.
- (2) In considering on a community council review whether a community should be divided into community wards, the Commission must have regard to
 - (a) whether the number or distribution of the local government electors for the community makes a single election of community councillors impractical or inconvenient, and
 - (b) whether it is desirable that any area of the community should be separately represented on the community council.
- (3) Where it is decided on a community council review that a community should be divided into community wards, or that changes should be made to any existing wards in a community, in considering the size and boundaries of the wards and in setting the number of community councillors to be elected for each ward the Commission must have regard to—
 - (a) any change in the number or distribution of local government electors for the community which is likely to take place within the period of five years immediately after any recommendation is made,
 - (b) the desirability of setting boundaries which are and will remain easily identifiable, and

10

15

20

25

30

35

40

- (c) any local ties which will be broken by the setting of any particular boundaries.
- (4) Where it is decided on a community council review that a community which is divided into wards should no longer be divided into wards, or that the number of councillors to be elected for a community which is not divided into wards should change, in setting the number of councillors to be elected for each community the Commission must have regard to—
 - (a) the number and distribution of local government electors for the community, and
 - (b) any change in the number or distribution of local government electors for the community which is likely to take place within the period of five years immediately after the number of community councillors is set.
- (5) For the purposes of subsections (2) to (4) account is to be taken of any discrepancy between the number of local government electors and number of persons who are eligible to be local government electors (as indicated by relevant official statistics).
- (6) In this section –

"relevant official statistics" ("ystadegau swyddogol perthnasol") means the official statistics within the meaning of section 6 of the Statistics and Registration Service Act 2007 (c.18) which the Commission considers appropriate;

"local government elector" ("etholwr llywodraeth leol") means a person registered as a local government elector in the register of electors in accordance with the provisions of the Representation of the People Acts.

155 Pre-review notification

Before conducting a community council review, the Commission must take the steps it considers appropriate to bring the review to the attention of —

- (a) every county council and county borough council established under section 20 of the 1972 Act whose area is in the area of the county in relation to which the review will be conducted;
- (b) the councils for communities and common community councils in the area of the county;
- (c) the local government electors for a community in the area of the county which does not have a separate council and is not grouped under a common community council;
- (d) every organisation which represents a council for a community or a common community council in the area of the county and which has asked the Commission to be consulted;
- (e) every trade union which is recognised (within the meaning of the Trade Union and Labour Relations (Consolidation) Act 1992 (c.52)) by a council for a community or a common community council in the area of the county;
- (f) any other persons it considers likely to be interested in the review.

156 Consultation, investigation and report

(1) In conducting a community council review, the Commission –

10

15

20

25

30

35

- (a) must consult
 - (i) every county council and county borough council established under section 20 of the 1972 Act whose area is in the area of the county in relation to which the review is being conducted,
 - (ii) the council for the county in relation to which the review is being conducted (after the ordinary elections of councillors for the that county held on 2 May 2019 under section 14),
 - (iii) the review consultees, and
 - (iv) any other persons it considers appropriate, and
- (b) must conduct the investigations it considers appropriate.
- (2) In this Chapter "the review consultees", in relation to a county, are
 - (a) the councils for communities and common community councils in the area of the county;
 - (b) the local government electors for a community in the area of the county which does not have a separate council and is not grouped under a common community council;
 - (c) every organisation which represents a council for a community or a common community council in the area of the county and which has asked the Commission to be consulted;
 - (d) every trade union which is recognised (within the meaning of the Trade Union and Labour Relations (Consolidation) Act 1992 (c.52)) by a council for a community or a common community council in the area of the county.
- (3) Every council for a community and common community council must provide the Commission which such information as the Commission may reasonably require in connection with a community council review.
- (4) After conducting a community council review, the Commission must prepare a report containing—
 - (a) any recommendations for change in respect of the matters in section 153(2) it considers appropriate or, if it does not consider any change appropriate, a recommendation to that effect, and
 - (b) details of the review it conducted.
- (5) The Commission must, on or after 1 April 2020
 - (a) send the report to—
 - (i) the council for the county to which the review relates,
 - (ii) the councils for communities and common community councils in the area of the county, and
 - (iii) the Welsh Ministers,
 - (b) publish the report on a website,
 - (c) secure that the report is available for inspection (without charge) during the period for representations at its offices and in the offices of the council for the

10

15

20

30

35

county to which the report relates,

- (d) secure that the report is available for inspection (without charge) during the period for representations at a location in each of the communities in the area of the county to which the report relates,
- (e) inform the following persons of how to obtain a copy of the report
 - (i) the review consultees (other than those mentioned in paragraph (a)(ii)),
 - (ii) any other person it consulted under subsection (1)(a)(iv), and
 - (iii) any other person who submitted evidence to it in the course of the review (other than a county council or county borough council which ceases to exist on 1 April 2020),
- (f) invite representations on the recommendations in the report, and
- (g) notify the following persons of the period for representations
 - (i) the council for the county in relation to which the review is being conducted,
 - (ii) the review consultees,
 - (iii) any other person it consulted under subsection (1)(a)(iv),
 - (iv) any other person who submitted evidence to it in the course of the review (other than a county council or county borough council which ceases to exist on 1 April 2020),
 - (v) the Welsh Ministers, and
 - (vi) any other person it considers appropriate.
- (6) For the purposes of subsection (5), "the period for representations" is a period of not less than 6, nor more than 12, weeks (as determined by the Commission) starting no earlier than one week after notice of the period is given under paragraph (g).

25 **157 Final report on review**

- (1) After the period for representations under section 156(5) has ended, the Commission must consider its recommendations having regard to any representations received by it during that period.
- (2) The Commission must then prepare a further report containing
 - (a) any recommendations for change in respect of the matters in section 153(2) it considers appropriate or, if it does not consider any change appropriate, a recommendation to that effect,
 - (b) details of the review conducted and the consultation carried out in respect of the recommendations, and
 - (c) details of any changes to the recommendations made in the light of the representations received, and an explanation of why those changes have been made.
- (3) The Commission must
 - (a) submit the report to the council for the county to which the report relates no later

10

15

20

25

30

35

than the date specified by the Welsh Ministers in regulations,

- (b) publish the report on a website,
- (c) secure that the report is available for inspection (without charge) at its offices and in the offices of the council for the county to which the report relates, for a period of at least 6 weeks starting with the date on which the report is submitted to the council.
- (d) secure that the report is available for inspection (without charge) at a location in each of the communities in the area of the county to which the report relates for a period of at least 6 weeks starting with the date on which the report is submitted to the council,
- (e) send a copy of the report to—
 - (i) the councils for communities and common community councils in the area of the county, and
 - (ii) the Welsh Ministers, and
- (f) inform the following persons of how to obtain a copy of the report
 - (i) the review consultees (other than those mentioned in paragraph (e)(i)),
 - (ii) any other person it consulted under section 156(1)(a)(iv), and
 - (iii) any other person who submitted evidence to it in the course of the review (other than a county council or county borough council which ceases to exist on 1 April 2020).

158 Implementation of Commission recommendations by county council

- (1) The council for the county to which a further report prepared by the Commission under section 157 relates must, where the report contains recommendations for change, by order
 - (a) implement those recommendations without modification, or
 - (b) implement those recommendations with such modification as may be agreed with the Commission.
- (2) The county council may not make an order under subsection (1)
 - (a) before the end of the period of 6 weeks starting with the date on which the Commission submitted its further report under section 157 to the council, and
 - (b) after the end of the period of 6 months starting with that date.
- (3) An order under subsection (1) must come into force before the date specified by the Welsh Ministers in regulations.
- (4) The county council must, on the day it makes the order
 - (a) send a copy of the order to the Commission and to each council for a community and common community council affected by the order, and
 - (b) publish the order on a website.

159 Community council arrangements order if Commission does not submit report

10

15

20

25

30

35

- (1) If the Commission has not submitted a final report under section 157 to the council for the county to which the report relates before the date specified by the Welsh Ministers under section 157(3) the Welsh Ministers may make an order under subsection (2).
- (2) The Welsh Ministers may by order make provision
 - (a) for any of the things set out in section 153(3), in relation to the county to which the report relates;
 - (b) changing the electoral arrangements of communities in that county.
- (3) An order under subsection (2) may also make relevant consequential provision in relation to the county in question.
- (4) The Commission must provide the Welsh Ministers with such information as the Welsh Ministers may reasonably require relating to any matters which have come to its attention in consequence of
 - (a) any consultation under section 156,
 - (b) any investigations under that section,
 - (c) the preparation of a report under section 156 or 157, or
 - (d) anything else done in the conduct of a community council review.
- (5) The Welsh Ministers must, on the day they make an order under subsection (2)
 - (a) send a copy of the order to the Commission, the county council and each council for a community and common community council affected by the order, and
 - (b) publish the order on a website.
- (6) The Welsh Ministers may by order vary or revoke an order under subsection (2) (or this subsection).
- (7) In this section –

"electoral arrangements" ("trefniadau etholiadol") has the meaning given in section 153(4), and

"relevant consequential provision" ("darpariaeth ganlyniadol berthnasol") means provision doing anything that the Commission could recommend under section 153(6).

160 Implementation by Welsh Ministers if county council does not implement

- (1) If a county council has not made an order under section 158 before the end of the period of 6 months starting with the date on which the Commission submitted its further report under section 157 to the council, the Welsh Ministers may by order—
 - (a) implement the recommendations for change contained in the report without modification, or
 - (b) implement those recommendations for change with such modification as may be agreed with the Commission.
- (2) The Commission must provide the Welsh Ministers with any further information in relation to its recommendations which the Welsh Ministers may reasonably require.
- (3) The Welsh Ministers must, on the day they make an order under subsection (1) –

10

15

20

25

30

35

- (a) send a copy of the order to the Commission, the county council and each council for a community and common community council affected by the order, and
- (b) publish the order on a website.

161 Implementation orders: transitional, consequential etc. provision

- (1) An order made under section 158, 159 or 160 may make such incidental, consequential, supplemental or transitional provision as the implementing authority considers necessary or expedient.
- (2) Such an order may in particular make provision about
 - (a) the assignment of existing members of a council for a community or a common community council to new or altered community wards;
 - (b) the first election of members of a council for a community or a common community council for any new or altered community ward;
 - (c) the holding of a fresh election of members for all community wards in the community in question;
 - (d) the order of retirement of members for a community ward;
 - (e) the constitution, election to and membership of any public body in any area or community ward affected by the order;
 - (f) any of the matters in section 163(2).
- (3) In this section –

"public body" ("corff cyhoeddus") includes —

- (a) a county council, a council for a community and a common community council;
- (b) any trustees, commissioners or other persons who, for public purposes and not for their own profit, act under any enactment or instrument for the improvement of any place, for the supply of water to any place, or for providing or maintaining a cemetery or market in any place;
- (c) any other authority having powers of levying or issuing a precept for any rate for public purposes.

"implementing authority" ("awdurdod gweithredu") means —

- (a) in the case of an order under section 158, the county council which made the order;
- (b) in the case of an order under section 159 or 160, the Welsh Ministers.

Variation and revocation of implementation orders

- (1) Other than as provided for by this section, orders made under this section or section 158 or 160 may not be varied or revoked.
- (2) An implementing authority may by order vary or revoke any provision which is
 - (a) contained in an order made under this section or section 158 or 160, and
 - (b) of a type described in section 161(2).

10

15

20

25

30

35

40

- (3) Before making an order under subsection (2) the implementing authority must comply with subsections (4) and (5).
- (4) The implementing authority must
 - (a) send a copy of a draft of the order to any public body it considers likely to be affected by the order,
 - (b) publish the draft order on a website and in such other manner as it considers likely to bring the order to the attention of persons who may have an interest in it,
 - (c) secure that a copy of the draft order is available for inspection (without charge) by interested persons at such places as the implementing authority considers appropriate, and
 - (d) invite representations in relation to the draft order to be made within the period of 2 months starting on the date of publication on a website under paragraph (b).
- (5) The implementing authority must consider any representations received within the 2 month period, and may modify the order in the light of such representations.
- (6) Where an implementing authority is satisfied that a mistake has occurred in the preparation of an order it has made under this section or section 158 or 160, it may by order make such provision as it considers necessary or expedient to rectify the mistake.
- (7) In subsection (6) "mistake", in relation to an order, includes a provision contained in or omitted from the order in reliance on information supplied by any public body which is inaccurate or incomplete.
- (8) An implementing authority must, on the day it makes an order under this section
 - (a) send a copy of the order
 - (i) to the Commission,
 - (ii) to each council for a community and common community council affected by the order,
 - (iii) if the implementing authority is the Welsh Ministers, to the council for the county to which the order relates, and
 - (iv) if the implementing authority is a county council, to the Welsh Ministers, and
 - (b) publish the order on a website.
- (9) In this section –

"implementing authority" means -

- (a) in the case of an order under section 158, and subject to subsection (10)(a), the county council that made the order;
- (b) in the case of an order under section 160, the Welsh Ministers;

"public body" has the meaning given in section 161.

- (10) Where an order under this section or section 158 was made by a county council which no longer exists ("the old council")
 - (a) "implementing authority" means a county council which is a successor to the old council, but an order made by a successor county council may vary or revoke

10

15

20

25

30

- provision in the order under section 158 only in so far as it relates to the successor county council's area;
- (b) subsection (6) is to be read as enabling the successor county council to make an order rectifying a mistake in the preparation of an order made by the old council, but only in so far as the mistake relates to the successor county council's area.

163 General consequential, transitional etc. provision

- (1) The Welsh Ministers may by regulations make such incidental, consequential, supplemental or transitional provision as they consider necessary or expedient for the purposes of, or in connection with, giving full effect to orders made under section 158, 159 or 160.
- (2) Regulations under this section may in particular make provision
 - (a) about the functions, area or jurisdiction in or over an area (or part of an area), of any public body or office within an area (or community ward) affected by an order made under this Chapter;
 - (b) about the costs and expenses of a public body or office affected by such an order;
 - (c) for the transfer of staff of any such public body or office;
 - (d) for the transfer, management or custody of property (whether real or personal) and the transfer of rights and liabilities (including criminal liabilities);
 - (e) for the transfer of civil or criminal proceedings;
 - (f) in respect of charter trustees.
- (3) Regulations under this section may apply (with or without modifications), amend, repeal or revoke any charter, whenever made or granted, or enactment.
- (4) In this section, "public body" has the same meaning as in section 161.

164 Transfer of staff

- (1) This section applies to a transfer of staff which is provided for by an order or regulations made under this Chapter.
- (2) The Transfer of Undertakings (Protection of Employment) Regulations 2006 (S.I. 2006/246) ("the TUPE Regulations") apply to a transfer of staff to which this section applies (whether or not the transfer is a relevant transfer for the purposes of the TUPE Regulations), subject to subsection (3).
- (3) The following regulations of the TUPE Regulations do not apply to a transfer of staff to which this section applies
 - (a) regulation 4(6) (criminal liability), and
 - (b) regulation 10 (pensions).

35 **165 Directions and guidance**

- (1) The Welsh Ministers may give the Commission directions relating to the exercise of its functions under this Chapter.
- (2) Before giving a direction under this section, the Welsh Ministers must consult the

10

15

20

25

30

35

Commission and such bodies representing the interests of county councils (and, before 1 April 2020, county borough councils), councils for communities and common community councils as they consider appropriate.

(3) In exercising any function under this Chapter, the Commission and every county council (and, before 1 April 2020, every county council and county borough council established under section 20 of the 1972 Act) must have regard to any guidance issued by the Welsh Ministers.

Modification and amendment of section 30 of the 1972 Act

- (1) In section 30 of the 1972 Act (restriction on community applications during and after reviews), subsection (1) is to be read as if after paragraph (ba) there were inserted—
 - "(bb) during the period of two years beginning with the coming into force of an order relating to the community made under section 158, 159 or 160 of the Local Government (Wales) Act 2017; or".
- (2) Section 30 of the 1972 Act is to be read as if for subsection (3) there were substituted
 - "(3) The Welsh Ministers may direct that no community applications may be made until the Welsh Ministers further direct."
- (3) The Welsh Ministers may by regulations repeal subsection (2).
- (4) In section 30 of the 1972 Act, in subsection (3) for "or Part 3" substitute "under Part 3".

CHAPTER 2

MEMBERS OF COMMUNITY COUNCILS TO COMPLETE TRAINING

Training for members of community councils

167 County councils to consider the training needs of community councillors and to provide training

- (1) A county council must
 - (a) consider whether it should be compulsory for members of community councils in the area of the county to complete any training (see section 168), and
 - (b) secure the provision of any training which it is compulsory for members of those councils to complete.
- (2) In exercising its functions under this section a county council must have regard to any guidance issued by the Welsh Ministers.

Duty to complete training

168 Members of community councils to complete training

- (1) A member of a community council must complete all compulsory training.
- (2) But a member is not in breach of the duty under subsection (1) if the member has a good reason for not complying with the duty.
- (3) Training is compulsory if –

10

15

20

25

30

35

- (a) the community council has been notified in writing by the county council in whose area the area of the community council falls that it is compulsory for the member to complete the training, and
- (b) the community council has notified the member in writing that he or she must complete the training.
- (4) If a county council notifies a community council that it is compulsory for a member to complete training specified in the notice, the community council must notify the member that he or she must complete that training.

169 Complaints about a member's failure to complete training

- (1) A person may make a complaint about a possible breach by a member of a community council of the duty imposed by section 168 to the clerk of that council.
- (2) A complaint under subsection (1) must be made in writing.

170 Notifying the community council that a member has failed to complete training

- (1) If the clerk of a community council—
 - (a) receives a complaint that a member of the community council has breached the duty under section 168, or
 - (b) otherwise has reason to believe that a member has breached that duty,
 - the clerk must give the member written notice that he or she may make representations to him or to her, either orally or in writing, in respect of the alleged failure.
- (2) The clerk must determine the period within which a member may make representations under subsection (1); but the period must not be less than 28 days starting with the date on which the member is given notice under that subsection.
- (3) After the period determined under subsection (2) for making representations has ended, the clerk must determine whether the member has breached the duty, having first—
 - (a) considered any representations made to him or to her, and
 - (b) in cases where the member in question is not the chair of the community council, consulted the chair of the community council.
- (4) If the clerk determines that the member has breached the duty, the clerk must notify the community council of that determination at a meeting of the council.

CHAPTER 3

COMMUNITY COUNCIL ELECTION DATES

171 Change to date of community council elections and extension of term of office

- (1) Subsections (2) and (3) come into force on a date specified by the Welsh Ministers by order under section 183, which must be after the date in 2017 on which the ordinary elections of community councillors take place.
- (2) In the 1972 Act, in subsection (2) of section 35 (election years)
 - (a) for "2017" substitute "2023", and

Tudalen 232

10

15

20

25

30

35

- (b) for "fourth" substitute "fifth".
- (3) In the 1972 Act, in subsection (2A) of section 35 (community councillors' terms of office) for "four years" substitute "five years".
- (4) During the period starting with the coming into force of subsections (2) and (3) and ending with the ordinary elections of community councillors in 2023, subsection (2A) of section 35 of the 1972 Act is to be read as if for "five years" there were substituted "six years".

PART 7

WORKFORCE MATTERS

CHAPTER 1

GUIDANCE TO PUBLIC BODIES ON WORKFORCE MATTERS

172 Ministerial guidance to public bodies on workforce matters

- (1) The Welsh Ministers may issue guidance to public bodies about workforce matters.
- (2) A public body to which guidance is issued must have regard to it.
- (3) Guidance under this section may be issued to a particular public body, public bodies of a particular description or to public bodies generally.
- (4) Before issuing, revising or withdrawing guidance under this section, the Welsh Ministers must consult
 - (a) each public body to which the guidance relates,
 - (b) any trade union which is recognised (within the meaning given by the Trade Union and Labour Relations (Consolidation) Act 1992 (c.52)) by a public body to which the guidance relates, and
 - (c) such other persons as the Welsh Ministers consider appropriate.
- (5) The Welsh Ministers must publish any guidance issued under this section (including revised guidance) on a website and in such other form as they think fit.

173 Meaning of "workforce matters"

In this Part "workforce matters" means –

- (a) planning by public bodies in relation to the size and composition of their workforce;
- (b) the recruitment and retention of staff of public bodies;
- (c) the management, organisation and remuneration of staff of public bodies;
- (d) training and development of staff of public bodies;
- (e) the provision of information to staff of public bodies (including trade unions representing staff) about decisions affecting their work and about opportunities to contribute to those decisions;
- (f) arrangements about any of the matters mentioned in paragraphs (a) to (e) where

two or more public bodies collaborate in exercising their functions;

(g) sharing information for the purpose of facilitating any of the matters mentioned in paragraphs (a) to (f).

174 Meaning of "public body"

5

10

15

20

25

30

35

- (1) In this Part "public body" means
 - (a) a county council;
 - (b) a community council;
 - (c) a fire and rescue authority;
 - (d) the governing body of a maintained school or federation;
 - (e) a Local Health Board;
 - (f) a National Health Service trust established under section 18 of the National Health Service (Wales) Act (c.42);
 - (g) a National Park authority for a National Park in Wales;
 - (h) the Arts Council of Wales;
 - (i) the Higher Education Funding Council for Wales;
 - (j) the Local Democracy and Boundary Commission for Wales;
 - (k) the National Library of Wales;
 - (1) the National Museum of Wales;
 - (m) the Natural Resources Body for Wales;
 - (n) the Royal Commission on the Ancient and Historical Monuments of Wales;
 - (o) Social Care Wales;
 - (p) the Sports Council for Wales.
- (2) The reference to a county council in subsection (1)(a) includes a reference to a shadow authority for a county set out in the table in Schedule 1 (on which, see section 15).
- (3) In subsection (1)(d) –

"maintained school" ("ysgol a gynhelir") means a school in Wales that is a community, foundation or voluntary school, a community or foundation special school or a maintained nursery school;

"federation" ("ffederasiwn") means a group of maintained schools that are federated by virtue of section 24 of the Education Act 2002 (c.32) or Chapter 1 of Part 2 of the Education (Wales) Measure 2011 (nawm 7).

175 Power to amend definition of "public body"

- (1) The Welsh Ministers may by regulations amend the definition of "public body" in section 174.
- (2) Before making regulations under subsection (1) which add or remove a person from the definition the Welsh Ministers must consult
 - (a) the person who is to be added or removed;

10

15

20

25

30

35

- (b) any trade union which is recognised (within the meaning given by the Trade Union and Labour Relations (Consolidation) Act 1992 (c.52)) by such a person;
- (c) any other persons they consider appropriate.
- (3) Regulations under subsection (1) may add a person to the definition only if the person exercises functions of a public nature which—
 - (a) have been conferred by or under an Act or Measure of the National Assembly for Wales, or
 - (b) could be conferred by an Act of the National Assembly for Wales.

CHAPTER 2

PUBLIC SERVICES STAFF COMMISSION

176 Public services staff commission

- (1) There is to be a public services staff commission (referred to in this Part as "the Commission").
- (2) The Commission is to consist of not more than 7 members appointed by the Welsh Ministers.
- (3) The Welsh Ministers must appoint one of the members to chair the Commission.
- (4) A person appointed to the Commission holds and vacates office subject to the terms of his or her appointment.
- (5) The Welsh Ministers may pay expenses, remuneration and allowances in respect of a member of the Commission.
- (6) The Welsh Ministers must provide the Commission with such staff, accommodation and other resources as the Welsh Ministers consider necessary for the Commission to perform the functions conferred upon it by section 177.

177 Commission's functions etc.

- (1) The Commission must provide such advice on workforce matters as the Welsh Ministers may request.
- (2) The Commission may, for the purpose of preparing any advice, request such information from a public body as the Commission considers necessary or appropriate.
- (3) A public body to which a request for information is made under subsection (2) must comply with the request unless the body considers that doing so would
 - (a) be incompatible with its own duties, or
 - (b) otherwise have an adverse effect on the exercise of its functions.
- (4) A public body which decides not to comply with a request for information made under subsection (2) must notify the Commission in writing of the reasons for the decision.
- (5) Before 30 September in each financial year the Commission must send a report (an "annual report") to the Welsh Ministers about the exercise of its functions during the previous financial year.
- (6) The Welsh Ministers must publish a copy of each annual report sent to them by the

Commission on a website and in such other form as they think fit.

178 Repeal of this Chapter

- (1) Subsections (1) to (4) of section 177 are repealed on 1 April 2021.
- (2) Section 176 and subsections (5) and (6) of section 177 are repealed on 1 October 2021.
- (3) This section is repealed at the end of the period of one year starting with the date on which the repeals in subsection (1) take effect.
- (4) The Welsh Ministers may by regulations substitute a different date for the date for the time being specified in subsection (1), but the new date may not be a date which is later than 1 April 2023.
- (5) Regulations under subsection (4) may also substitute a different date for the date for the time being specified in subsection (2).

PART 8

GENERAL

179 Interpretation

In this Act –

"elected mayor" ("maer etholedig") has the same meaning as in section 39(1) of the 2000 Act:

"executive" ("gweithrediaeth") has the same meaning as in section 11 of the 2000 Act;

"executive arrangements" ("trefniadau gweithrediaeth") has the same meaning as in section 10 of the 2000 Act;

"financial year" ("blwyddyn ariannol") means a period of 12 months ending on 31 March;

"fire and rescue authority" ("awdurdod tân ac achub") means a fire and rescue authority constituted by a scheme under section 2 of the Fire and Rescue Services Act 2004 (c.21), or a scheme to which section 4 of that Act applies;

"mayor and cabinet executive" ("gweithrediaeth maer a chabinet") has the same meaning as in section 11(2) of the 2000 Act;

"the 1972 Act" ("Deddf 1972") means the Local Government Act 1972 (c.70);

"the 2000 Act" ("Deddf 2000") means the Local Government Act 2000 (c.22);

"the 2011 Measure" ("Mesur 2011") means the Local Government (Wales) Measure 2011 (nawm 4);

"voluntary body" ("corff gwirfoddol"), other than in section 64, means a body, other than a body exercising functions of a public nature, whose activities are not carried out for profit.

Tudalen 236

25

5

10

15

20

30

10

15

20

25

30

35

Application of Act in relation to counties and county boroughs until 1 April 2020

- (1) Until 1 April 2020 this Act, other than Part 1 and Chapter 1 of Part 6, is to be read in accordance with this section.
- (2) Any reference to a county is to be read as a reference to a county or county borough established under section 20 of the 1972 Act.
- (3) Any reference to a county council is to be read as a reference to a council for a county or county borough council established under that section.

181 Power to make consequential and transitional provision etc.

- (1) If the Welsh Ministers consider it necessary or expedient for the purpose of giving full effect to any provision of Parts 2 to 7 (other than Chapter 1 of Part 6) of this Act, or in consequence of any such provision, they may by regulations make—
 - (a) any supplemental, incidental, or consequential provision;
 - (b) any transitory, transitional or saving provision.
- (2) Regulations under subsection (1) may amend, modify, repeal or revoke any enactment.

182 Regulations and orders under this Act

- (1) A power of the Welsh Ministers to make regulations or orders under this Act
 - (a) is exercisable by statutory instrument;
 - (b) includes power to make different provision for different purposes and areas.
- (2) A statutory instrument to which this subsection applies may not be made unless a draft of the instrument has been laid before, and approved by a resolution of, the National Assembly for Wales.
- (3) Subsection (2) applies to a statutory instrument containing regulations under any of the following provisions
 - (a) section 20(1) (power to amend certain dates specified in legislation);
 - (b) section 27(1) or (2), where the regulations amend, repeal or disapply a provision of an Act of Parliament or a Measure or Act of the National Assembly for Wales, unless the regulations are made only for a purpose described in subsection (7) of that section;
 - (c) section 27(3) or (4), unless the regulations are made only for a purpose described in subsection (7) of that section;
 - (d) section 34(1) (competent community councils: competency requirements);
 - (e) section 37(4) (public participation duty: meaning of authorities connected with a county council);
 - (f) section 64(6)(b) (improvement requests: prescribing a "relevant authority");
 - (g) section 74(3) (improvement requests: further provision);
 - (h) section 111(3) (principles etc. relating to good governance);

- (i) section 178(4) (changing date on which repeal of Chapter 2 of Part 7 takes effect).
- (4) Subsection (2) also applies to any other statutory instrument containing regulations under this Act, other than regulations made under section 175 or paragraph 2(1) of Schedule 1, which amend, modify, disapply or repeal a provision of
 - (a) an Act or Measure of the National Assembly for Wales, or
 - (b) an Act of Parliament.
- (5) A statutory instrument containing regulations under this Act to which subsection (2) does not apply, other than regulations made under paragraph 2(1) of Schedule 1, is subject to annulment in pursuance of a resolution of the National Assembly for Wales.

10 **183** Coming into force

5

15

20

- (1) This Part comes into force on the day on which this Act receives Royal Assent.
- (2) The following provisions come into force 2 months after Royal Assent
 - (a) Chapters 1, 3 and 4 of Part 1 (local government areas and county councils), other than section 21;
 - (b) Part 2 (general power of competence);
 - (c) Chapter 4 of Part 3 (promoting access to local government: improvement requests);
 - (d) Chapter 1 of Part 6 (review of community council arrangements by the Local Democracy and Boundary Commission).
- (3) The other provisions of this Act come into force on a day appointed by the Welsh Ministers in an order made by statutory instrument.
- (4) An order under subsection (3) may
 - (a) make transitional, transitory or saving provision;
 - (b) appoint different days for different purposes or areas.

25 **184 Short title**

The short title of this Act is the Local Government (Wales) Act 2017.

25

30

35

SCHEDULE 1

(introduced by section 1)

COUNTIES IN WALES

Tables 1 and 2, below, set out alternative proposals for the constitution of counties in Wales on and after 1 April 2020 —

TABLE 1

	Name	Area (determined by reference to the areas which were counties and county boroughs immediately before 1 April 2020)
10	County 1	The counties of the Isle of Anglesey and Gwynedd, and the county borough of Conwy
	County 2	The counties of Denbighshire and Flintshire and the county borough of Wrexham
15	County 3	The counties of Ceredigion, Pembrokeshire and Carmarthenshire
	County 4	The county of Swansea and the county borough of Neath Port Talbot
	County 5	The county boroughs of Bridgend, Rhondda Cynon Taff and Merthyr Tydfil
20	County 6	The county boroughs of Cardiff and the Vale of Glamorgan
	County 7	The county of Monmouthshire and the county boroughs of Blaenau Gwent, Caerphilly, Torfaen and Newport
	Powys	The county of Powys

OR

TABLE 2

Name	Area (determined by reference to the areas which were counties and county boroughs immediately before 1 April 2020)
County 1	The counties of the Isle of Anglesey and Gwynedd
County 2	The county borough of Conwy and the county of Denbighshire
County 3	The county of Flintshire and the county borough of Wrexham
County 4	The counties of Ceredigion, Pembrokeshire and Carmarthenshire
County 5	The county of Swansea and the county borough of Neath Port Talbot

County 6	The county boroughs of Bridgend, Rhondda Cynon Taff and Merthyr Tydfil
County 7	The county boroughs of Cardiff and the Vale of Glamorgan
County 8	The county of Monmouthshire and the county boroughs of Blaenau Gwent, Caerphilly, Torfaen and Newport
Powys	The county of Powys

- 2 (1) The Welsh Ministers may by regulations amend the entries in the first column (name of county) in each of the above Tables, other than the entries for the county of Powys.
 - (2) Regulations under sub-paragraph (1) may make consequential amendments to other provisions of this Part.

15

20

25

35

SCHEDULE 2

(introduced by section 14)

NEW COUNCILS: FIRST ORDINARY ELECTIONS AND MEETINGS

Returning officers at first ordinary elections

- 5 1 (1) At the first ordinary elections of councillors of each new council, the returning officer is to be appointed by the county council or county borough council directed by the Welsh Ministers to make that appointment (subject to sub-paragraph (3)); and accordingly section 35(1A) of the Representation of the People Act 1983 (c.2) (requirement to appoint officer of the council) does not apply.
 - (2) A county council or county borough council to whom a direction has been given under sub-paragraph (1) must appoint one of its officers.
 - (3) The Welsh Ministers may give a direction under sub-paragraph (1) to a council only if
 - (a) the council is to be abolished by this Act, and
 - (b) it is the council for an area that will be part of the new county to which the election relates.
 - (4) Section 36(4) of the Representation of the People Act 1983 does not apply to the first ordinary elections of councillors of each new council (payment of expenditure).
 - (5) All expenditure properly incurred by a returning officer or other officer in relation to the holding of the first ordinary elections of councillors of a new council must be paid by the council that appointed the returning officer; and that council may recover some or all of that expenditure from the councils in the area of the new county
 - (a) in such proportions as they may agree, or
 - (b) if they cannot reach agreement, in such proportions as the Welsh Ministers determine.
 - (6) In Parts 2 and 3 of the Representation of the People Act 1983 as they apply in relation to the first ordinary elections of councillors of a new council, references to "the appropriate officer" are to be read as references to the returning officer appointed under this paragraph (accordingly, the definition of "appropriate officer" in section 67(7) of that Act does not apply).
- 30 Declarations of acceptance of office
 - For the purpose of taking and receiving delivery of declarations of acceptance of the office of councillor of any new council before the first meeting of that council, the chief executive of a designated council—
 - (a) is deemed to be the proper officer of the new council, and
 - (b) must transfer any declaration which has been delivered to him or her to the custody of the proper officer of the new council on the appointment of that proper officer.

First meetings of new councils

5

10

15

20

25

30

35

40

- 3 (1) The first meeting of each new council (which, under section 15, will be a shadow authority at the time of the meeting) must be held before the end of the period of 14 days starting with the day of the election held under section 14; this meeting is to be treated as the annual meeting of the new council for the year in which it takes place.
 - (2) The meeting must be convened by the chief executive of a designated council ("the designated chief executive"), and must be held at the place appointed by that chief executive.
 - (3) The notice of the meeting required by paragraph 4(2) of Schedule 12 to the 1972 Act must be published at the place where the meeting is to be held, and the summons to attend the meeting must be signed by the designated chief executive (and paragraph 4 of Schedule 12 to the 1972 Act is to be read accordingly).
 - (4) If a member of a new council provides the designated chief executive with an e-mail address before the summons to attend the meeting is sent, the designated chief executive may send the summons to the member by sending it to the e-mail address provided.
- 4 (1) Until the election of a chair at the first meeting of a new council, the persons designated by the relevant transition committee are to exercise the functions of the chair and vice-chair of the council.
 - (2) Any person designated by the relevant transition committee under sub-paragraph (1) must not vote in the first instance at the election of the chair, unless that person is a councillor for the new county (but he or she may give a casting vote in accordance with section 23(3) of the 1972 Act).
 - (3) At the first meeting of a new council the designated chief executive must exercise any functions of the proper officer of the new council relating to the meeting.
 - (4) At the first meeting of a new council, the standing orders of the designated council apply.

Qualification for membership of new council

For the purposes of section 79 of the 1972 Act (qualifications for election and holding office), in its application to a candidate for membership of a new council, the new counties are to be treated as having been established not less than 12 months before the day of the candidate's nomination.

Interpretation

6 In this Schedule –

"chief executive" ("prif weithredwr"), in relation to a county or county borough council, means the person appointed as chief executive under section 103;

"designated council" ("cyngor dynodedig"), in relation to a new council, means a county or county borough council which is to be abolished by this Act, and which is the council for an area that will be part of the new county, designated by the relevant transition committee (and the relevant transition committee may designate different county and county borough councils for different purposes under this Schedule);

Tudalen 242

"proper officer of the new council" ("swyddog priodol y cyngor newydd"), in relation to a purpose or function specified in this Schedule, means an officer of the new council appointed for that purpose or as the case may be to exercise that function;

"the relevant transition committee" ("y pwyllgor pontio perthnasol"), in relation to a new council, means the committee established by virtue of section 11 of the Local Government (Wales) Act 2015 by the councils abolished by this Act whose areas will constitute the area of the new council.

SCHEDULE 3

(introduced by section 16)

NEW COUNCILS: FINANCE

Council tax and non-domestic rating

- 5 1 (1) The new councils are billing authorities in relation to the financial year starting on 1 April 2020 and in relation to subsequent financial years.
 - (2) In this Schedule, "billing authority" means an authority which is a a billing authority for the purposes of
 - (a) Part 1 of the Local Government Finance Act 1992 (c.14) (council tax), and
 - (b) Part 3 of the Local Government Finance Act 1988 (c.41) (non-domestic rating).
 - (3) From 1 April 2020 the new councils have the same functions in relation to council tax and non-domestic rating as the abolished councils had in relation to any financial year starting before 1 April 2020.

Valuation lists

10

15

20

25

30

35

2 After section 22B of the Local Government Finance Act 1992 (c.14) insert –

"22C Amalgamated valuation lists for new Welsh billing authorities

- (1) Every new listing officer must, on 1 April 2020, compile a list ("the amalgamated list") for the new billing authority for which the officer is appointed, based on the information provided to the officer under this section.
- (2) The amalgamated list must contain all relevant information which was included in the current valuation lists compiled for the old billing authorities
- (3) The amalgamated list must also include all relevant information which was included in any current valuation list by way of an alteration.
- (4) A new listing officer's amalgamated list
 - (a) is to be treated for the purposes of this Act as the valuation list for the officer's new billing authority,
 - (b) is deemed to have come into force on
 - (i) 1 April 2005, or
 - (ii) if an order made under section 22B(3) specified a date which is before 1 April 2020, the date specified in the order, and
 - (c) remains in force until a new list is compiled under section 22B.
- (5) An amalgamated list must be maintained for so long as is necessary for the purposes of this Part, and the duty to maintain the list is not affected by the list ceasing to be in force.

- (6) Section 22B(4) does not apply in relation to an amalgamated list.
- (7) Where an amalgamated list contains information which is derived from any alteration made to any valuation list or lists from which it is derived, the amalgamated list is to be treated as having been varied on the date on which the alteration was made.
- (8) Every old listing officer must
 - (a) on or before 15 November 2019, provide the appropriate new listing officer with all relevant information recorded in the old listing officer's valuation list as at 31 October 2019, and
 - (b) on the day before 1 April 2020, provide the appropriate new listing officer with all relevant information recorded in the old listing officer's valuation list as at that date.
- (9) A new listing officer receiving any information under subsection (8)(a) must send a copy of it to his or her new billing authority as soon as is reasonably practicable.
- (10) As soon as is reasonably practicable after compiling the amalgamated list, a new listing officer must send a copy of it to his or her new billing authority.
- (11) A new billing authority receiving a copy of an amalgamated list under subsection (10) must, as soon as is reasonably practicable, deposit it at its principal office.
- (12) For the purposes of this section
 - (a) references to an old listing officer's valuation list are references to the valuation list maintained by him or her under this Act,
 - (b) a new listing officer's area is the area of the new billing authority for which he or she is appointed,
 - (c) the appropriate new listing officer, in relation to any information which relates to a dwelling, is the new listing officer for the new billing authority in whose area the dwelling is situated, and
 - (d) information is relevant in relation to a new listing officer, or the officer's area, if it relates to a dwelling which is in the officer's area.
- (13) In this section –

"current valuation list" means a valuation list that was compiled on 1 April 2005 (see section 22B(2)(b)), unless an order has been made under section 22B(3) specifying a date before 1 April 2020; in which case, it means a valuation list compiled on the date specified in the order;

"old listing officer" means a listing officer for an old billing authority;

"new billing authority" means a billing authority which is a

5

10

15

20

25

30

35

new council;

"new council" has the same meaning as in the Local Government (Wales) Act 2017 (see section 13 of that Act);

"new listing officer" means a listing officer for a new billing authority;

"old billing authority" means a billing authority which is a council for a county or county borough abolished by the Local Government (Wales) Act 2017."

Local non-domestic rating lists

3 After section 41A of the Local Government Finance Act 1988 (c.41) insert —

"41B Local non-domestic rating lists for new Welsh billing authorities

- (1) Every new valuation officer must, on 1 April 2020, compile a list ("the amalgamated list") for the new billing authority for which the officer is appointed, based on the information provided to the officer under this section.
- (2) The amalgamated list must contain all relevant information which was included in the current local non-domestic rating lists.
- (3) The amalgamated list must also include all relevant information which was included in any current local non-domestic rating list by way of an alteration.
- (4) A new valuation officer's amalgamated list
 - (a) is to be treated for the purposes of this Act as the local nondomestic rating list for the officer's new billing authority,
 - (b) is deemed to have come into force on
 - (i) 1 April 2017, or
 - (ii) if another date is specified in an order made under section 54A, that date, and
 - (c) remains in force until the next list is compiled under section 41.
- (5) Where an amalgamated list contains information which is derived from any alteration made to any list or lists from which it is derived, the amalgamated list is to be treated as having been varied on the date on which the alteration was made.
- (6) Subsections (2) to (6B) of section 41 do not apply in relation to an amalgamated list.
- (7) Every old valuation officer must—
 - (a) on or before 15 October 2019, provide the appropriate new valuation officer with all relevant information recorded in the old valuation officer's local non-domestic rating list as at 30 September 2019, and

35

30

5

10

15

20

- (b) on the day before 1 April 2020, provide the appropriate new valuation officer with all relevant information recorded in the old valuation officer's local non-domestic rating list as at that date.
- (8) A new valuation officer receiving any information under subsection (7)(a) must send a copy of it to his or her new billing authority as soon as is reasonably practicable.
- (9) As soon as is reasonably practicable after compiling an amalgamated list, a new valuation officer must send a copy of it to his or her new billing authority.
- (10) A new billing authority receiving a copy of an amalgamated list under subsection (9) must deposit it at its principal office as soon as is reasonably practicable.
- (11) For the purposes of this section
 - (a) references to an old valuation officer's local non-domestic rating list are references to the local non-domestic rating list maintained by him or her under this Act,
 - (b) a new valuation officer's area is the area of the new billing authority for which he or she is appointed,
 - (c) the appropriate new valuation officer, in relation to any information which relates to any hereditament is the new valuation officer for the new billing authority in whose area the hereditament is situated, and
 - (d) information is relevant in relation to a new valuation officer, or the officer's area, if it relates to a hereditament which is in the officer's area.
- (12) In this section –

"current local non-domestic rating list" means a local non-domestic rating list compiled on 1 April 2017, unless another date is specified in an order made under section 54A, in which case it means a valuation list compiled on the date specified in the order;

"old valuation officer" means a valuation officer for an old billing authority;

"new billing authority" means a billing authority which is a new council;

"new council" has the same meaning as in the Local Government (Wales) Act 2017 (see section 13 of that Act);

"new valuation officer" means a valuation officer for a new billing authority;

"old billing authority" means a billing authority which is a council for a county or county borough abolished by the Local

5

10

15

20

25

30

35

Government (Wales) Act 2017."

Council funds

5

10

15

20

25

30

35

- 4 (1) Each new council must, on 1 April 2020, establish a fund to be known as its council fund.
 - (2) Each new council must maintain its council fund.
 - (3) Any sums received by a new council must be paid into its council fund.
 - (4) All payments by a new council must be made out of its council fund.
 - (5) Sub-paragraphs (3) and (4) do not apply in relation to any sums to be paid into, or payments to be made out of, a trust fund.
 - (6) Section 101(1)(b) of the 1972 Act (delegation) does not apply as regards the functions of a new council in relation to its council fund.
 - (7) Each new council must keep accounts of sums paid into, and of payments made out of, its council fund.
 - (8) Any account kept only in respect of the general expenses of a new council is to be known as the council's general account and any account kept only in respect of any class of the council's special expenses (see section 35 of the Local Government Finance Act 1992 (c.14)) is to be known as a special account.
 - (9) The Welsh Ministers may make regulations
 - (a) requiring assets of a description specified in the regulations which fall within a council fund to be held in a separate fund within the council fund;
 - (b) requiring any fund (other than a trust fund) of a description specified in the regulations which is established by a new council to be maintained as a separate fund within its council fund.
 - (10) The Welsh Ministers may by regulations make provision with respect to the liability of new councils to make payments from their council funds in respect of precepts issued under Chapter 4 of Part 1 of the Local Government Finance Act 1992 (c.14).
 - (11) The Welsh Ministers may by regulations make provision for the sharing among a new council and major precepting authorities, in accordance with rules specified in the regulations, of an amount equal to all or part of any deduction that, in accordance with provision under paragraph 4(4A) of Schedule 8 to the Local Government Finance Act 1988 (c.41) (local retention of rates), falls to be made in calculating the council's non-domestic rating contribution for a financial year.
 - (12) Regulations under sub-paragraph (10) or (11) may, in particular, include provision
 - (a) that anything falling to be paid must be paid
 - (i) within a period specified in the regulations, and
 - (ii) in instalments of such amounts, and at such times, as are determined by the billing authority in accordance with rules specified in the regulations;
 - (b) that the billing authority must inform any precepting authorities when instalments will be paid and how they are to be calculated;
 - (c) that if an instalment is not paid to a precepting authority in accordance with the regulations, it is to be entitled to interest on the amount of the instalment;

- (d) as to the circumstances in which the billing authority is to be treated as having discharged the liability mentioned in sub-paragraph (10) or any liability arising under sub-paragraph (11);
- (e) as to the recovery (by deduction or otherwise) of any excess amount paid by the billing authority to any precepting authority in purported discharge of the liability mentioned in sub-paragraph (10) or any liability arising under sub-paragraph (11).

(13) In this paragraph –

5

10

"major precepting authority" ("prif awdurdod praeseptio") has the meaning given by section 39(1) of the Local Government Finance Act 1992 (c.14);

"precepting authority" ("awdurdod praeseptio") means a major precepting authority or a local precepting authority (which has the meaning given by section 39(2) of that Act).

10

15

20

SCHEDULE 4

(introduced by section 18)

TRANSITIONAL ETC. PROVISION

PART 1

MISCELLANEOUS TRANSITIONAL PROVISION

Suspension of elections of councillors to councils to be abolished by this Act

- 1 (1) No election of councillors of a council which is to be abolished by this Act is to be held after 31 December 2018 ("the suspension date") except—
 - (a) to fill a casual vacancy in the office of councillor of that council where before the suspension date
 - (i) the office has been declared to be vacant, or
 - (ii) notice of the vacancy has been given under section 89(1) of the 1972 Act, or
 - (b) where the number of casual vacancies in the office of councillor of the council occurring after the suspension date exceeds half of the total number of such offices.
 - (2) Any councillor of a council which is to be abolished by this Act holding office immediately before the suspension date, or elected after that date to fill a casual vacancy, must, unless he or she resigns his or her office or it otherwise becomes vacant, continue to hold office until 1 April 2020.
 - (3) There is no requirement to fill any casual vacancy in the office of councillor of a council which is to be abolished by this Act which occurs after the suspension date; section 89 of the 1972 Act is to be read accordingly.
 - (4) There is no requirement to fill any casual vacancy occurring during March 2020 in the office of chair or vice-chair of a council which is to be abolished by this Act.
- Existing and new local government areas with the same name
 - 2 (1) Where a new county has the same name as a local government area in existence at any time before 1 April 2020, references in any enactment passed before this Act to the local government area by name are not to be read as references to the new county.
 - (2) This section is subject to any provision to the contrary made by or under this Act.
- 30 Fire and rescue authorities
 - The Welsh Ministers are not required to cause an inquiry to be held under subsection (6) of section 4 of the Fire and Rescue Services Act 2004 (combined authorities) in respect of an order under subsection (4) of that section made in consequence of changes to local government areas made by or under this Part 1 of this Act.

PART 2

TRANSFER OF STAFF, PROPERTY, RIGHTS AND LIABILITIES

"Relevant new council"

5

10

15

20

25

30

In this Part of this Schedule "relevant new council", in relation to an abolished council, means the new council for the area which includes the area which was the area of the abolished council.

Transfer of staff of abolished councils

- 5 (1) On 1 April 2020, the members of staff of an abolished council are transferred to the employment of the relevant new council.
 - (2) The Transfer of Undertakings (Protection of Employment) Regulations 2006 (S.I. 2006/246) ("the TUPE Regulations") apply to a transfer under sub-paragraph (1) or regulations under section 19 (whether or not the transfer is a relevant transfer for the purposes of the TUPE Regulations), subject to sub-paragraph (3).
 - (3) The following regulations of the TUPE Regulations do not apply to a transfer under subparagraph (1) or regulations under section 19—
 - (a) regulation 4(6) (criminal liability), and
 - (b) regulation 10 (pensions).

Transfer of property, rights and liabilities

- 6 (1) On 1 April 2020, the property, rights and liabilities (including criminal liabilities) to which an abolished council was entitled or subject are transferred to and vest in the relevant new council.
 - (2) Anything (including civil or criminal proceedings) which—
 - (a) relates to any property, rights or liabilities transferred under sub-paragraph (1) or regulations under section 19, and
 - (b) is in the process of being done by or in relation to an abolished council immediately before 1 April 2020,

may be continued on and after that day by or in relation to the relevant new council.

- (3) Anything which was done by or in relation to an abolished council—
 - (a) for the purposes of or in connection with any property, rights or liabilities transferred under sub-paragraph (1) or regulations under section 19, and
 - (b) which is in effect immediately before 1 April 2020,

has effect on and after that day as if done by or in relation to the relevant new council.

- (4) In any instruments, contracts or civil or criminal proceedings which
 - (a) relate to any property, rights or liabilities transferred under sub-paragraph (1) or regulations under section 19, and
 - (b) are made or commenced before 1 April 2020,

a reference to an abolished council is to be treated on and after that day as a reference to,

Tudalen 251

- or as including a reference to, the relevant new council.
- (5) This paragraph does not apply in relation to rights and liabilities under a contract of employment transferred to a relevant new council under paragraph 5 or regulations under section 19.
- 5 7 (1) A certificate issued by the Welsh Ministers that property has been transferred under paragraph 6(1) or regulations under section 19 is conclusive evidence of the transfer.
 - (2) Paragraph 6, and regulations under section 19, have effect in relation to property, rights or liabilities to which they apply in spite of any provision (of whatever nature) which would otherwise prevent, penalise or restrict the transfer of the property, rights or liabilities.

10

15

20

25

SCHEDULE 5

(introduced by section 21)

THE 1972 ACT: AMENDMENTS AND TRANSITIONAL PROVISION

PART 1

MINOR AND CONSEQUENTIAL AMENDMENTS

- 1 The 1972 Act is amended in accordance with this Part of this Schedule.
- 2 In section 2 (constitution of principal councils in England) omit subsection (2B).
- 3 (1) Section 20 (new principal local government areas in Wales) is amended as follows.
 - (2) Before subsection (1) insert
 - "(A1) Section 1 of the Local Government (Wales) Act 2017 makes provision about—
 - (a) the local government areas in Wales (which are the counties set out in the table in Schedule 1 to that Act, and the communities originally established under this section subject to any subsequent changes to those communities), and
 - (b) councils for the counties set out in the table in Schedule 1 to that Act."
 - (3) Omit subsections (1) to (7).
 - (4) For subsections (9) and (10) substitute
 - "(9) The Welsh Ministers may by regulations change the name by which any of the preserved counties is for the time being known.
 - (10) A statutory instrument containing regulations under subsection (9) is subject to annulment in pursuance of a resolution of the National Assembly for Wales."
 - (5) Omit subsection (11).
 - (6) For the title of section 20, substitute "Local government areas and preserved counties in Wales".
 - (7) Accordingly, for the cross-heading preceding section 20 substitute "Local government areas and preserved counties after 1 April 2020".
- 30 4 (1) Schedule 4 is amended as follows.
 - (2) Omit Parts 1 and 2.
 - (3) For the title of that Schedule substitute "PRESERVED COUNTIES"
 - 5 Sections 21 to 26 (councils and their members) are repealed (see Part 2 of this Schedule for further provision about the repeal of section 26 (elections)).
- 35 6 (1) Section 74 (change of name of county) is amended as follows.
 - (2) In subsection (1), omit ", county borough" from both places it occurs.
 - (3) Omit subsection (2A).
 - (4) In subsection (3), for "the relevant Minister", in both places it occurs, substitute "the

Tudalen 253

10

15

20

30

Secretary of State".

- (5) Omit subsection (3A).
- (6) In subsection (4) omit ", county borough".
- (7) Omit subsections (7) and (8).
- (8) In the title, for "county, district" substitute "county or district in England".
- 7 Section 245A (petition by council of a county to Her Majesty for grant of a charter conferring county borough status) is repealed.
- 8 In section 270(1) of the 1972 Act (definitions) after the definition of "mayor and cabinet executive" insert —

""member", in relation to a principal council which is operating executive arrangements which involve a mayor and cabinet executive, means—

- (a) the elected mayor of the council;
- (b) the chair of the council;
- (c) a councillor of the council;".

PART 2

PROVISION RELATED TO REPEAL OF SECTION 26 OF THE 1972 ACT (ELECTIONS)

Ordinary elections of councillors of the council for the county of Powys in 2020

- 9 (1) There is to be an ordinary election of councillors of the council for the county of Powys in 2020 (and the subsequent ordinary election of such councillors will take place in 2023, in accordance with section 10).
 - (2) Until the ordinary elections of councillors of the county councils in 2023, section 10(3) applies in relation to the councillors of the council for the county of Powys as if for "five years" there were substituted "three years".
- 25 Councillors' terms of office
 - 10 (1) Sub-paragraph (2) applies during the period starting with the coming into force of section 10 and ending with the ordinary elections of councillors of the county councils in 2023.
 - (2) Section 10(3) applies in relation to the councillors of the councils of the descriptions set out in the first column of table 1 as if for "five years" there were substituted the words set out in the second column of table 3 —

TABLE 3

Description of council	Period of years
new councils (to be elected on 2 May 2019)	"four years"
councils which are to be abolished by this Act	"three years"

Tudalen 254

Consequential amendments of Local Government Act 2000 (c.22)

- 11 (1) The Local Government Act 2000 (c.22) is amended as follows.
 - (2) In section 85 (options for elections)
 - (a) in subsection (1) after "Part" insert "as it applies in relation to a principal council for an area in England,",
 - (b) in that subsection, before "a principal council" insert "such", and
 - (c) in the title, after "election" insert ": England".
 - (3) In section 86 (power to specify a scheme for elections), omit subsection (1).

(introduced by section 101)

ASSISTANTS TO LOCAL AUTHORITY EXECUTIVES: FURTHER AMENDMENTS

Local Government Act 1972 (c.70)

5

10

15

20

25

35

- 1 (1) The 1972 Act is amended as follows.
 - (2) In each of sections 22(1A), 24(1A), 24A(5) and 24B(3), after "principal council" insert ", or an assistant to the executive,".
 - (3) In section 80(1)(a), for "or member of the executive" substitute ", member of the executive or assistant to the executive".
 - (4) In section 116, for "or member of the executive" substitute ", member of the executive or assistant to the executive".
 - (5) In section 270(1), after the definition of "appropriate Minister" insert –

""assistant to the executive", in relation to a principal council in Wales, has the same meaning as in Schedule 1 to the Local Government Act 2000 (see paragraph 3A of that Schedule);".

(6) In Schedule 12, in paragraph 5(4), after "principal council" insert ", or an assistant to the executive,".

Pilotage Act 1987 (c.21)

- 2 In section 3 of the Pilotage Act 1987
 - (a) in subsection (9A)(a), after "local authority" insert ", or an assistant to the executive,";
 - (b) in subsection (10), after the opening words insert –

""assistant to the executive" has the same meaning as in Schedule 1 to the Local Government Act 2000 (see paragraph 3A of that Schedule);".

Local Government Finance Act 1992 (c.14)

- In section 106(2A) of the Local Government Finance Act 1992, after "to whom this section applies" insert ", and no assistant to the executive (within the meaning of paragraph 3A of Schedule 1 to the Local Government Act 2000) to whom this section applies,".
- Local Government Act 2000 (c.22)
 - In section 21(9) of the 2000 Act (overview and scrutiny committee not to include members of executive), after "executive" insert "or any assistant to the executive (within the meaning of paragraph 3A of Schedule 1)".

Local Government (Wales) Measure 2011 (nawm 4)

5 In each of the following provisions of the 2011 Measure, after "authority's executive"

insert "or an assistant to its executive" –

- (a) section 12(2)(b);
- (b) section 14(2);
- (c) in section 82, subsections (2)(c) and (3);
- (d) section 83(2).

(introduced by section 103)

CONSEQUENTIAL AMENDMENTS: CHIEF EXECUTIVE

Local Government Act 1972 (c.70)

In section 112(2A) of the 1972 Act (appointment of staff), for "heads of paid service" substitute "chief executives".

Local Government Finance Act 1988 (c.41)

- 2 The Local Government Finance Act 1988 is amended as follows.
- In section 114(3A) (consultation by chief finance officer in preparing reports), at the end of paragraph (a) insert "or, in the case of a council of a county or county borough in Wales, its chief executive".
 - In section 114A(3) (consultation by chief finance officer of authority operating executive arrangements), at the end of paragraph (a) insert "or, in the case of a council of a county or county borough in Wales, its chief executive".
- 15 Local Government and Housing Act 1989 (c.42)
 - 5 The Local Government and Housing Act 1989 is amended as follows.
 - In section 2 (politically restricted posts), at the end of paragraph (a) insert "or, in the case of a council of a county or county borough in Wales, the person appointed as its chief executive under section 103 of the Local Government (Wales) Act 2017".
- 20 7 In section 4(6) (definition of "relevant authority")
 - (a) in paragraph (a), omit "and Wales";
 - (b) omit the "and" after paragraph (a);
 - (c) after that paragraph insert
 - "(aa) in relation to Wales, means an elected local policing body; and".
 - 8 In section 5 (designation and reports of monitoring officer)
 - (a) in subsection (1), omit the words after paragraph (b) (and substitute a full stop for the semi-colon at the end);
 - (b) after subsection (1) insert
 - "(1ZA) But a relevant authority may not designate its chief finance officer as the monitoring officer.
 - (1ZB) In the case of a relevant authority which is a council of a county or county borough in Wales, the officer designated under subsection (1) above may not be the authority's chief executive.";
 - (c) in subsection (1B)
 - (i) omit "and Wales", and

35

30

25

15

20

25

30

35

- (ii) omit paragraph (b);
- (d) in subsection (3)(a), after "above" insert "or, in the case of a council of a county or county borough in Wales, its chief executive,".
- In section 5A(5)(a) (consultation by monitoring officer on reports: executive arrangements), after "above" insert "or, in the case of a council of a county or county borough in Wales, its chief executive,".

Local Government (Wales) Measure 2011 (nawm 4)

- The 2011 Measure is amended as follows.
- In section 8(4) (officers who may not be designated head of democratic services), for paragraph (a) substitute—
 - "(a) the authority's chief executive appointed under section 103 of the Local Government (Wales) Act 2017;".
- In section 9(4) (functions of head of democratic services), for the words from "head of paid service" to the end substitute "chief executive in section 103(2) of the Local Government (Wales) Act 2017".
- In section 143A (functions of the Independent Remuneration Panel for Wales in respect of salaries of chief executives)
 - (a) in subsection (1)(a) and (b), for "head of paid service" substitute "chief executive";
 - (b) in subsection (3), for "head of paid service" substitute "chief executive";
 - (c) in subsection (3B), for "head of paid service" substitute "chief executive";
 - (d) in subsection (5A), for "head of paid service" substitute "chief executive";
 - (e) in subsection (7)
 - (i) omit the definition of "head of paid service", and
 - (ii) in the definition of "salary", for "head of paid service", at both places it occurs, substitute "chief executive";
 - (f) accordingly, the heading becomes "Functions relating to salaries of chief executives".

Localism Act 2011 (c.20)

In section 43(2) of the Localism Act 2011 (definition of "chief officer": pay policy statements), at the end of paragraph (a) insert "or, in the case of a council of a county or county borough in Wales, its chief executive".

Anti-social Behaviour, Crime and Policing Act 2014 (c.12)

- In section 77 of the Anti-social Behaviour, Crime and Policing Act 2014 (duration of notices prohibiting access to certain premises), for subsection (6) substitute
 - "(6) In this section "chief executive officer" means
 - (a) in relation to a local authority in England, the authority's head of paid service designated under section 4 of the Local

Government and Housing Act 1989;

- (b) in relation to a local authority in Wales, the authority's chief executive appointed under section 103 of the Local Government (Wales) Act 2017)."
- 5 Well-being of Future Generations (Wales) Act 2015 (anaw 2)
 - In the second column of the table in paragraph 7 of Schedule 3 to the Well-being of Future Generations (Wales) Act 2015 (representation at meetings of public services boards), for "head of the authority's paid service designated under section 4 of the Local Government and Housing Act 1989 (c.42)" substitute "authority's chief executive (see section 103 of the Local Government (Wales) Act 2017)".

Local Government (Wales) Act 2015

10

20

- 17 The Local Government (Wales) Act 2015 is amended as follows.
- In section 1(2)(a) (overview of Act), for "heads of paid service" substitute "chief executives".
- In section 29(4) (restraining appointment of chief officers by direction), for paragraph (a) substitute—
 - "(a) its chief executive appointed under section 103 of the Local Government (Wales) Act 2017;".
 - 20 In section 39 (temporary extension of functions of Independent Remuneration Panel for Wales)
 - (a) in subsection (1)
 - (i) for "heads of paid service" substitute "chief executives", and
 - (ii) for "head of paid service" at both places it occurs substitute "chief executive";
 - (b) in subsection (2), for ""head of paid service" ("pennaeth gwasanaeth cyflogedig") and "salary" ("cyflog") have the meanings" substitute ""salary" ("cyflog") has the meaning";
 - (c) in the heading, for "heads of paid service" substitute "chief executives".

(introduced by section 63)

COMMUNITY AREA COMMITTEES: FURTHER AMENDMENTS AND REPEALS

Local Government Act 1972 (c.70)

10

15

20

25

30

35

- 21 (1) The 1972 Act is amended as follows.
 - (2) In section 101, after subsection (6A) insert
 - "(6B) A principal council in Wales may not make arrangements under this section for a function of the council to be discharged by a community area committee appointed under section 43 of the Local Government (Wales) Act 2017 or by a sub-committee of such a committee (but see sections 55, 58 and 61 of that Act)."
 - (3) In section 102(1A), omit "or under regulations made under section 18 of that Act".

Local Government and Housing Act 1989 (c.42)

- 22 In section 9 of the Local Government and Housing Act 1989
 - (a) in subsection (8A), for "18" substitute "17";
 - (b) for subsection (8B) substitute
 - "(8B) A community area committee appointed by a relevant authority in Wales under section 43 of the Local Government (Wales) Act 2017 shall not exercise any power under—
 - (a) section 58 of that Act (delegation of functions by community area committee), or
 - (b) section 14 or 15 of the Local Government Act 2000 (discharge of functions under executive arrangements),

so as to arrange for the discharge of any of the authority's functions by any person who holds a post under the authority to which the person was appointed in pursuance of this section.";

(c) in subsection (11), omit the definition of "community area committee".

Local Government (Wales) Act 1994 (c.19)

- 23 (1) The Local Government (Wales) Act 1994 is amended as follows.
 - (2) Sections 27 to 31 are repealed.
 - (3) In section 64(1), omit the definition of "decentralisation scheme".

Local Government Act 2000 (c.22)

- 24 (1) The 2000 Act is amended as follows.
 - (2) In section 14
 - (a) in subsection (1), for "18, 19 or 20," substitute "19 or 20, or under section 57 of the

10

15

20

25

30

35

Local Government (Wales) Act 2017,"; (b) in subsection (2)(b), omit the "or" after sub-paragraph (iii) and insert — "(iiia) by a community area committee of the authority, or"; (c) in subsection (3), omit the "or" after paragraph (a) and insert — "(aa) by a community area committee of the authority, or"; (d) in subsection (4), after "those functions" insert — (a) by a community area committee of the authority, or (b) "; (e) in subsection (5), after "those functions" insert — (a) by a community area committee of the authority, or (b) "; (f) after subsection (5) insert — "(5A) Where by virtue of this section any functions may be discharged by a community area committee of a local authority, then, unless the elected mayor otherwise directs, the committee may arrange for the discharge of any of those functions – (a) by a sub-committee of the community area committee, or (b) by an officer of the authority. (5B) Arrangements under this section may not provide for a community area committee of a local authority to discharge functions otherwise than in relation to the area of the committee." (3) In section 15— (a) in subsection (1), for "18, 19 or 20," substitute "19 or 20, or under section 57 of the Local Government (Wales) Act 2017,"; (b) in subsection (4)(b), omit the "or" after sub-paragraph (iii) and insert — "(iiia) by a community area committee of the authority, or"; (c) in subsection (5), omit the "or" after paragraph (a) and insert — "(aa) by a community area committee of the authority, or"; (d) in subsection (6), after "those functions" insert – (a) by a community area committee of the authority, or (b) "; (e) in subsection (7), after "those functions" insert — (a) by a community area committee of the authority, or

10

15

20

25

30

35

- (b) ";
- (f) after subsection (7) insert
 - "(7A) Where by virtue of this section any functions may be discharged by a community area committee of a local authority, the committee may arrange for the discharge of any of those functions—
 - (a) by a sub-committee of the community area committee, or
 - (b) by an officer of the authority.
 - (7B) Arrangements under this section may not provide for a community area committee of a local authority to discharge functions otherwise than in relation to the area of the committee.";
- (g) in subsection (8)
 - (i) for "or (iii)" substitute ", (iii) or (iiia)";
 - (ii) for "or (7)" substitute ", (7) or (7A)".
- (4) In section 17(4), omit "18,".
- (5) Section 18 is repealed.
- (6) In section 21
 - (a) in subsection (13), after paragraph (a) insert
 - "(aza) may require any member of a community area committee of the authority to attend before it to answer questions relating to any function which is exercisable by the community area committee by virtue of—
 - (i) section 14 or 15, or
 - (ii) section 55 of the Local Government (Wales) Act 2017,";
 - (b) in subsection (14), after "(a)" insert ", (aza)".
- (7) In section 48(1), at the appropriate place insert –

""community area committee" means a committee appointed under section 43 of the Local Government (Wales) Act 2017,".

Localism Act 2011 (c.20)

- 25 In Schedule 3 to the Localism Act 2011
 - (a) omit paragraph 4(b);
 - (b) omit paragraph 16.

Local Government (Wales) Measure 2011 (nawm 4)

26 Section 55 of the 2011 Measure is repealed.

Well-being of Future Generations (Wales) Act 2015 (anaw 2)

In section 38(3) of the Well-being of Future Generations (Wales) Act 2015, after paragraph (ga) insert —

"(gb) the most recent statement of priorities and objectives submitted to the local authority by each of its community area committees under section 52 of the Local Government (Wales) Act 2017;".

(introduced by section 148)

RENAMING OF COUNTY COUNCIL AUDIT COMMITTEES

- The 2011 Measure is amended as follows.
 In the heading of Chapter 2 of Part 6, before "AUDIT COMMITTEES" insert
- In the heading of Chapter 2 of Part 6, before "AUDIT COMMITTEES" ins "CORPORATE GOVERNANCE AND".
 - 3 In section 81 (local authorities to appoint audit committees)
 - (a) in the heading, after "appoint" insert "corporate governance and";
 - (b) in subsection (2), after "its "insert "corporate governance and";
 - (c) in subsection (3), for "an" substitute "a corporate governance and".
 - 4 In section 82 (membership) –

10

15

- (a) in subsection (1), after "of its" insert "corporate governance and";
- (b) in subsection (2)
 - (i) in paragraph (a) for "its audit committee" substitute "that committee";
 - (ii) in paragraph (c) for "its audit committee" substitute "that committee";
 - (iii) in paragraph (d) for "its audit committee" substitute "that committee";
- (c) in subsection (3), after "local authority's" insert " corporate governance and";
- (d) in subsection (4), for "an" substitute "a corporate governance and";
- (e) in subsection (5), for "an" substitute "a corporate governance and";
- (f) in subsection (7), for "an" substitute "a corporate governance and".
- 5 In section 83 (proceedings etc.)
 - (a) in subsection (3), for "an" substitute "a corporate governance and";
 - (b) in subsection (4), for "an" substitute "a corporate governance and";
 - (c) in subsection (7), for "an" substitute "a corporate governance and".
- 25 6 In section 84 (frequency of meetings)
 - (a) in subsection (1), for "an" substitute "a corporate governance and";
 - (b) in subsection (2), after "the" insert "corporate governance and";
 - (c) in subsection (3), for "an" substitute "a corporate governance and";
 - (d) in subsection (4), for "an" substitute "a corporate governance and".
- 30 7 In section 85 (guidance)
 - (a) in subsection (1)
 - (i) in paragraph (a) after "functions of" insert "corporate governance and";
 - (ii) in paragraph (b) for "audit" substitute "such";
 - (b) in subsection (2), after "and its" insert " corporate governance and".
- 35 8 In section 86 (termination of membership)
 - (a) in subsection (1), for "an" substitute "a corporate governance and";

- (b) in subsection (2), after "member of the" in the second place insert "corporate governance and";
- (c) in subsection (4), after "or the" insert "corporate governance and".
- 9 In section 87 (Interpretation etc.), in subsection (2) omit the definition of "audit committee".

Eitem Agenda 13

Yn rhinwedd Paragraff(au) 12, 13 Rhan 4, Atodlen 12A Deddf Llywodraeth Leol 1972.

Document is Restricted

